On Degenerations and Deformations of Sklyanin Algebras

by

Chelsea M. Walton

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Mathematics) in The University of Michigan 2011

Doctoral Committee:

Professor Karen E. Smith, Co-Chair Professor J. Tobias Stafford, Co-Chair Professor Hendrikus G. Derksen Professor Melvin Hochster Associate Professor Leopoldo A. Pando Zayas

Like many doctorate theses of others, this one is not perfect. Please note that the material in Chapters 3 and 4 has been published, but:

(1) There is a corrigendum to "Degenerate Sklyanin algebras and Generalized Twisted Homogeneous Coordinate rings", J. Algebra 322 (2009) 2508-2527, regarding Section 3.2.2.

(2) There is a minor error in the appendix (Proposition A.1).

(3) The material in Chapter 5 is not published.

Please feel free to contact me at notlaw@temple.edu if you have any questions or concerns, especially if you would like to continue the work in Chapter 5.

-Chelsea Walton, Temple University



<u>Chelsea Walton</u> 2011 All Rights Reserved For my family and friends.

ACKNOWLEDGEMENTS

It is difficult to picture my graduate career without the encouragement and influence of Professor Karen Smith. She recruited me to the University of Michigan's graduate mathematics program. We met regularly throughout my time at Michigan. She introduced me to algebraic geometry. She even became my 'foster' adviser when my adviser transferred from Michigan's math department. This list goes on and on, and I could honestly fill pages with examples of her generosity and mentorship. In short, I cannot thank her enough!

Professor Jeanne Wald has been my mentor since my sophomore year at Michigan State University. I enrolled in four of her courses at MSU, one of which she taught me how to write proofs, and another of which sparked my interest in noncommutative algebra. Needless to say, she has been a great professional influence. More importantly, it is our candid conversations about navigating academia (and life!) that has made the most impact. I greatly appreciate our friendship.

Sue Sierra has been a true 'math sibling' and a dear friend. From her providing advice over lunch, to us venting in the office, to her reading the introduction of this dissertation, she has been an invaluable resource and a tremendous support. I look forward to visiting her in the UK often.

My family and friends have been an unwavering support of my endeavors and I sincerely thank them. I would particularly like to acknowledge my parents Linda and Bill Walton, my sister Loren Walton, my aunt Theresa Robinson, my late grandfather Donald Walton, and my best friends Lance Keene, N'Zinza Wilkins, Tatyana Ray, and Scheherazade Washington.

I have enjoyed conversations with the following individuals, who have kept me mathematically excited and inspired: Raf Bocklandt, Harm Derksen, Jacob George, Hester Graves, Daniel Hernández, Gerardo Hernández, Peter Jørgensen, Brian Jurgelewicz, Ryan Kinser, Ellen Kirkman, Rajesh Kulkarni, Jim Kuzmanovich, Diane Maclagan, Robert Marsh, Tom Nevins, Dan Rogalski, Lance Small, Paul Smith, Marie Snipes, Jorge Vitoria, Ben Weiss, Ashley Wheeler, Emily Witt, Mike Zieve.

I would like thank my doctoral committee members, Professors Harm Derksen, Mel Hochster, Karen Smith, Toby Stafford, and Leopoldo Pando Zayas, for participating in my thesis defense.

I thank the faculty, staff, and fellow students (too many to list!) of the University of Michigan's mathematics department. In particular, Bennet Fauber was a tremendous help for implementing the computer algebra program Affine. I also appreciate the University of Manchester's mathematics department for welcoming me as a visiting student for the better part of four years.

I have been financially supported by the mathematics department at the University of Michigan, Rackham graduate school, and NSF grants DMS-0555750, 0502170.

Finally, I would like to thank my adviser Professor Toby Stafford. He has been amazingly generous with his time and knowledge throughout my six years as a graduate student. Not only has he introduced me to my field of research, he has dedicated ample time towards my professional development, which has included training in technical writing, speaking, networking, etc. I honestly could not imagine having a better adviser, and I look forward working with him in the future. Thanks!

TABLE OF CONTENTS

DEDICATIO	N	ii			
ACKNOWLEDGEMENTS					
LIST OF FIGURES vii					
INDEX OF N	NOTATION AND TERMINOLOGY	'iii			
CHAPTER					
I. Intro	$\operatorname{duction}$	1			
1.1 1.2 1.3	Overview	1 5 9 10			
1.4	Computational results on $Skly_3$	12			
II. Back	ground Material	14			
2.1	Noncommutative projective algebraic geometry (NCPAG) 2.1.1 Commutative motivation 2.1.2 Techniques of Artin-Tate-van den Bergh (ATV) 2.1.3 Analysis of Sklyanin algebras via Artin-Tate-van den Bergh methods 2.1.4 Linear noncommutative geometry of higher dimensions 2.1.5 Noncommutative projective schemes Gelfand-Kirillov dimension	15 16 18 26 30 31 32			
2.3	Representation theory via noncommutative projective algebraic geometry 2.3.1 Commutative motivation 2.3.2 Decomposing noncommutative projective schemes (or categories -qgr) 2.3.3 Examples Polymomial identity (PI) rings	35 37 39 41 42			
2.4	2.4.1 Basic properties of PI rings 2.4.2 On central simple algebras (CSAs) 2.4.3 On Azumaya algebras	42 42 43 44			
2.5	Bergman's Diamond Lemma	47			
III. Deger ordin	nerate Sklyanin algebras and generalized twisted homogeneous co- ate rings	53			
3.1	Structure of degenerate Sklyanin algebras	56			
3.2	Truncated point schemes of S_{deg} \ldots \ldots 3.2.1On the truncated point schemes of some S_{deg} \ldots 3.2.2Computation of V_d for $S(1,1,1)$ \ldots	61 63 64			
3.3	Point parameter ring of $S(1,1,1)$	73			

	3.3.1 Computing the dimension of P_d	5
	3.3.2 The multiplication maps $\mu_{ij}: P_i \times P_j \to P_{i+j} \ldots \ldots \ldots 7$	'8
3.4	Open problems pertaining to point parameter rings	37
3.5	On coherent noncommutative algebraic geometry	0
IV. Repre	esentation theory of deformed Sklyanin algebras I:	
On th	e three-dimensional Sklyanin algebras $Skly_3$)4
4.1	Physical Motivation	95
4.2	Strategy to classify simple finite-dimensional modules over S_{def})7
4.3	Irreducible finite dimensional representations of Sklyanin algebras 9	9
V Bopre	pontation theory of deformed Sklyanin algebras II:	
On th	The central extension D of $Skly_3$	9
		.0
5.1	Background on the geometry of D , and computation of $\text{Simp}_1 D \dots \dots \dots 11$.4
$5.1 \\ 5.2$	Background on the geometry of D , and computation of $\text{Simp}_1 D$ 11 On the computation of $\text{Simp}_{>1} D$ 11	.4
$5.1 \\ 5.2 \\ 5.3$	Background on the geometry of D , and computation of $\text{Simp}_1 D$ 11 On the computation of $\text{Simp}_{>1}D$ 11 On the center of D 11	478
$5.1 \\ 5.2 \\ 5.3 \\ 5.4$	Background on the geometry of D , and computation of $\text{Simp}_1 D$	3 4 7 8
5.1 5.2 5.3 5.4 APPENDIX	Background on the geometry of D , and computation of $\operatorname{Simp}_1 D$.4 .7 .8 24 6

LIST OF FIGURES

Figure 2.1:	Interactions between commutative graded algebras and commutative projective geometry 18
Figure 2.2:	Example of the discussion in §2.1.1 \ldots 18
Figure 2.3:	ATV technique: Algebra to Geometry
Figure 2.4:	Examples of ATV technique: Algebra to Geometry
Figure 2.5:	Interactions between noncommutative algebra and noncommutative geometry $\ldots \ldots \ldots 24$
Figure 2.6:	Example of the discussion in §2.1.2 \ldots 25
Figure 2.7:	ATV technique applied to $Skly_3$
Figure 2.8:	(Category equivalences of) the decomposition of R -qgr $\ldots \ldots \ldots \ldots \ldots \ldots \ldots 39$
Figure 2.9:	The class of PI rings contains the subclass of CSAs
Figure 2.10:	The class of PI rings contains Azumaya algebras, which in turn contains CSAs $\ldots \ldots \ldots 45$
Figure 2.11:	Resolution of ambiguities
Figure 3.1:	The curve $E = E_{111} \subseteq \mathbb{P}^2$: $x^3 + y^3 + z^3 - 3xyz = 0$
Diagram 1:	Understanding $\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(1, \dots, 1)$
Diagram 2:	Induced Cohomology from Diagram 1

INDEX OF NOTATION AND TERMINOLOGY

 $A_{\leq n}, 32$ α -derivation, 56 algebraic stack, 21 ambiguity, 49 Artin-Schelter (AS) regular algebra, 2 three-dimensional, 2 type A three-dimensional, 99 four-dimensional, 10 Azumaya algebra, 44 Azumaya locus, 47 Azumaya points, 47 $B(E_{abc}, i^*\mathcal{O}_{\mathbb{P}^2}(1), \sigma_{abc}), 29$ $B(X,\mathcal{L}), 16$ $B(X, \mathcal{L}, \sigma), 23$ Bergman's Diamond lemma, 50 central annihilator, 124 central extension, 113 central simple algebra, 43 critical module, 33 $\cosh X, 15$ Cohen Macaulay (CM) ring, module, 129 coherent, 90 coherent noncommutative algebraic geometry, 90 coherent noncommutative projective line, 92 coherent noncommutative projective scheme, 91connected graded, 1 CSA, 43 cyclic derivative, 96 D, 113 $\mathfrak{D}, 2$ D-brane, 95 deformed Sklyanin algebra, 9, 95

degenerate Sklyanin algebra, 6, 54

 $E = E_{abc}, 4, 27$ exponential growth, 34 $F_{cyc}, 96$ F-term constraint, 96 fat point module, 40 flat, 93 g, 29 $\hat{g}_2, 121$ $\hat{g}_3, 119$ $\Gamma, 61$ $\Gamma_d, 61$ gauge field, 96 gauge group, 96 Gelfand-Kirillov dimension, 33 GKdim, 33 Goldie quotient ring, 44 -gr, 15 -Gr, 14 grade, 129 graded coherent, 90 $H_A(t), 16$ hidden relation, 49 Hilbert series, 16

Hilbert series, 16 homogenized Weyl algebra, 41

inclusion ambiguity, 49 injective, 93 Irred(-qgr), 37 irreducible (monomial), 48 irrelevant ideal, 35

j(M), 129

Kaplansky's theorem, 44

Koszul, 58 Koszul dual, 59 Künneth's formula, 75

 \mathcal{L}^{σ} , 23 A, 102 A₀, 102 line module, 30 line scheme, 31 linear module, 30 linear scheme, 30 linear term, 116

 \tilde{M} , 36 $M_{\leq n}$, 32 M(p), 22 μ_{ij} , 78 marginal deformation, 97, 99 mass term, 116 max^o, 108 minimal degree, 42 -mod, 14 -Mod, 14 mult, 33 multiplicity, 33

naïve blowup, 88 noncommutative projective algebraic geometry, 15 noncommutative projective scheme, 32 noncommutative Serre's theorem, 26 noncommutative surface, 89

Ore extension, 56 overlap ambiguity, 49

P, 7

 $\mathbb{P}^{1}_{A}, \mathbb{P}^{1}_{B}, \mathbb{P}^{1}_{C}, 66 \\ \mathbb{P}^{1}_{D}, \mathbb{P}^{1}_{E}, \mathbb{P}^{1}_{F}, 67 \\ P(S(a, b, c)), 74 \\ P(S_{deg}), 8, 74 \\ \Phi_{marg}, 97, 99 \\ \Phi_{rel}, 97 \\ \pi_{d}, 20$

PBW-deformation, 95 PI degree of a central simple algebra, 43 of a prime PI ring, 44 PI ring, 42 plane curve module, 126 plane module, 30 point module, 19, 30, 40 point parameter ring, 7, 74 point scheme, 21 point scheme data, 21 polarized variety, 32 polynomial bounded growth, 34 polynomial identity ring, 42 Posner's theorem, 44 primitive, 44 Proj, 3, 16 projective scheme, 16 pro-scheme, 21

 $q_1, q_2, q_3, 72$ q-deformed theory, 116 qcoh X, 15 -qgr, 15 -QGr, 15

 R_q , 19 rank two tensor, 126 reduction, 48 reduction finite, 48 reduction unique, 48 regular algebra, 91 relevant deformation, 97 Repr_m, 97 Repr_{< ∞}, 97 resolvable, 49 right bounded, 15

$S^{!}, 59$ $S^{\sigma}, 63$ S(a, b, c), 2 $S_{def}, 9, 95$ $S_{deg}, 6, 54$ $\sigma = \sigma_{abc}, 137$ induced by the shift functor, 22 $\sigma_{D}, 115$

 $\sigma_{\infty}(t), 48$ σ -ample, 26 section ring, 16Serre's theorem, 17 $Simp_m, 97$ $\operatorname{Simp}_{<\infty}, 97$ $\operatorname{Simp}^o_{<\infty},\,107$ singular locus, 47 $Skly_3, 2$ Sklyanin algebra, 2three-dimensional, 2 smooth locus, 47strongly noetherian, 88 sub-exponential growth, 34 superpotential algebra, 96 supersymmetric string theory, 95

tors, 15 Tors, 15 torsion, 15 trivial module, 35 truncated point module, 19 truncated point scheme, 20 twisted homogeneous coordinate ring, 23

 V_d , 20 V'_d , 75 $v_{d,1}, \ldots, v_{d,6},$ 75 $\mathbb{V}(f)$, 38 vacua, 96

 $W_{d,1}, \ldots, W_{d,6}, 71$

 $X_f, 38$

 $Y_n, \, 108$ Yang-Mills theory, 94, 95, 96

 $Z_1, Z_2, Z_3, Z_0, 63$ Zhang twist, 63

CHAPTER I

Introduction

1.1 Overview

Noncommutative rings and their representations have long had a presence in mathematics, at least since the birth of quantum mechanics in the 1930s gave rise to the Weyl algebra. To study these objects, we naturally first use algebraic techniques, yet when these methods fail, one must reach outside the world of algebra. Recently the development of *noncommutative projective algebraic geometry* studies noncommutative rings and their representations by exploiting the interplay of commutative graded rings and projective geometric objects in classical algebraic geometry. This thesis contributes to this enterprise by enlarging the class of noncommutative rings which have been analyzed in this fashion.

Let k denote an algebraically closed field of characteristic not equal to 2 or 3. A kalgebra A is **connected graded** if $A = \bigoplus_{i \in \mathbb{N}} A_i$ is N-graded with $A_0 = k$. Furthermore we always consider algebras that are finitely generated in degree one. To exhibit classes of noncommutative graded algebras that are analyzed with techniques of noncommutative projective algebraic geometry, we provide a few historical remarks. In the mid-1980s, Michael Artin and William Schelter launched noncommutative projective algebraic geometry in their attempt to complete the following project.

Classify the noncommutative graded analogues of the polynomial ring in three variables.

This task proved difficult as the ring-theoretic and homological behavior of these rings could not be determined using purely algebraic techniques. First it is necessary to clarify what is meant by a noncommutative analogue of k[x, y, z]. Artin and Schelter settled on the following definition.

Definition I.1. [AS87] A connected graded ring A is called a **three-dimensional Artin-Schelter (AS) regular algebra** if A satisfies the following three properties:

- (i) global dimension 3;
- (ii) finite Gelfand Kirillov dimension; and
- (iii) the AS-Gorenstein condition: $\operatorname{Ext}_{A}^{i}(k, A) = \delta_{i,3} \cdot k$.

A few years later, the classification of three-dimensional AS-regular algebras was achieved in the seminal paper of Michael Artin, John Tate, and Michel van den Bergh [ATVdB90]. The toughest challenge was the investigation of the following family of algebras.

Definition I.2. The three-dimensional Sklyanin algebras, denoted by S(a, b, c) or $Skly_3$, are generated by three noncommuting variables x, y, z, subject to three relations:

(1.1)
$$ayz + bzy + cx^{2} = 0$$
$$azx + bxz + cy^{2} = 0$$
$$axy + byx + cz^{2} = 0$$

for $[a:b:c] \in \mathbb{P}^2_k \smallsetminus \mathfrak{D}$ where

 $\mathfrak{D} = \{ [0:0:1], [0:1:0], [1:0:0] \} \ \cup \ \{ [a:b:c] \ | \ a^3 = b^3 = c^3 = 1 \}.$

This definition is more general than the standard definition given in [ATVdB90]. Namely the latter is associated to the geometry of a smooth cubic curve in \mathbb{P}^2 , whereas our version is associated to either \mathbb{P}^2 or a cubic curve in \mathbb{P}^2 (Theorem I.4).

We now provide a brief description of Artin-Tate-van den Bergh's geometric approach to understanding noncommutative graded rings. To contrast with the noncommutative setting, note that the geometric object typically associated to a commutative graded ring is a projective scheme Proj, the set of all homogeneous prime two-sided ideals except the irrelevant ideal. One associates to the polynomial ring in two variables k[x, y], for example, the projective line $\operatorname{Proj}(k[x, y]):=\mathbb{P}^1$. However, noncommutative graded rings do not have many two-sided ideals in general. For instance, let R_q be the ring generated by noncommuting variables x, y, subject to the relation xy = qyx for q nonzero, not a root of unity (a noncommutative analogue of k[x, y]). Then $\operatorname{Proj}(R_q)$ consists of merely three non-irrelevant prime ideals! Unless one wants to consider a space of three points corresponding to the ring R_q , an alternative geometry is required.

Hence given a noncommutative ring A, rather than employing ideals, the points of the corresponding geometric object are thought of module-theoretically. More explicitly, a noncommutative point is associated to the following A-module, which behaves like a homogeneous coordinate ring of a closed point in the commutative setting.

Definition I.3. A point module over a graded ring A is a cyclic graded left Amodule $M = \bigoplus_{i \ge 0} M_i$ where dim_k $M_i = 1$ for all $i \ge 0$.

The parameterization of point modules leads us to a noncommutative generalization of the commutative object Proj. Namely for an arbitrary connected graded algebra A, the parameterization of isomorphism classes of A-point modules (if it exists as a projective scheme) is called the **point scheme** X of A; see Definition II.6. In fact, the point scheme of R_q is \mathbb{P}^1 , yet such a predictable parameterization does not always occur; cf. Theorem I.4. If R is a commutative graded ring generated in degree 1, then the point scheme of R is simply $\operatorname{Proj}(R)$.

Returning to algebra, we build a (noncommutative) graded ring B corresponding to the point scheme X of A. Furthermore this ring B is often used to study the ring-theoretic behavior of A as we will see later. Now to build B, take a projective scheme X, with invertible sheaf \mathcal{L} and automorphism σ on X. Then one can form the **twisted homogeneous coordinate ring** $B = B(X, \mathcal{L}, \sigma)$ as defined in Definition II.9. In fact when $\sigma = \mathrm{id}_X$, then we get the commutative section ring $B(X, \mathcal{L})$ of Xwith respect to \mathcal{L} , a ring that is typically used to understand (coherent sheaves on) projective schemes (Theorem II.1).

Thus point schemes and twisted homogeneous coordinate rings are respectively genuine noncommutative analogues of projective schemes Proj and section rings $B(X, \mathcal{L})$ in classical algebraic geometry.

Now in the theorem below, Artin-Tate-van den Bergh use these constructions to provide a geometric framework specifically associated to the Sklyanin algebras.

Theorem I.4. [ATVdB90] (i) The point scheme of S = S(a, b, c) is isomorphic to:

(1.2)
$$E = E_{abc} : \mathbb{V}\left((a^3 + b^3 + c^3) xyz - (abc)(x^3 + y^3 + z^3) \right) \stackrel{i}{\subset} \mathbb{P}^2$$

Here $E = \mathbb{P}^2$ if one of the a, b, or c is 0 and the sum of the other two parameters is 0; E is a smooth cubic curve if $abc \neq 0$ and $(3abc)^3 \neq (a^3 + b^3 + c^3)^3$; and E is a singular cubic curve otherwise.

(ii) Assume that E is smooth. Given the invertible sheaf $\mathcal{L} = i^* \mathcal{O}_{\mathbb{P}^2}(1)$ on E, and a certain automorphism σ of E (which corresponds to the shift functor on point mod-

ules), we have that the twisted homogeneous coordinate ring $B = B(E, i^* \mathcal{O}_{\mathbb{P}^2}(1), \sigma)$ is a noetherian domain. Moreover $\dim_k B_0 = 1$, and for $d \ge 1$: $\dim_k B_d = 3d$ if E is a smooth cubic curve; and $\dim_k B_d = \dim_k k[x, y, z]_d$ if $E = \mathbb{P}^2$.

(iii) If $E \neq \mathbb{P}^2$, then there exists a regular normal element $g \in S$, homogeneous of degree 3, so that $B \cong S/Sg$ as graded rings. Moreover if E is also smooth, then $Skly_3$ is also a noetherian domain, with the same Hilbert series as the polynomial ring in three variables; namely $H_S(t) = (1-t)^{-3}$.

Thus part (iii) implies that twisted homogeneous coordinate rings play a crucial role in determining the ring-theoretic and homological behavior of the corresponding Sklyanin algebras. The aim of this dissertation is to exploit and expand the above geometric techniques to study variants of Sklyanin algebras and other noncommutative connected graded rings.

We now list the contents of each chapter in more depth. Chapter 2 contains the background material for this dissertation. Chapter 3 analyzes degenerations of Sklyanin algebras with use of generalized twisted homogeneous coordinate rings; introductory remarks are provided in §1.2. Chapters 4 and 5 investigate the representation theory of ungraded deformations of Sklyanin algebras; introduction given in §1.3. Computational results and computer routines are presented in Chapter 4 and in the appendix; a synopsis of these results is provided in §1.4.

1.2 Generalizing twisted homogeneous coordinate rings

During the last few years, a number of examples of (even noetherian) algebras have appeared for which the techniques of [ATVdB90, ATVdB91] are inapplicable. This is because the point modules for these algebras cannot by parameterized by a projective scheme of finite type (see for example [KRS05]). Consequently one cannot form a corresponding twisted homogeneous coordinate ring. In Chapter 3, we explore a recipe suggested in [ATVdB90, §3] for building a generalized twisted homogeneous coordinate ring for *any* connected graded ring. In particular, we provide a geometric approach to examine the following degenerations of Sklyanin algebras. Moreover the results listed in this section appear in [Wal09].

Definition I.5. The rings S(a, b, c) from Definition I.2 with $[a:b:c] \in \mathfrak{D}$ are called **degenerate Sklyanin algebras**, denoted by S_{deg} .

The name is motivated by the fact that the geometry of S_{deg} involves a degenerate cubic curve. On the other hand, we establish algebraic properties of S_{deg} in the following theorem.

Theorem I.6. (Lemma III.8, Corollary III.9, Propositions III.11 and III.12) The degenerate three-dimensional Sklyanin algebras have Hilbert series

 $H_{S_{deg}}(t) = (1+t)(1-2t)^{-1}$. They are neither left or right noetherian, nor are they domains. Moreover S_{deg} is Koszul and has both infinite global dimension and infinite Gelfand Kirillov dimension. We also have that $Z(S_{deg}) = k$.

By this result, we see that degenerate Sklyanin algebras possess entirely different ring-theoretic characteristics than Sklyanin algebras. However we can still construct geometric data for these algebras with the methods of [ATVdB90]. More precisely, we make use of the point modules over S_{deg} to construct a generalization of a point scheme for S_{deg} . Although this geometric data is not of finite type, the geometry of degenerate Sklyanin algebras is remarkably nice and still describes their class of point modules. We provide the details of this phenomenon in the theorem below. **Definition-Lemma I.7.** [ATVdB90] Let A be a connected graded ring generated by n+1 elements of degree one.

- 1. A truncated point module of length d over A is a cyclic graded left Amodule $M = \bigoplus_{i \ge 0} M_i$ where $\dim_k M_i = 1$ for $0 \le i \le d$ and $\dim_k M_i = 0$ for i > d.
- 2. The d^{th} truncated point scheme V_d parameterizes isomorphism classes of length d truncated point modules.
- 3. Assume that the inverse limit of {V_d}, with respect to restrictions of projection maps onto the first d−1 coordinates of (Pⁿ)^{×d}, stabilizes to a projective scheme. This inverse limit is referred to as the point scheme of A, and it parameterizes A-point modules.

Theorem I.8. (Proposition III.26) For $d \ge 2$, the truncated point schemes $V_d \subset (\mathbb{P}^2)^{\times d}$ corresponding to S_{deg} are isomorphic to either:

(i) a union of three copies of (P¹)^{×^d-1}/₂ and three copies of (P¹)^{×^d+1}/₂, if d is odd, or;
(ii) a union of six copies of (P¹)^{×^d/₂}, if d is even.

As previously mentioned, the inverse limit of the truncated point schemes, V_d , of degenerate Sklyanin algebras do not stabilize to produce a projective scheme of finite type. Hence we cannot mimic the approach of [ATVdB90] to construct a twisted homogeneous coordinate ring associated to S_{deg} . So to build a graded ring associated to the geometry of S_{deg} we instead make use of the V_d , or rather the **point scheme data** of S_{deg} , and a method from [ATVdB90, 3.17] that supplies us with the recipe to construct the following geometric ring.

Definition I.9. [ATVdB90, §3] The **point parameter ring** $P = \bigoplus_{d \ge 0} P_d$ is an \mathbb{N} -graded, associative ring corresponding to the subschemes V_d of $(\mathbb{P}^2)^{\times d}$ (Definition

I.7). Here $P_d = H^0(V_d, \mathcal{L}_d)$ where \mathcal{L}_d is the restriction of the invertible sheaf

$$pr_1^*\mathcal{O}_{\mathbb{P}^2}(1)\otimes\ldots\otimes pr_d^*\mathcal{O}_{\mathbb{P}^2}(1)$$

to V_d . The multiplication map $P_i \times P_j \to P_{i+j}$ is defined by taking global sections of the isomorphism $pr^*_{1,\dots,i}(\mathcal{L}_i) \otimes_{\mathcal{O}_{V_{i+j}}} pr^*_{i+1,\dots,i+j}(\mathcal{L}_j) \xrightarrow{\sim} \mathcal{L}_{i+j}$.

Point parameter rings have rarely appeared in the literature since their origin in [ATVdB91], so these rings are not understood in general. The next results of this thesis describe the behavior of the first non-trivial examples of these rings. As a consequence, we extend the class of noncommutative algebras that are traditionally studied in noncommutative projective algebraic geometry to *all* connected graded algebras (for which a point scheme does not necessarily exist).

Theorem I.10. (Proposition III.35, Theorem III.36, Corollary III.40) The point parameter ring $P = P(S_{deg})$ is generated in degree one. Since $(S_{deg})_1 \cong (P(S_{deg}))_1$, we have that P is a factor of the corresponding S_{deg} . Furthermore it has Hilbert series $H_P(t) = (1 + t^2)(1 + 2t)[(1 - 2t^2)(1 - t)]^{-1}$.

Thus Theorem I.10 yields a result surprisingly similar to Theorem I.4 (pertaining to the ring surjection from $Skly_3$ onto B(E)), despite the fact that by Theorem I.6, the rings $Skly_3$ and S_{deg} are completely unalike.

Corollary I.11. (Corollary III.41) The ring $P = P(S_{deg})$ has exponential growth. Moreover P is neither right noetherian, Koszul, nor a domain.

Therefore the behavior of $P(S_{deg})$ resembles that of S_{deg} and it is natural to ask if other noncommutative algebras can be analyzed in a similar fashion. Some of these further directions are discussed in §3.4.

Section 3.5 in particular discusses the investigation of the rings S_{deg} and $P(S_{deg})$ via Piontkovskii and Polishchuk's recent work on *coherent noncommutative algebraic*

geometry [Pio08, Pol05]. Their results thus far include the classification of noncommutative projective lines and the definition of a coordinate ring of a noncommutative coherent projective scheme. Now if S_{deg} and $P(S_{deg})$ are coherent, then the rings will contribute to the theory on *coherent noncommutative projective schemes*. In Proposition III.55, we show that the degenerate Sklyanin algebras are indeed coherent. However, we do not know if this is true for $P(S_{deg})$.

1.3 Representations of deformed Sklyanin algebras

Traditional techniques of noncommutative projective algebraic geometry analyze (noncommutative) graded rings with projective geometric data. Chapter 4 considers the question of how one might study ungraded variants of these algebras. Specifically, we examine the following ungraded deformations of three-dimensional Sklyanin algebras.

Definition I.12. For i = 1, 2, 3, let a, b, c, d_i, e_i be scalars in k with $[a : b : c] \notin \mathfrak{D}$ (Definition I.2). The **deformed Sklyanin algebra**, S_{def} , is generated by three noncommuting variables x, y, z, subject to three relations:

$$ayz + bzy + cx^{2} + d_{1}x + e_{1} = 0$$

$$azx + bxz + cy^{2} + d_{2}y + e_{2} = 0$$

$$axy + byx + cz^{2} + d_{3}z + e_{3} = 0.$$

One can show that for (almost all) parameters (a, b, c), the ring S_{def} is a PBWdeformation of $Skly_3$ [BT07][EG, §3]. The study of these rings is physically motivated as they first appeared in the study of vacua in string theory. More precisely, the analysis of *m*-dimensional simple modules over S_{def} is relevant to the study of *supersymmetric Yang-Mills theories* with *m* D-branes [BJL00]. We now discuss significant results towards achieving the following aim:

Classify the simple finite-dimensional modules over the algebra S_{def} .

Our approach is as follows. First, homogenize the relations of S_{def} with a central element w. Then, study the representation theory of the resulting *central extension* D of $Skly_3$. Since simple finite-dimensional D-modules are precisely those over $Skly_3$ or S_{def} (Lemma II.32), our objective can be reformulated in terms of studying the (not necessarily graded) simple finite-dimensional modules over the <u>graded</u> algebras, $Skly_3$ and D. The latter are nice objects in the sense of [Sta02], as they are threeand four-dimensional AS-regular algebras respectively.

Results on the classification of simple finite-dimensional modules over $Skly_3$ are provided in Chapter 4, and these are summarized in Theorem I.13 below. First we consider the following properties of $Skly_3$. Note that $Skly_3$ satisfies a *polynomial identity* or is *PI* [MR01, Chapter 13] precisely when $|\sigma| = n < \infty$ (Theorem I.18, [For74, ST94]). This scenario is of interest in representation theory as there are many finite-dimensional simple modules over such rings [BG97, §3]. Recall that $S = Skly_3$ contains a regular central element g, homogeneous of degree 3, so that $S/Sg \cong B$ a twisted homogeneous coordinate ring (Theorem I.4). Moreover B is PI if and only if S is PI.

Theorem I.13. (*Propositions IV.13, IV.16, IV.18, IV.19*)

(i) In the case of $|\sigma| = \infty$, the only finite-dimensional simple module of $Skly_3$ is the trivial module.

- (ii) Assume that $|\sigma| < \infty$. Let M be a finite-dimensional simple module over $Skly_3$.
- (a) If M is g-torsionfree, then $\dim_k M = |\sigma|$ when $(3, |\sigma|) = 1$; and $|\sigma|/3 \le \dim_k M \le |\sigma|$ when $(3, |\sigma|) \ne 1$.
- (b) Otherwise if M is g-torsion, then $\dim_k M = |\sigma|$.

In fact, we will see in Proposition I.21 that $\operatorname{PIdeg}(S) = \operatorname{PIdeg}(B) = |\sigma|$ when $|\sigma| < \infty$.

Therefore to study the representation theory of S_{def} , we now analyze the representation theory of D. This is done in Chapter 5. Fortunately, the noncommutative geometry of D has been examined in [LBSVdB96]. It is known that the point scheme P_D of D is the union of the point scheme of $Skly_3$ (the object E from Theorem I.4) and a finite set of points $\{s_i\}$. The automorphism σ_D on P_D is given by σ on Eand the identity on the finite set of points. Therefore $|\sigma_D| = |\sigma|$, and conjectures pertaining to the aforementioned research objective are formed involving this order. We claim the following dichotomy for D.

Conjecture I.14. (Conjecture V.9) The central extension D is either PI or all simple finite-dimensional D-modules are 1-dimensional.

In fact, we can describe the class of simple 1-dimensional *D*-modules.

Lemma I.15. (Lemma V.5, Corollary V.7) The nontrivial 1-dimensional simple D-modules are simple quotients of the point modules:

$$\left\{\frac{D}{Dy_1+Dy_2+Dy_3} \mid \mathbb{V}(y_1,y_2,y_3) = s_i \in P_D\right\}.$$

The set $S = \{s_i\}$ is described in the cases of physical significance in the appendix.

The approach to verifying Conjecture I.14 first involves the analysis of (fat) point **modules**, which by definition are 1-critical graded modules (of multiplicity greater than 1) over D. (These differ slightly from the commutative version.)

Lemma I.16. [LS93] [SS93] Let M be a simple finite-dimensional D-module. Then M is a quotient of some 1-critical graded module. Moreover if M arises as the quotient of a fat point module, then $\dim_k M > 1$.

Hence given a central extension D of $Skly_3$, we aim to show that either fat point modules of D do not exist or D is PI. Now we proceed by studying D according to $|\sigma_D| = |\sigma|$. If $|\sigma| = \infty$, then D is not PI (Proposition V.10). For such a D, we study its fat point modules in §5.4, which we conjecture do not exist.

On the other hand, our claim pertaining to the PI property for the central extension D of $Skly_3$ is as follows. If $|\sigma| = 1, 2$, then we claim that the center of Dis generated by only two elements of low degree, so D would not be PI in this case (Conjecture V.12, V.13). In fact in §5.4, we also analyze the fat point modules of D for the $|\sigma| = 2$ case. On the other hand if $|\sigma| = 3, 6$, then we claim that D is PI (Conjecture V.15, Remark V.16); evidence is provided in §5.3. In summary:

Conjecture I.17. For d_i and e_i generic, D is not PI when $|\sigma| = 1, 2$, and is PI when $|\sigma| = 3$ or 6.

Thus in contrast with Sklyanin algebras (i.e. Theorem I.18 below), we believe that the relationship between order of the automorphism σ_D and the structure of the center of D is quite subtle.

1.4 Computational results on *Skly*₃

Certain results presented in Chapter 4 and in the appendix deal with computational aspects of the three-dimensional Sklyanin algebras. Recall that $Skly_3 = S(a, b, c)$ comes equipped with geometric data: E, a cubic curve or \mathbb{P}^2 , with automorphism σ ; and these are defined by the parameters a, b, c. For the result of this section we assume that E is smooth, equivalently we have that either: one of a, b, c equals 0 with the sum of the other two parameters equal to 0; or $abc \neq 0$ and $(a^3 + b^3 + c^3)^3 \neq (3abc)^3$. We determine the role of these parameters in governing the behavior of the Sklyanin algebra, work motivated by the following crucial result of [ATVdB91]: **Theorem I.18.** [ATVdB91, Theorem 7.1] The algebra $Skly_3 = S(a, b, c)$ is module finite over its center if and only if $|\sigma| < \infty$.

In fact, the first condition is equivalent to S being PI [For74, ST94]. Hence if the conditions of the above theorem hold, then the behavior of S(a, b, c) is intimately tied to the structure of its center; how 'close the tie' is made precise by the *PI degree*. Furthermore recall that the twisted homogeneous coordinate ring $B(E, \mathcal{L}, \sigma)$ arises as a homomorphic image of S (Theorem I.4), so B is PI when S is PI. This naturally gives rise to the following problems.

Questions I.19.

- 1. Fix $n \in \mathbb{N}$. Let σ_{abc} be the automorphism of the point scheme E_{abc} introduced in Theorem I.4. Classify parameters (a, b, c) for which σ_{abc} has order n.
- 2. Given (a, b, c) with $|\sigma_{abc}| < \infty$, determine the PI degree of the rings S(a, b, c) and $B(E_{abc}, i^* \mathcal{O}_{\mathbb{P}^2}(1), \sigma_{abc}).$

The first problem has been settled for Sklyanin algebras associated to elliptic curves over \mathbb{Q} [LB94]. For an arbitrary field k, however, the question remains open. Partial progress is reported in Proposition A.1 in the appendix. More precisely, we have the following result.

Proposition I.20. Given n = 1, ..., 6, we know the parameters (a, b, c) of S for which $|\sigma_{abc}| = n$.

On the other hand, the result below completes the second task.

Proposition I.21. (Corollaries IV.15 and IV.19) Provided parameters a, b, c of $Skly_3 = S(a, b, c)$ for which $|\sigma_{abc}| < \infty$, both the PI degree of S(a, b, c) and the PI degree of $B(E_{abc}, i^* \mathcal{O}_{\mathbb{P}^2}(1), \sigma_{abc})$ are equal to $|\sigma_{abc}|$.

CHAPTER II

Background Material

This chapter discusses the background material for the study of the noncommutative geometry of degenerate Sklyanin algebras, and the representation theory of deformed Sklyanin algebras. In §2.1 we introduce noncommutative projective algebraic geometry (NCPAG), a field launched by the study of noncommutative graded rings via techniques of classical algebraic geometry. The next section introduces the notion of Gelfand-Kirillov dimension, which pertains to the growth of the algebras and of the modules considered in this thesis. Section 2.3 discusses the representation theory of (not necessarily graded) noncommutative rings with methods from NC-PAG. Lastly, §§2.4-2.5 are dedicated to the theory behind polynomial identity rings and Bergman's diamond lemma respectively.

Notation. Fix a \mathbb{Z} -graded ring A generated in degree one. As presented in [AZ94, §2], we consider the following module categories:

A-Mod, the category of left A-modules;

- A-mod, the category of noetherian left A-modules, which is simply the category of finitely generated left A-modules if A is noetherian;
- A-**Gr**, the category of \mathbb{Z} -graded left A-modules with degree preserving homomorphisms;

A-gr, the subcategory of A-Gr consisting of noetherian graded left A-modules.

A module $M \in A$ -Gr is **right bounded** if $M_i = 0$ for all i >> 0. Moreover M is **torsion** if all $m \in M$, we have that $A_{\geq t} \cdot m = 0$ for some t. This is equivalent to the condition that M is the (not necessarily finite) sum of right bounded modules. Now we form the following categories:

Tors(A), the full subcategory of A-Gr consisting of torsion graded left A-modules; **tors**(A), Tors(A) \cap A-gr, which is the category of finite dimensional graded A-modules if A is \mathbb{N} -graded.

We can also form the quotient categories:

 $A-\mathbf{QGr} = A-\mathrm{Gr}/\mathrm{Tors}(A);$

A-qgr = A-gr/tors(A).

The objects of the second category, A-qgr, consist of equivalence classes [M] for $M \in A$ -gr. Namely we have that $M \sim N$ if there exists an integer n for which $M_{\geq n} \cong N_{\geq n}$, where $M_{\geq n} = \bigoplus_{d \geq n} M_d$.

Furthermore for a projective variety X, we also consider $(\mathbf{q})\mathbf{coh}(X)$, the category of (quasi-)coherent sheaves on X.

2.1 Noncommutative projective algebraic geometry (NCPAG)

By no means is this a complete introduction to noncommutative projective algebraic geometry; see the survey article of Stafford and van den Bergh [SvdB01] for a more detailed discussion. The paper [AZ94] is the standard reference to the categorical approach to this field.

2.1.1 Commutative motivation

The field of noncommutative projective algebraic geometry is motivated by techniques of classical projective algebraic geometry. Namely we develop noncommutative analogues of projective schemes and corresponding homogeneous coordinate rings to study noncommutative graded rings that are suspected to behave like commutative graded rings. For instance such methods can be used to determine whether a noncommutative graded ring $A = \bigoplus_{i \in \mathbb{N}} A_i$ has the Hilbert series as that of a polynomial ring $k[x_1, \ldots, x_n]$, to say:

$$H_A(t) \coloneqq \sum_{i \in \mathbb{N}} \dim_k A_i \cdot t^i = (1-t)^{-n}$$

Let us recall some results of classical projective algebraic geometry used to study commutative graded algebras. The remaining material of this subsection is summarized from [Har77, §II.5]. First we construct projective geometric data from a given algebra, then we use this data to build a geometric ring. Let $R = \bigoplus_{i \in \mathbb{N}} R_i$ be an \mathbb{N} -graded commutative algebra, generated in degree 1 over $R_0 = k$. We first associate to R the geometric object $\operatorname{Proj}(R)$, which by definition is the set of all homogeneous prime ideals \mathfrak{p} except the irrelevant ideal $R_+ := \bigoplus_{i \geq 1} R_i$. In particular, the (not necessarily closed) points of the projective scheme $X = \operatorname{Proj}(R)$ are in 1-1 correspondence with such \mathfrak{p} . However noncommutative rings do not possess many two-sided ideals in general, and since we want these methods to generalize, we instead consider the 1-1 correspondence between points of X and quotients of R by these prime ideals \mathfrak{p} . Specifically, each closed point of X corresponds to a homogeneous prime ideal \mathfrak{p} whose quotient, R/\mathfrak{p} , is a cyclic graded R-module with Hilbert series $H_{k[x]}(t) = \sum_{i=0}^{\infty} t^i$.

Next we consider an invertible sheaf \mathcal{L} of X and construct the section ring $B = B(X, \mathcal{L}) := \bigoplus_{d \ge 0} H^0(X, \mathcal{L}^{\otimes d})$. Here B has the natural multiplication $B_d \times B_e \to B_{d+e}$

induced from taking global sections of isomorphisms: $\mathcal{L}^{\otimes d} \otimes_{\mathcal{O}_X} \mathcal{L}^{\otimes e} \cong \mathcal{L}^{\otimes d+e}$. When (X, \mathcal{L}) are nice (e.g. irreducible and normal, and ample respectively), then B is a well behaved ring (e.g. a Noetherian domain) [Gro61, Chapter 2], [Smi97, §1].

How does this help us analyze our ring R? This is answered by the following theorem of Serre.

Theorem II.1. [Ser55, Proposition 7.8] (Serre's Theorem)

(i) Let R be a commutative graded ring, finitely generated in degree one over a field R_0 . Let X = Proj(R). Then the functor R- $gr \rightarrow coh(X)$ sending M to \tilde{M} , induces an equivalence of categories: R- $qgr \sim coh(X)$.

(ii) Conversely, let $B = B(X, \mathcal{L})$ where \mathcal{L} is an ample invertible sheaf on X. Then the map $\mathcal{M} \mapsto \Gamma_*(\mathcal{M}) = \bigoplus_{d \ge 0} H^0(X, \mathcal{M} \otimes \mathcal{L}^{\otimes d})$ induces an equivalence of categories: $coh(X) \sim B$ -qgr.

(iii) If R defined as in part (i) and $B = B(ProjR, \mathcal{O}_{ProjR}(1))$, then $R_d = B_d$ for d >> 0. Hence we have that R-qgr ~ B-qgr.

Therefore we have that R and B in the theorem above reflect each other ringtheoretically. In fact if R is normal, then R is isomorphic to B. This discussion and an example are depicted in the figures below.



Figure 2.1: Interactions between commutative graded algebras and commutative projective geometry



Figure 2.2: Example of the discussion in §2.1.1.

2.1.2 Techniques of Artin-Tate-van den Bergh (ATV)

We now discuss techniques of Artin, Tate, and van den Bergh to generalize the results of §2.1.1 to the noncommutative setting. Let us consider a noncommutative \mathbb{N} -graded ring $A = \bigoplus_{i \in \mathbb{N}} A_i$, generated in degree 1 over $A_0 = k$. To study this ring geometrically, then we need to make sense of $\operatorname{Proj}(A)$ in the noncommutative setting.

As mentioned in Chapter 1, if $A = R_q = k\{x, y\}/(xy - qyx)$ where $q \neq 0, \sqrt{1}$, then A has only three two-sided homogeneous non-irrelevant prime ideals. Instead we consider the modules given in Figure 2.1, and associate to A a geometric object X whose closed points are in 1-1 correspondence with the following modules.

Definition II.2. A cyclic graded left A-module M with Hilbert series $H_M(t) = \sum_{i=0}^{\infty} t^i$ is called a **point module** of A.

Loosely speaking if X exists, then it is a commutative projective scheme which is referred to as the *point scheme* of A (see Definition-Lemma II.6 for the formal statement).



Figure 2.3: ATV technique: Algebra to Geometry

Since point modules are generally difficult to understand, we first parameterize the following modules.

Definition II.3. A truncated point module of A of length d is a cyclic graded left A-module M with Hilbert series $H_M(t) = \sum_{i=0}^{d-1} t^i$.

The parameterization of truncated point modules is achieved by the following method.

Step 1: Present A as T(V)/I where V is an n + 1 dimensional k-vector space, T(V) is the free algebra $k\{x_0, \ldots, x_n\}$, and I is the two-sided ideal of relations.

<u>Step 2:</u> Given a relation f of degree d, we have that $f \in I_d \subseteq V^{\otimes d} \cong ((V^*)^{\otimes d})^*$, which induces the linear functional $\tilde{f}: (V^*)^{\times d} \to k$.

Step 3: Next form truncated point schemes:

Definition II.4. The d^{th} truncated point scheme of A is

$$V_d \coloneqq \{(p_1,\ldots,p_d) \in \mathbb{P}(V^*)^{\times d} \mid \tilde{f}(p_1,\ldots,p_d) = 0, \forall f \in I_d\},\$$

which lies in $(\mathbb{P}^n)^{\times d}$. Observe that V_d is the set of zeros of multilinearizations of the degree d relations of A.

<u>Step 4:</u> Finally we have the parameterization of truncated point modules of A via the next result.

Lemma II.5. [ATVdB90, Proposition 3.9] The scheme V_d parameterizes the set of isomorphism classes of truncated point modules of length d. More precisely, V_d represents the functor of flat families of such modules.

Proof sketch. We verify the bijective correspondence. Given a point $(p_1, \ldots, p_d) \in V_d$ where $p_i = [\lambda_{0i} : \lambda_{1i} : \cdots : \lambda_{ni}]$, we can build a truncated module of length d, $M = \bigoplus_{j=0}^{d-1} k \cdot m_j$, where m_j is the generator of the j^{th} graded piece of M. Here the Aaction of M is: $x_i * m_j = \lambda_{ij} \cdot m_{j+1}$. Conversely when given such a truncated module, the scalars λ_{ij} from its A-action yield a point in V_d .

Now we can parameterize the set of isomorphism classes of point modules.

Notation. Let $\pi_d : (\mathbb{P}^n)^{\times d+1} \to (\mathbb{P}^n)^{\times d}$ be projection onto the first d coordinates of $(\mathbb{P}^n)^{\times d+1}$. Note that $V_d \subseteq (\mathbb{P}^n)^{\times d}$ and that $\pi_d(V_{d+1}) \subseteq V_d$ according to [ATVdB90, Equation (3.18)]. We also refer to the induced map $V_{d+1} \to V_d$ by π_d .

Definition-Lemma II.6. [ATVdB90, Corollary 3.13] The pairs $(V_d, \pi_d : V_{d+1} \rightarrow V_d)$ form an inverse system of schemes. Note that if one of the projections π_d is an isomorphism, then the inverse system $\{V_n\}$ is constant for $n \ge d$. Hence $\varprojlim V_n \cong V_d$ and we denote this limit by X. We call X the **point scheme** of A.

In general we refer to the family $\{(V_d, \pi_d)\}_{d\geq 0}$ as the point scheme data of A.

More generally we can consider the pro-scheme structure of $\varprojlim V_d$, though this limit may not have carry the structure of a projective scheme. Refer to [NSb] for details of its realization of as an *algebraic stack*.

Let us illustrate some examples of point scheme data.

Examples II.7. (a) Let $A = k\{x, y, z\}$, the free algebra. Since there are no relations of A, we have that V_d is $(\mathbb{P}^2)^{\times d}$. The point scheme data does not stabilize to form a projective scheme, yet note that $\varprojlim V_d \cong (\mathbb{P}^2)^{\times \infty}$. Hence the highly noncommutative algebra $k\{x, y, z\}$ is associated to the large geometric projective object $(\mathbb{P}^2)^{\times \infty}$. (b) For the other extreme, consider A = k[x, y, z], the polynomial algebra. We have that V_d is isomorphic to the diagonal $\{(p, \ldots, p) \in (\mathbb{P}^2)^{\times d} \mid p \in \mathbb{P}^2\}$. A computation of V_2 is given below. Now we see that the point scheme of A is isomorphic to \mathbb{P}^2 . (b') We explicitly show how V_2 is the set of zeros of multilinearizations of degree 2 relations of $A = k[x_0, x_1]$. To begin, note that

$$V_{2} = \{ ([\lambda_{01} : \lambda_{11}], [\lambda_{02} : \lambda_{12}]) \in \mathbb{P}^{1}_{[x_{01}:x_{11}]} \times \mathbb{P}^{1}_{[x_{02}:x_{12}]} \mid \tilde{f}([\lambda_{01} : \lambda_{11}], [\lambda_{02} : \lambda_{12}]) = 0, \ \forall f \in I_{2} \}.$$

Observe that the only relation of A of degree two is

$$f = x_0 x_1 - x_1 x_0 \subseteq V \otimes V \coloneqq (k \cdot x_{01} \oplus k \cdot x_{11}) \otimes (k \cdot x_{02} \oplus k \cdot x_{12}).$$

Thus $\tilde{f}: V^* \times V^* \to k$ is defined by $\tilde{f}(\phi) = \phi(f) \coloneqq x_{01}x_{12} - x_{11}x_{02}$. By the construction of V_2 , we have that $\lambda_{01}\lambda_{12} - \lambda_{11}\lambda_{02} = 0$. In other words, $[\lambda_{01}:\lambda_{11}] = [\lambda_{02}:\lambda_{12}]$ and $V_2 = \{(p,p) \in \mathbb{P}^1 \times \mathbb{P}^1 \mid p \in \mathbb{P}^1\}$ as desired. (c) The point scheme of a commutative ring R generated in degree one is isomorphic to $\operatorname{Proj}(R)$. Hence point schemes are genuine noncommutative analogues of commutative projective schemes.



Figure 2.4: Examples of ATV technique: Algebra to Geometry

Another piece of geometric data that we frequently employ is the following automorphism σ of the point scheme X.

Definition II.8. Assuming existence, let X be the point scheme of a connected graded k-algebra A. For a point $p \in X$, let M(p) be a corresponding A-point module. We say that an automorphism σ of X is **induced by the shift functor on point modules of** A if $M(p)[1]_{\geq 1} = M(\sigma^{-1}p)$ for all $p \in X$.

All automorphisms of point schemes in this thesis are induced by the shift functor on point modules; see [ATVdB91, §6] for more details about this notion.

Now we return from noncommutative geometry back to noncommutative algebra, under the vital condition that $\varprojlim V_d$ stabilizes as a projective scheme (to X). We can then construct the coordinate ring below. Notation. Given a projective scheme X, an invertible sheaf \mathcal{L} on X, and $\sigma \in \operatorname{Aut}(X)$, we write \mathcal{L}^{σ} for the pullback of \mathcal{L} along σ . For any open set U of X, we have that $\mathcal{L}^{\sigma}(U) = \mathcal{L}(\sigma U)$.

Definition II.9. [ATVdB90, §6] Given a point scheme X as above, let \mathcal{L} be an invertible sheaf on X, and let $\sigma \in \operatorname{Aut} X$. The **twisted homogeneous coordinate** ring $B = B(X, \mathcal{L}, \sigma)$ of X with respect to \mathcal{L} and σ is an N-graded ring:

$$B = \bigoplus_{d \in \mathbb{N}} H^0(X, \mathcal{L}_d)$$

where $\mathcal{L}_0 = \mathcal{O}_X$, $\mathcal{L}_1 = \mathcal{L}$, and $\mathcal{L}_d = \mathcal{L} \otimes_{\mathcal{O}_X} \mathcal{L}^{\sigma} \otimes_{\mathcal{O}_X} \dots \otimes_{\mathcal{O}_X} \mathcal{L}^{\sigma^{d-1}}$ for $d \ge 2$. Here multiplication is given by the natural map:

$$B_d \otimes B_e \cong H^0(X, \mathcal{L}_d) \otimes_k H^0(X, \mathcal{L}_e)$$
$$\cong H^0(X, \mathcal{L}_d) \otimes_k H^0(X, \mathcal{L}_e^{\sigma^d}) \xrightarrow{\phi} H^0(X, \mathcal{L}_{d+e}) \cong B_{d+e}$$

where ϕ is obtained by taking global sections of the isomorphism $\mathcal{L}_d \otimes_{\mathcal{O}_X} \mathcal{L}_e^{\sigma^d} \cong \mathcal{L}_{d+e}$.

Observe that when $\sigma = \mathrm{id}|_X$, then $B(X, \mathcal{L}, \sigma)$ is the commutative section ring $B(X, \mathcal{L})$ from §2.1.1 (see Figure 2.1). We illustrate in the following figure the interaction between noncommutative algebra and geometry in the fashion of Artin-Tate-van den Bergh as described above; an example is then provided.



Figure 2.5: Interactions between noncommutative algebra and noncommutative geometry

Now consider the ring $R_q = k\{x, y\}/(yx - qxy)$ where $q \neq 0, \sqrt{1}$, which is suspected to behave like a polynomial ring in two variables.

Lemma II.10. There is a 1-1 correspondence between closed points of \mathbb{P}^1 and isomorphism classes of point modules over R_q . Hence the point scheme of R_q is \mathbb{P}^1 .

Proof. First note that the Hilbert series $H_{R_q}(t)$ is equal to $(1-t)^{-2}$.

Take a closed point $[\alpha : \beta] \in \mathbb{P}^1$. Here $[\alpha : \beta]$ represents an equivalence class of tuples $(p_1, p_2) \in k^2 \setminus \{(0, 0)\}$ with $(p_1, p_2) = (\lambda \alpha, \lambda \beta)$ for some nonzero $\lambda \in k$. The point $[\alpha : \beta]$ is in bijective correspondence with the left ideal $R_q(\alpha y - \beta x)$ of R_q . Now the cyclic graded left R_q -module $M = R_q/R_q(\alpha y - \beta x)$ is an R_q -point module as

$$H_M(t) = H_{R_q}(t) - H_{R_q(\alpha y - \beta x)}(t) = H_{R_q}(t) - t H_{R_q}(t) = (1 - t)^{-1}$$

Conversely let M be a point module over R_q with generator m. Then $\operatorname{ann}_{R_q}(m) = \{r \in R_q \mid rm = 0\}$ is a left ideal of R_q , and $M \cong R_q/\operatorname{ann}_{R_q}(m)$. Since $H_M(t) = (1-t)^{-1}$ and $H_{R_q}(t) = (1-t)^{-2}$, we have that $H_{\operatorname{ann}_{R_q}(m)}(t) = t(1-t)^{-2}$. Hence $\operatorname{ann}_{R_q}(m)$ is of the form $R_q a$ where $a \in (R_q)_1$. In other words, $\operatorname{ann}_{R_q}(m) = R_q(\alpha y - \beta x)$ for some
$(\alpha, \beta) \in k^2 \setminus \{(0, 0)\}$. Moreover M is isomorphic to another R_q -point module M'with generator m' if and only if $\operatorname{ann}_{R_q}(m) = \operatorname{ann}_{R_q}(m')$. Here by the same reasoning $\operatorname{ann}_{R_q}(m') = R_q a' = R_q (\alpha' y - \beta' x)$ for some $a' \in (R_q)_1$ and $(\alpha', \beta') \in k^2 \setminus \{(0, 0)\}$. Hence $M \cong M'$ if and only if $a = \lambda a'$ for some nonzero $\lambda \in k$. Therefore the module M corresponds to the closed point $[\alpha:\beta] \in \mathbb{P}^1$.

Lemma II.11. Given the projective line \mathbb{P}^1 with invertible sheaf $\mathcal{L} = \mathcal{O}_{\mathbb{P}^1}(1)$ and automorphism $\sigma[x:y] = [x:qy]$, we have that the twisted homogeneous coordinate ring $B(\mathbb{P}^1, \mathcal{L}, \sigma)$ is isomorphic to the ring R_q .

Proof. We refer the reader to [SvdB01, Example 3.4].



Figure 2.6: Example of the discussion in §2.1.2.

Observe that the example above did not require the computation of truncated point modules; this is rarely the case.

Though tedious, constructing twisted homogeneous coordinate rings is worthwhile due to the following remarkable results of Artin-van den Bergh and Keeler. First we consider some terminology. **Definition II.12.** [AVdB90] Given a projective scheme X, an invertible sheaf \mathcal{L} on X, and $\sigma \in \operatorname{Aut}(X)$, then \mathcal{L} is **right** σ -ample if for all $\mathcal{F} \in \operatorname{coh}(X)$ and $n \gg 0$:

(a) $\mathcal{F} \otimes \mathcal{L}_n$ is generated by global sections where $\mathcal{L}_n = \mathcal{L} \otimes_{\mathcal{O}_X} \mathcal{L}^{\sigma} \otimes_{\mathcal{O}_X} \cdots \otimes_{\mathcal{O}_X} \mathcal{L}^{\sigma^{n-1}}$;

(b) $H^q(X, \mathcal{F} \otimes \mathcal{L}_n) = 0$ for all q > 0.

We say that \mathcal{L} is left σ -ample if (a) and (b) hold for the sheaf $\mathcal{F}^{\sigma^n} \otimes \mathcal{L}_n$.

Theorem II.13. [Kee00, Corollary 5.1] The right σ -ample and left σ -ample conditions are equivalent.

The concept of σ -ampleness is rather delicate [Kee00], but for the work in this thesis, we point out that invertible sheaves on irreducible curves and on projective spaces are ample if and only if they are σ -ample [Kee00, Proposition 5.6, Corollary 5.4] [Sta].

Theorem II.14. [AVdB90] (Noncommutative Serre's Theorem) If X is an irreducible projective scheme, \mathcal{L} is a σ - ample invertible sheaf, then the twisted homogeneous coordinate ring $B = B(X, \mathcal{L}, \sigma)$ is a Noetherian domain. Secondly we have an equivalence of categories: B-qgr ~ coh(X).

We can often relate our connected graded ring A to a twisted homogeneous coordinate ring B through a ring surjection $A \twoheadrightarrow B$, and in this case the nice properties of B (discussed in Theorem II.14) lift to A. This is extremely useful when the ringtheoretic behavior of A cannot be determined using purely algebraic techniques; an example of such an algebra A is provided in the next subsection.

2.1.3 Analysis of Sklyanin algebras via Artin-Tate-van den Bergh methods

Here we apply methods of Artin-Tate-van den Bergh to study three-dimensional Sklyanin algebras (Definition I.2), first computing their point schemes.. **Proposition II.15.** [ATVdB90, §4] The point scheme X (Definition-Lemma II.6) of $S = Skly_3$ is isomorphic to the object $E_{abc} \stackrel{i}{\subset} \mathbb{P}^2$ from Equation (1.2):

$$E = E_{abc} : \mathbb{V}\left((a^3 + b^3 + c^3) xyz - (abc)(x^3 + y^3 + z^3) \right) \subset \mathbb{P}^2,$$

where $E = \mathbb{P}^2$ or a cubic curve. In particular, we have the following statements. Recall that V_d denotes the d^{th} truncated point scheme of S (Definition II.4).

- (i) For $d \ge 2$, the morphism $\pi_d : V_{d+1} \to V_d$ is an isomorphism of varieties.
- (ii) The morphism $\pi_1 : V_2 \to \mathbb{P}^2$ is an isomorphism onto its image, which is the object E.
- (iii) Let $\pi'_1: V_2 \to \mathbb{P}^2$ be the morphism induced by projection onto the second copy of \mathbb{P}^2 . Then π'_2 is an isomorphism onto its image, which again is E above.
- (iv) The morphism $\sigma = \pi'_1 \circ \pi_1^{-1} : E \to E$ is an automorphism of E. The closed subvariety V_2 of $\mathbb{P}^2 \times \mathbb{P}^2$ then is isomorphic to the graph of σ .

Proof. We borrow the presentation of this proof from [Ste, Proposition 5.8.3].

(i) We must show that a point $p = (p_1, \ldots, p_d) \in V_d$ extends uniquely to a point $p' = (p_1, \ldots, p_d, p_{d+1}) \in V_{d+1}$ and that the morphism sending p to p' is a morphism of varieties. Consider multilinearizations of the three relations of $Skly_3$:

$$\begin{split} f_i &\coloneqq ay_i z_{i+1} + bz_i y_{i+1} + cx_i x_{i+1}, \\ g_i &\coloneqq az_i x_{i+1} + bx_i z_{i+1} + cy_i y_{i+1}, \\ h_i &\coloneqq ax_i y_{i+1} + by_i x_{i+1} + cz_i z_{i+1}. \end{split}$$

Note that $V_d = \mathbb{V}(f_i, g_i, h_i)_{1 \le i \le d-1} \subseteq (\mathbb{P}^2)^{\times d}$ for $d \ge 2$ with $V_1 \cong \mathbb{P}^2$. Take M_i to be the 3×3 matrix:

$$\left(egin{array}{cccc} cx_i & bz_i & ay_i \ az_i & cy_i & bx_i \ by_i & ax_i & cz_i \end{array}
ight)$$

and note that $f_i = g_i = h_i = 0$ is equivalent to:

(2.1)
$$\mathbf{M}_{i} \cdot (x_{i+1}, y_{i+1}, z_{i+1})^{T} = 0.$$

Moreover $f_i = g_i = h_i = 0$ is also equivalent to:

$$(2.2) \qquad (x_i, y_i, z_i) \cdot \mathbf{M}_{i+1} = 0.$$

To show the uniqueness of the morphism sending p to p', we must verify that the solution $(x_{d+1}, y_{d+1}, z_{d+1})$ of Equation (2.1) is projectively unique. In other words, the matrix M_d must have rank 2 at every point of V_d . Since Equation (2.2) holds for i = d - 1, we know that rank $(M_d) \leq 2$. Suppose that rank $(M_d) = 1$, i.e. every 2×2 minor of M_d vanishes. This yields the following set of equations:

$$\begin{aligned} c^{2}x_{d}y_{d} &= abz_{d}^{2}, \quad b^{2}y_{d}z_{d} = acx_{d}^{2}, \quad a^{2}x_{d}z_{d} = bcy_{d}^{2}, \\ a^{2}y_{d}z_{d} &= bcx_{d}^{2}, \quad c^{2}x_{d}z_{d} = aby_{d}^{2}, \quad b^{2}x_{d}y_{d} = acz_{d}^{2}, \\ b^{2}x_{d}z_{d} &= acy_{d}^{2}, \quad a^{2}x_{d}y_{d} = bcz_{d}^{2}, \quad c^{2}y_{d}z_{d} = abx_{d}^{2}. \end{aligned}$$

We leave it to the reader to show that these equations imply that the point [a:b:c]is in the set \mathfrak{D} of Definition I.2, a contradiction. Thus rank $(M_d) = 2$ for every point of V_d . Moreover since the coordinates of p_{d+1} are given locally by 2×2 minors of M_d , we have that π_d is an isomorphism of varieties.

(ii) The image of $\pi_1 : V_2 \to \mathbb{P}^2$ is the set of all points $[x_1 : y_1 : z_1] \in \mathbb{P}^2$ so that there exists $[x_2 : y_2 : z_2] \in \mathbb{P}^2$ with Equation (2.1) satisfied for i = 1. This is equivalent to

$$\{[x_1:y_1:z_1] \in \mathbb{P}^2 \mid \det(M_1) = 0\}.$$

Hence $\operatorname{im}(\pi_1)$ is the closed subset $E_{abc} \subseteq \mathbb{P}^2$ above. Furthermore π_1 is an isomorphism onto E as $\operatorname{rank}(M_1)=2$ for every point $[x_1:y_1:z_1] \in E$. The function $\pi_1^{-1}: E \to V_2$ is a morphism of varieties as it is defined locally by the 2×2 minors of M_1 . (iii) This is symmetric to the proof of (ii).

(iv) [ATVdB90, Proposition 3.7] By the definition of σ , we have that

$$V_2 = \{ (p, \sigma(p)) \mid p \in E \} \subseteq \mathbb{P}^2 \times \mathbb{P}^2$$

Thus V_2 is isomorphic to the graph of σ .

Note that $E = \mathbb{P}^2$ if one of a, b, c equals 0 and the sum of the other two parameters equals 0; E is an elliptic curve if $abc \neq 0$ and $(a^3 + b^3 + c^3)^3 \neq (3abc)^3$; and E is a singular cubic curve otherwise.

Next for E smooth, we form the twisted homogeneous coordinate ring $B = B(E, i^*\mathcal{O}_{\mathbb{P}^2}(1), \sigma)$ where σ is given in Proposition II.15. We have by Theorem II.14 that B is a Noetherian domain. Moreover if E is a smooth cubic curve, then $\dim_k B_d = 3d$ for $d \ge 1$ [ATVdB90, Theorem 6.6(ii)]. If $E = \mathbb{P}^2$, then $\dim_k B_d = \dim_k [x, y, z]_d$ for $d \ge 0$. To yield results about $Skly_3$, we employ the following result of Artin-Tate-van den Bergh.

Theorem II.16. [ATVdB90, Theorem 6.8] Assume that the point scheme E_{abc} of S(a,b,c) is smooth. If $E = \mathbb{P}^2$, then S(a,b,c) is isomorphic to a twisted homogeneous coordinate ring $B(\mathbb{P}^2, \mathcal{L}, \sigma)$. If E is an elliptic curve, then there is a ring surjection from S(a,b,c) to the twisted homogeneous coordinate ring $B = B(E_{abc}, i^*\mathcal{O}_{\mathbb{P}^2}(1), \sigma)$, with kernel generated by a degree 3 central, regular element (often denoted by g).

Remark II.17. In fact, the element g is explicitly given in [AS87, (10.17)]. It was computed with Affine, a noncommutative package for the computer algebra system Maxima. For S(a, b, c), we have that:

$$g = c(c^{3} - b^{3})y^{3} + b(c^{3} - a^{3})yxz + a(b^{3} - c^{3})xyz + c(a^{3} - c^{3})x^{3}.$$

Now we have the result below, which forms the heart of [ATVdB90].

Corollary II.18. [ATVdB90, Theorem 8.1][RZ08, proof of Corollary 4.6(2)] The three-dimensional Sklyanin algebras Skly₃, with a smooth point scheme E_{abc} , are Noetherian domains of polynomial growth; indeed their Hilbert series are $H_S(t) = (1-t)^{-3}$.



Figure 2.7: ATV technique applied to $Skly_3$

2.1.4 Linear noncommutative geometry of higher dimensions

We introduced in §2.1.2 the notion of a point module and of a point scheme associated to a connected graded k-algebra A. Likewise we can define higher dimensional noncommutative geometric objects corresponding to A, objects which are particularly useful in Chapter 5.

Definition II.19. A cyclic graded left A-module M with Hilbert series

 $H_M(t) = (1-t)^{-(d+1)}$ is called a *d*-linear module. In particular, *d*-linear modules with d = 0, 1, 2 are point modules, line modules, and plane modules respectively.

We can also discuss the parameterizations of isomorphism classes of such modules; these are referred to as *d*-linear schemes. We point out that the d = 1 schemes, or rather **line schemes**, have been studied in several articles including [ATVdB91, LBSVdB96, LS93, SV02a, SV02b].

Continuing with our analysis of three-dimensional Sklyanin algebras, we describe their line and plane schemes below.

Example II.20. [ATVdB91] The line modules of $S = Skly_3$ are precisely modules of the form S/Sv where v is a nonzero element of S_1 . These modules are in bijective correspondence with lines $l = \mathbb{V}(v) \subseteq \mathbb{P}^2 = \mathbb{P}(S_1)$. Thus the line scheme of $Skly_3$ is isomorphic to the dual projective space $(\mathbb{P}^2)^*$.

Moreover if M is a plane module of $Skly_3$, then M must be isomorphic to the left S-module S. Therefore the plane scheme of $Skly_3$ is simply a point.

2.1.5 Noncommutative projective schemes

In this section, we briefly discuss the concept of a noncommutative projective scheme as introduced in [AZ94]. Such structures are prompted by the commutative setting; we employ Serre's theorem (Theorem II.1) illustrate this motivation.

Recall that one can associate to any commutative graded algebra R a pair (X, \mathcal{O}_X) , where $X=\operatorname{Proj}(R)$ is the projective scheme introduced in §2.1.1 and \mathcal{O}_X is its structure sheaf. The first part of Serre's theorem states that we have an equivalence of categories: $\operatorname{coh}(X) \sim R$ -qgr. Here the structure sheaf $\mathcal{O}_{\operatorname{Proj}R}$ corresponds to the object $\pi(RR)$, where π is the quotient functor $\pi: R$ -gr $\rightarrow R$ -qgr. Hence the commutative projective scheme $\operatorname{Proj}(R)$ corresponds the pair (R-qgr, $\pi(RR))$.

On the other hand, the second part of Serre's theorem states that the category $\operatorname{coh}(X)$ is equivalent to *B*-qgr, where $B = B(X, \mathcal{L})$ is the section ring where $X = \operatorname{Proj}(R)$ and \mathcal{L} an ample invertible sheaf on X. In particular if $\mathcal{L} = \mathcal{O}_X(1)$, then \mathcal{L} corresponds to the image of the shifted module $\pi(BB[1])$ under this equivalence.

Moreover the final part of Serre's result pertains to the *polarized variety* $(X, \mathcal{O}_X(1))$; namely that we have an equivalence of categories, R-qgr ~ $B(X, \mathcal{O}_X(1))$ -qgr, provided that R is also generated in degree one.

We now present Artin and Zhang's scheme construction pertinent to the (noetherian) noncommutative setting.

Definition II.21. [AZ94, §2] Given a graded left-noetherian (not necessarily commutative) k-algebra A, we say that a **noncommutative projective scheme** is the triple:

$$(A - \operatorname{qgr}, \pi(A), s_A).$$

Here s_A is the autoequivalence on A-qgr induced by the shift functor $M \mapsto M[1]$ in A-gr.

For future reference, we will use this construction in §2.3 and §3.5.

2.2 Gelfand-Kirillov dimension

We introduce in this section the notion of Gelfand-Kirillov dimension, a growth measure of algebras and modules in terms of a generating set. The main reference for this theory is a text of Krause and Lenagen [KL00], yet we refer to various sources for the discussion below.

Notation. Unless stated otherwise, let A be a finitely generated, locally finite, \mathbb{N} graded, connected k-algebra. Take $M \in A$ -gr, and put $A_{\leq n} \coloneqq \bigoplus_{i=0}^{n} A_i$ and $M_{\leq n} \coloneqq \bigoplus_{i=0}^{n} M_i$.

Definition II.22. [MR01, 8.1.6, 8.1.11] Given finitely generated $M \in A$ -gr, define a function $f_M : \mathbb{Z} \to \mathbb{N}$ by $f_M(d) \coloneqq \sum_{n \leq d} \dim_k M_n$.

1. The **Gelfand-Kirillov dimension** of M, denoted $\operatorname{GKdim}(M)$ is defined to be

$$s = \overline{\lim}_d \frac{\log f_M(d)}{\log d}.$$

Equivalently, we have that:

 $\operatorname{GKdim}(M) = \liminf \{ \alpha \mid \exists c \ge 0 \text{ so that } \dim_k(M_{\le n}) \le cn^{\alpha} \text{ for all } n >> 0 \}.$

2. A module M is *s*-critical if $\operatorname{GKdim}(M) = s$ and all of its proper quotients have strictly smaller Gelfand-Kirillov dimension.

Remark II.23. In the thesis, all graded algebras A under consideration have Hilbert series $H_A(t) = (1-t)^{-r}$ for some positive integer r. Hence graded A-modules M have Hilbert series of the form $H_M(t) = q_M(t)(1-t)^{-r}$ for some $q_M(t) \in \mathbb{Z}[t, t^{-1}]$ [ATVdB91, (2.18, 2.19)]. Moreover GKdim(M) is the order of the pole at t = 1 of $H_M(t)$, to say $H_M(t) = g_M(t)(1-t)^{-\text{GKdim}(M)}$, for some $g_M(t) \in \mathbb{Z}[t, t^{-1}]$.

Another useful invariant of the module $M \in A - \text{gr}$ is given below.

Definition II.24. The multiplicity of M is defined to be

$$\operatorname{mult}(M) \coloneqq g_M(1) = [(1-t)^{\operatorname{GKdim}(M)} H_M(t)]|_{t=1} \in \mathbb{N}$$

Example II.25. We have that the *d*-linear modules (Definition II.19) are examples of modules with multiplicity 1 and GK-dimension d + 1.

Likewise we can define Gelfand-Kirillov dimension for a graded algebra A by $\operatorname{GKdim}(A) = \operatorname{GKdim}(_A A)$. In other words, we have:

Definition II.26. Let A be a finitely generated \mathbb{N} -graded connected k-algebra. Then the **Gelfand-Kirillov dimension of the algebra** A is

 $\operatorname{GKdim}(A) = \liminf \{ \alpha \mid \exists c \ge 0 \text{ so that } \dim_k(A_{\le n}) \le cn^{\alpha} \text{ for all } n >> 0 \}.$

For example, the GK-dimension of the commutative polynomial ring $k[x_1, \ldots, x_n]$ is n, whereas the GK-dimension of the free algebra $k\{x_1, \ldots, x_r\}$ is ∞ for $r \ge 2$. We also consider the following properties presented in the next two propositions.

Proposition II.27. [MR01, 8.1.17, 8.1.18, 8.2.7] [KL00, Chapter 7] [Sta] Let A be a finitely generated, locally finite, \mathbb{N} -graded, k-algebra. Take $M \in A$ -mod. We have that the following facts hold.

- 1. If A is commutative, then GKdim(A) equals the degree of the Hilbert-Samuel polynomial.
- 2. If A is not commutative, then GKdim(A) need not be an integer.
- 3. GKdim(M) = 0 if and only if M is finite-dimensional.
- 4. If GKdim(M) < 1, then GKdim(M) = 0.
- 5. If A is also a connected graded domain, then

 $GKdim(A) = 1 + \liminf\{\beta \mid \exists c \ge 0 \text{ so that } \dim_k(A_n) \le cn^\beta \text{ for all } n >> 0\}.$

6. GKdim(A[t]) = GKdim(A) + 1

Proposition II.28. [KL00, Proposition 3.15] Let I be an ideal of a k-algebra A and assume that I contains a (left or right) regular element of A. Then

$$GKdim(A/I) \leq GKdim(A) - 1.$$

Note that $\operatorname{GKdim}(A) < \infty$ if and only if A has polynomial bounded growth. We now introduce terms pertaining to the case that $\operatorname{GKdim}(A) = \infty$.

Definition II.29. [SZ97] Let A be a graded locally finite k-algebra. We say that A has sub-exponential growth if $\overline{\lim}_d f_A(d)^{1/d} \leq 1$, and has exponential growth if $\overline{\lim}_d f_A(d)^{1/d} > 1$.

So if A has polynomial growth, then it has sub-exponential growth. However the converse does not hold.

These notions of growth slightly differ from the standard definitions (say as presented in [KL00]), yet the advantage of the [SZ97] version is that it is used to study the ring-theoretic behavior of A. Namely we have the following result of Stephenson and Zhang, which we use throughout this thesis.

Theorem II.30. [SZ97, Theorem 0.1] If A is a graded locally finite left noetherian k-algebra, then A has sub-exponential growth.

2.3 Representation theory via noncommutative projective algebraic geometry

In this section, we provide the background information for the study of the representation theory of certain noncommutative algebras U that are not necessarily graded. We are particularly interested in classifying simple finite-dimensional Umodules, a set denoted by $\operatorname{Simp}_{<\infty} U$. We refer to the set of finite-dimensional U-modules by $\operatorname{Repr}_{<\infty} U$. The idea is to interpret U as analogous to the coordinate ring of an affine open subset of a projective scheme $\operatorname{Proj}(A)$. Classifying $\operatorname{Simp}_{<\infty} A$ is also of interest.

Recall from Definition II.22 the notion of a critical module. We now present an association between irreducible finite-dimensional representations of A and 1-critical A-modules. We also require the following terminology.

Definition II.31. Let $A = \bigoplus_{i \in \mathbb{N}} A_i$ be a connected graded k-algebra. The ideal $A_+ := \bigoplus_{i \ge 1} A_i$ is referred to as the **irrelevant ideal** (or as the **augmentation ideal**). Moreover the module $A/A_+ \cong_A k$ is called the **trivial module**.

Lemma II.32. [SS93, Lemma 4.1] Let A be a connected graded k-algebra generated in degree one that is noetherian and locally finite $(\dim_k A_i < \infty)$. Assume that $GKdim(A) \ge 1$. Now if M is a nontrivial finite-dimensional simple A-module, then M is a quotient of some 1-critical graded A-module.

Proof. We repeat the proof presented in [SS93]. Let M be a nontrivial simple finitedimensional left S-module. Define a new left S-module: $\tilde{M} := M \otimes_k k[t]$, with $s \in S_d$ acting by $s * (m \otimes t^i) = (sm) \otimes t^{i+d}$. Then \tilde{M} is a graded S-module with $\tilde{M}_d = M \otimes (k \cdot t^d)$. By considering the map $\pi : \tilde{M} \to M$ given by $\pi(m \otimes t^d) = m$, we see that M is isomorphic to the factor $\tilde{M}/\tilde{M}(t-1)$. Hence M is the quotient of a graded module \tilde{M} , a module which has GK-dimension 1 (Proposition II.27) and multiplicity equal to $\dim_k M$.

Since there is a filtration

$$\tilde{M} = \tilde{M}^0 \supseteq \tilde{M}^1 \supseteq \tilde{M}^2 \supseteq \cdots \supseteq \tilde{M}^k = 0$$

by graded submodules so that $\tilde{M}^i/\tilde{M}^{i+1}$ is critical, we have that M is a quotient of one of these factors. Thus M is the quotient of a 1-critical graded A-module.

Remark II.33. [SS93, Remarks after Lemma 4.1] Say a nontrivial module M in Simp_{<∞}S is the quotient of some 1-critical graded module N. In other words, there is an A-module map $\psi : N \to M$. Then there exists a unique degree 0 A-module map $\tilde{\psi} : N \to \tilde{M}$ such that $\psi = \pi \circ \tilde{\psi}$. This map $\tilde{\psi}$ is given by $\tilde{\psi}(n) = \psi(n) \otimes t^d$ for $n \in N_d$. Now N[k] embeds into \tilde{M} for some $k \in \mathbb{Z}$, and $\operatorname{mult}(N) \leq \operatorname{mult}(\tilde{M}) = \dim_k M$.

The rest of section is dedicated to understanding 1-critical graded A-modules, which is used for the study of $\operatorname{Simp}_{<\infty} A[c^{-1}]_0$ for some central element c of A. Here $A[c^{-1}]_0$ is the algebra U mentioned at the beginning of this section. Moreover we will see that the class of 1-critical graded A-modules is associated to the set of irreducible objects of the category A-qgr.

Notation. We denote by Irred(A-qgr), the subcategory of A-qgr whose objects consist of irreducible objects of A-qgr.

2.3.1 Commutative motivation

To motivate §2.3.2, we illustrate the geometry of 1-critical graded R-modules for R a commutative connected graded ring generated in degree one. Of course if R is commutative, then the finite-dimensional representations of R are 1-dimensional, so this setting is rather different from the noncommutative notions discussed later.

Lemma II.34. Let R be a commutative graded k-algebra, generated in degree one over $R_0 = k$. We have an equivalence of categories whose objects are respectively:

- (a) 1-critical graded R-modules;
- (b) irreducible objects of R-qgr;
- (c) R-point modules.

Proof. We remind the reader that the field k is algebraically closed. The equivalence of categories (a) and (c) is discussed in [Smi94, §6], whereas the equivalence between categories (b) and (c) can be understood geometrically via Serre's theorem (Theorem II.1). Namely $\operatorname{Irred}(R\operatorname{-qgr}) \sim \operatorname{Irred}(\operatorname{coh}(X))$, where the latter corresponds to closed points of $X=\operatorname{Proj}(R)$.

Thus by Lemma II.32 when R is commutative, all simple finite-dimensional Rmodules arise as simple quotients of point modules.

However when given a noncommutative algebra A, the category A-qgr is generally not equivalent to coh(X) for some projective scheme X. Thus it is useful to consider factors of A for which this scenario occurs. We depict such a scenario for the commutative setting as follows.

Let f be a regular element of R, homogeneous of positive degree. Consider the decomposition of the projective scheme $\operatorname{Proj}(R) =: X$ into a union of a closed subscheme $\operatorname{Proj}(R/(f))$ and its open complement X_f . We aim to investigate this union categorically. First consider $\mathbb{V}(f) = \operatorname{Proj}(R/(f))$, a closed subscheme of X. We have by Theorem II.1 an equivalence of categories:

(2.3)
$$\operatorname{coh}(\mathbb{V}(f)) \sim \operatorname{Irred}(R/(f) - \operatorname{qgr}) \sim \operatorname{Irred}(R - \operatorname{qgr})_{f \text{-torsion}}$$

In other words, the closed subscheme $\mathbb{V}(f)$ of X is thought of categorically as the subcategory $\operatorname{Irred}(R\operatorname{-qgr})$ of $\operatorname{Irred}(R\operatorname{-qgr})$.

On the other hand, consider the complement X_f of $\mathbb{V}(f)$ in X:

$$X_f = \{ p \in X \mid f(p) \neq 0 \},\$$

or rather the affine scheme $\operatorname{Spec}(R[f^{-1}]_0)$. Instead of considering coherent sheaves on X_f , we use the equivalence of 'complementary' categories below:

(2.4)
$$\operatorname{Irred}(R\operatorname{-qgr}) \sim \operatorname{Simp}_{<\infty} R[f^{-1}]_0.$$

This equivalence is induced by the localization map: R-qgr $\longrightarrow R[f^{-1}]_0$ -mod, given by $[M] \mapsto M[f^{-1}]_0$ [ATVdB91, §7]. Hence the open complement X_f of $\mathbb{V}(f)$ in Xis viewed categorically as the category Irred(R-qgr).

We summarize the decompositions discussed above in the following figure. The last diagram, in particular, references the equivalences in (2.3) and (2.4), and Lemmas II.32 and II.34.



Figure 2.8: (Category equivalences of) the decomposition of R-qgr

2.3.2 Decomposing noncommutative projective schemes (or categories -qgr)

Consider Figure 2.8 with R = A, an arbitrary noncommutative connected graded algebra generated in degree 1. Recall that our goal is not only understand the class of finite-dimensional simple A-modules $\operatorname{Simp}_{\infty} A$, but to also $\operatorname{classify} \operatorname{Simp}_{<\infty} U$ for some ungraded algebra U prompted by the affine geometry of $\operatorname{Proj}(A)$. From the discussion of §2.1.2, we know that it is undesirable to use $\operatorname{Proj}(A)$, so instead we generalize the relationship between the last two diagrams of Figure 2.8 to the noncommutative setting. In particular, we employ a noncommutative version of Serre's Theorem (Theorem II.14).

Considering the quotient functor $\pi : A$ -gr $\rightarrow A$ -qgr, we now study the representatives $F \in A$ -gr of irreducible objects in A-qgr.

Proposition II.35. [Smi94, Proposition 7.1] Every irreducible object in A-qgr has a representative $F \in A$ -gr so that:

$$1. \ F = \bigoplus_{i \ge 0} F_i;$$

- 2. F is generated by F_0 ;
- 3. $\dim_k F_i$ is constant for all *i*, and this constant is equal to mult(F);
- 4. F has no nonzero finite dimensional submodule.

Hence this prompts the following terminology for the graded A-modules with images in Irred(A-qgr).

Definition II.36. A cyclic graded left *A*-module *M*, generated in degree 0, of GKdimension 1 and multiplicity ϵ is called a **point module** if $\epsilon = 1$, and a **fat point module** if $\epsilon > 1$.

We now use point modules and fat point modules of A to yield results about $\operatorname{Simp}_{<\infty}(A)$. To achieve a decomposition similar to Figure 2.8, first take a central regular element c of A, homogeneous of positive degree. If we are given a module $M \in A$ -qgr that is c-torsion, then $M \in A/(c)$ -qgr. Otherwise by [ATVdB91, Proposition 7.5], we have that M yields an object in $\operatorname{Repr}_{<\infty}A[c^{-1}]_0$. Hence an irreducible object of A-qgr is either an irreducible object of the category A/(c)-qgr or it corresponds to a finite-dimensional simple module over the ungraded algebra $A[c^{-1}]_0$.

A nice scenario occurs when (in addition to the hypotheses on A and c above) the factor A/(c) is isomorphic to a twisted homogeneous coordinate ring $B(\mathcal{P}_A, \mathcal{L}, \sigma)$, where \mathcal{P}_A is the point scheme of A, and \mathcal{L} is a σ -ample invertible sheaf of \mathcal{P}_A . Morally we say that the noncommutative projective scheme (A-qgr, $\pi(_AA)$, s_A) (Definition II.21) decomposes into the "closed subscheme" (B-qgr, $\pi(_BB)$, s_B), and its open complement corresponding to objects of $\operatorname{Simp}_{<\infty} A[c^{-1}]_0$. Such a notion is the noncommutative analogue of the third diagram of Figure 2.8.

In fact we understand the category, Irred(B-qgr), by Noncommutative Serre's theorem (Theorem II.14); the irreducible objects of B-qgr correspond to B-point

modules. Hence by the comments after Definition II.36, finite-dimensional simple modules over $A[c^{-1}]_0$ correspond to fat point modules over A. In other words, classifying fat point modules of A produces results about the representation theory of the (ungraded) algebra $A[c^{-1}]_0$. We will see later that this can be achieved by studying the representation theory of the graded algebras A and A/(c), techniques which are particularly useful in Chapter 4 where we study *deformed Sklyanin algebras* (see Example 3 in §2.3.3).

2.3.3 Examples

We provide examples of the discussion in the previous subsection. Namely we illustrate the decomposition of a noncommutative scheme (A-qgr, $\pi(A)$, s_A) into a closed subscheme and open complement, and to yield results in representation theory.

(1) Let A be the homogenized Weyl algebra,

$$k\{x, y, w\}/(xy - yx - w^2, xw - wx, yw - wy),$$

and let c = w. Let k is an algebraically closed field of characteristic zero. Then A/(w) = k[x, y], a commutative polynomial ring. Moreover $A[w^{-1}]_0$ is isomorphic to the first Weyl algebra $W = k\{x, y\}/(xy - yx - 1)$. Since the Weyl algebra has no irreducible finite-dimensional representations, then all (nontrivial) simple finite-dimensional A-modules arise as simple quotients of 1-critical modules (or point modules in this case) over k[x, y]. These can be explicitly computed.

(2) Let A be $Skly_3(a, b, c)$ with $abc \neq 0$ and $(a^3 + b^3 + c^3)^3 \neq (3abc)^3$, and let c = gbe the degree 3 central element from Remark II.17. Then S/(g) is isomorphic to a twisted homogeneous coordinate ring $B(E, \mathcal{L}, \sigma)$ for E an elliptic curve. Moreover $S[g^{-1}]_0$ is a simple ring if $|\sigma| = \infty$, and Azumaya (Definition II.46) if $|\sigma| < \infty$ [ATVdB91, Proposition 7.3]. Thus one can apply results such as Theorem II.51 to understand simple finite-dimensional modules over $S[g^{-1}]_0$. Further work about the sets $\operatorname{Simp}_{<\infty}S$, $\operatorname{Simp}_{<\infty}B$, and $\operatorname{Simp}_{<\infty}S[g^{-1}]_0$ is presented in §4.3.

(3) In Chapter 4 we will define a large class of *deformed Sklyanin algebra*, S_{def} , for which this discussion of §2.3.2 is particularly useful. The algebras have the following properties. Let A = D, a *central extension* of $Skly_3$ (Definition IV.6), and let c = w. Then $A/(w) = Skly_3$ and $A[w^{-1}]_0 = S_{def}$ (Definition IV.2).

2.4 Polynomial identity (PI) rings

Polynomial identity (PI) rings have been of interest to algebraists and geometers since the 1920s; refer to [Ami74] for a historical account of these structures. This section will review basic definitions, properties, and examples of PI rings as presented in [MR01, Chapter 13]. Special subclasses of PI rings are also discussed in §§2.4.2-2.4.3.

2.4.1 Basic properties of PI rings

Definition II.37. A polynomial identity (PI) ring is a ring A for which there exists a monic multilinear polynomial $f \in \mathbb{Z}\{x_1, \ldots, x_n\}$ so that $f(a_1, \ldots, a_n) = 0$ for all $a_i \in A$. The minimal degree of such a polynomial is referred as the **minimal degree** of A.

Proposition II.38. (1) Any ring that is module-finite over its center is PI.
(2) Any subring or homomorphic image of a PI ring is PI.

Examples II.39. (1) Commutative rings R are PI as the elements satisfy the polynomial identity f(x, y) = xy - yx for all $x, y \in R$.

(2) The ring of $n \times n$ matrices over a commutative ring R, $Mat_n(R)$, is PI.

(3) The ring $A = k\{x, y\}/(xy + yx)$ is PI as A is a rank 4 module over its center $Z(A) = k[x^2, y^2]$.

In fact if A is PI, then $Mat_n(A)$ is also PI for all $n \ge 1$.

2.4.2 On central simple algebras (CSAs)

Now we study a nice subset of PI rings: central simple algebras (CSAs); details can be found in [MR01, §13.3].

Definition II.40. A ring is a **central simple algebra (CSA)** if its simple, artinian, and a finite-dimensional module over its center. Hence by Wedderburn's theorem, $A \cong \operatorname{Mat}_n(D)$ for some division ring D, where D is module-finite over its center.



Figure 2.9: The class of PI rings contains the subclass of CSAs

Provided a CSA, we can explicitly measure the ring's noncommutativity. This prompts the notion of the PI degree.

Lemma II.41. [MR01, Corollary 13.3.5] Let $A = Mat_n(D)$ be a CSA with Z(D) = C. Then $\dim_C D = m^2$ and $\dim_C A = (mn)^2 =: p^2$ for some $m, n \in \mathbb{N}$.

Considering the theorem above, we have that 2p is the minimal degree of A. The value p is called the **PI degree of a CSA** A. Moreover, the value p^2 is the rank of A over its center.

The notion of PI degree can be extended to the class of prime PI rings via Posner's theorem. This is discussed in [MR01, 13.6.5] and we restate their remarks here.

Theorem II.42. (Posner) Let A be a prime PI ring with center Z and with minimal degree 2p. Let $S = Z \setminus \{0\}$, let $Q = AS^{-1}$, and let $F = ZS^{-1}$ denote the quotient field of Z. Then Q is a CSA with center F and dim_F $Q = p^2$.

Considering the notation above, the **PI degree of a prime PI ring** A is equal to $(\dim_F Q)^{1/2}$, where Q is the *Goldie quotient ring* of A.

Examples II.43. (1) Let R be a commutative ring. Then R has minimal degree 2, has PI degree 1, and is rank 1 over its center.

(2) Let R be a commutative ring. Then $Mat_n(R)$ has minimal degree 2n, has PI degree n, and is rank n^2 over its center.

(3) The ring $A = k\{x, y\}/(xy + yx)$ has minimal degree 4, has PI degree 2, and is rank 4 over its center.

Now provided that an extra condition holds, a PI ring is in fact a CSA.

Definition II.44. A ring A is said to be (left) primitive if A has a faithful simple module M. In other words, M is simple with the condition that aM = 0 only if a = 0 for any $a \in A$.

Theorem II.45. [MR01, Theorem 13.3.8] (Kaplansky) If A is a primitive PI ring of minimal degree d, then A is a CSA of rank $(d/2)^2$ over its center.

2.4.3 On Azumaya algebras

We introduce Azumaya algebras, an extension of the class of central simple algebras (CSAs) which is a well behaved subclass of PI rings. A further discussion of these rings is found in [MR01, §13.7].

Definition II.46. Let A be a ring with center C, and let $E := \text{End}A_C$. Then A is **Azumaya** over C if

- (i) A_C is finitely generated and projective, and
- (ii) $A \otimes_C A^{op} \longrightarrow E$ is an isomorphism.

Note that the definition above is rarely used in practice. Instead we may want to think of an Azumaya algebra as a CSA over C, where C is a commutative ring that is not necessarily a field. Here are some basic properties.

Proposition II.47. (1) If A is Azumaya, then A is PI.

- (2) If A is a CSA, then A is Azumaya.
- (3) If R is a commutative ring, then $Mat_n(R)$ is Azumaya.



Figure 2.10: The class of PI rings contains Azumaya algebras, which in turn contains CSAs

Again these rings are closely tied to their centers.

Proposition II.48. [MR01, Proposition 13.7.9] Say A is Azumaya with center Z(A). Then there is a 1-1 correspondence between the set of ideals of A and the set of ideals of Z(A). This is given by the maps: $I \mapsto I \cap Z(A)$ and $HA \leftrightarrow H$, respectively.

The above result is useful from the standpoint of noncommutative geometry, as in this setting: $\operatorname{Spec}(A) = \operatorname{Spec}(Z(A)).$

Corollary II.49. If A is Azumaya, then the following are equivalent:

- (i) A has the a.c.c. on ideals;
- (ii) A is right and left Noetherian; and

(iii) Z(A) is Noetherian.

We end with a couple of structure theorems on (the representations of) prime Azumaya algebras.

Theorem II.50. [MR01, Theorem 13.7.14] (Artin-Procesi) Let A be a prime ring. Then the following are equivalent

- (i) A is Azumaya of rank n^2 ;
- (ii) A is a prime PI ring so that for all primes \mathfrak{p} : $PIdeg(A/\mathfrak{p}) = PIdeg(A)$, with PIdeg(A) = n.

Pertaining to Chapter 4 in particular, it is worth pointing out that prime Azumaya algebras are nice in terms of representation theory due to the following result of Artin.

Theorem II.51. [Art69] Let A be a prime Azumaya algebra. Then the dimension of a simple finite-dimensional A-module is equal to the PI degree of A.

We can apply this discussion to arbitrary PI algebras provided we restrict our attention to the *Azumaya locus* as we define below.

Proposition II.52. [BG97, Proposition 3.1] Let A be a prime noetherian ring that is module finite over its center Z. Assume that Z is an affine algebra over an algebraically closed field k. Then the following statements hold.

(a) The maximum k-vector space dimension of simple finite-dimensional A-modules is the PI degree of A.

(b) Let M be a simple A-module, let P denote $ann_A M$, and let $\mathfrak{m} = P \cap Z$. Then $\dim_k M = PIdeg A$ if and only if $A_\mathfrak{m}$ is an Azumaya over $Z_\mathfrak{m}$. Here $A_\mathfrak{m}$ is the localization $A \otimes_Z Z_\mathfrak{m}$.

The condition in part (b) prompts the following definition.

Definition II.53. Assume that the hypotheses of Proposition II.52 hold. Consider the sets:

 $\mathcal{A}_A = \{ \mathfrak{m} \in \operatorname{maxspec} Z \mid A_\mathfrak{m} \text{ is Azumaya over } Z_\mathfrak{m} \},$ $\mathcal{S}_A = \{ \mathfrak{m} \in \operatorname{maxspec} Z \mid \operatorname{gldim}(Z_\mathfrak{m}) = \infty \}.$

These are called the **Azumaya locus** and **singular locus** of A respectively. The ideals in the Azumaya locus are referred to as **Azumaya points**. Moreover the complement of S_A in maxspecZ is referred to as the **smooth locus**.

Thus when A is Azumaya, we have that $\mathcal{A}_A = \text{maxspec}Z$ and we recover the result of Theorem II.51. More generally if the hypotheses of Proposition II.52 hold, then we get that:

Proposition II.54. [BG02, §III.1] \mathcal{A}_A is an open, Zariski dense subset of maxspecZ.

Moreover we have that \mathcal{A}_A and \mathcal{S}_A are related in the following manner.

Lemma II.55. [BG97, Lemma 3.3] Assume the hypotheses of Proposition II.52. If $gldim(A) < \infty$, then $\mathcal{A}_A \subseteq maxspecZ \smallsetminus \mathcal{S}_A$.

Equality can actually be achieved, to say the Azumaya and smooth loci of A can coincide, provided additional conditions on A hold [BG97, Theorem 3.8].

Therefore we see that the representation theory of noncommutative PI k-algebras can be analyzed geometrically.

2.5 Bergman's Diamond Lemma

This section is dedicated to results of Bergman, which construct a k-vector space basis for a given k-algebra A. The details for the following work is found in [Ber78], and the presentation of this material is borrowed from [Sta].

Notation. Given a k-algebra A = T(V)/I where V is a finite-dimensional k-vector space, pick an ordered basis $x_1 < x_2 < \cdots < x_n$ for V. We form a basis of T(V) by considering the set of all monomials in $\{x_i\}$ under the lexicographical ordering on $\langle V \rangle$. Let I be generated as an ideal by $\{s_i\}$ where (after nonzero scalar multiplication) $s_i = w_i - f_i$, so that w_i is a monic monomial $\rangle f_i$.

We introduce a way of simplifying elements of A by using the following maps.

Definition II.56. Given monomials $m, n \in V >$, we have maps:

$$\sigma :< V > \longrightarrow T(V)$$
, given by $\mathbf{m}w_i \mathbf{n} \mapsto \mathbf{m}f_i \mathbf{n}$.

This extends by linearity to: $T(V) \longrightarrow T(V)$; products of these maps are called **reductions**.

Definition-Lemma II.57. For all $t \in T(V)$, we have that t is reduction finite in the sense that for reductions σ_i , we have that: $\sigma_s \cdots \sigma_1(t) = \sigma_{s+1} \sigma_s \cdots \sigma_1(t), \forall s >> 0.$

Now considering reductions, we define the following terms of T(V).

Definition II.58. (a) We say that $t \in T(V)$ is **irreducible** if $\sigma(t) = t$ for all reductions σ . Let $T(V)_{irred}$ denote the set of irreducible elements of T(V). This is clearly spanned by $\langle V \rangle_{irred}$.

(b) We say $t \in T(V)$ is **reduction unique** if for all sequences of reductions from t to an irreducible element, then we get the same element. In this case, write this unique element as $\sigma_{\infty}(t)$.

Example II.59. Consider the k-algebra $A = k\{x, y\}/(yx - xy - x^2)$.

(a) Take y > x so w = yx and $f = xy + x^2$. Then we see that the set of irreducible monomials of A are of the form $\{x^i y^j\}_{i,j \in \mathbb{N}}$.

(b) Take x > y so $w = x^2$ and f = xy - yx. The element x^3 is not reduction unique due to the following computations:

$$x^{3} = x^{2} \cdot x = (xy - yx)x = xyx - y(x^{2})$$
$$= xyx - y(xy - yx) = xyx - yxy + y^{2}x$$

and

$$x^{3} = x \cdot x^{2} = x(xy - yx) = (x^{2})y - xyx$$
$$= (xy - yx)y - xyx = xy^{2} - yxy - xyx$$

Therefore $2xyx + y^2x - xy^2 = 0$ is a *hidden relation*. Continue by including this hidden relation in the ideal of relations I of A and repeat reductions for the other monomials that are potentially not reduction unique. For a given degree, this process eventually terminates and we get a basis of irreducible monomials of the form

$$\{x^{\epsilon} \cdot y^{i_1} \cdot x \cdot y^{i_2} \cdot x \cdot y^{i_3} \cdots\}$$

for $i_j \in \mathbb{N}, \ \epsilon = 0, 1$.

We now examine conditions where reduction uniqueness fails.

Definition II.60. (a) Say that there exist monomials $\mathbf{m}, \mathbf{n}, \mathbf{p} \in V$ so that $\mathbf{mn} = w_1$ and $\mathbf{np} = w_2$ for some relations $s_1 = w_1 - f_1$ and $s_2 = w_2 - f_2$. This is called an **overlap ambiguity**. It is *resolvable* (we are able to find σ_{∞}) provided that $f_1\mathbf{p}$ and $\mathbf{m}f_2$ are reduction unique with $\sigma_{\infty}(f_1\mathbf{p}) = \sigma_{\infty}(\mathbf{m}f_2)$.

(b) Say that there exist monomials $\mathbf{m}, \mathbf{n}, \mathbf{p} \in V$ so that $\mathbf{mnp} = w_1$ and $\mathbf{n} = w_2$ for some relations $s_1 = w_1 - f_1$ and $s_2 = w_2 - f_2$. This is called an **inclusion ambiguity**. It is *resolvable* provided that $\sigma_{\infty}(f_1) = \sigma_{\infty}(\mathbf{m}f_2\mathbf{p})$.



Figure 2.11: Resolution of ambiguities

The following theorem provides us with the precise conditions for which we can write down an irreducible basis of monomials for the algebra A = T(V)/I. If A is graded, then the theorem yields an algorithm for computing such a basis.

Theorem II.61. [Ber78, Theorem 1.2] (Bergman's Diamond Lemma) The following are equivalent:

(1) A = T(V)/I has a k-vector space basis of monomials $\langle V \rangle_{irred}$,

(2) All elements of T(V) are reduction unique,

(3) All ambiguities of A are resolvable.

Below are two examples of Diamond Lemma computations that are required in Chapter 3.

Example II.62. Consider the algebra S = S(1, b, c) in Lemma III.8. Namely we have that $b^3 = c^3 = 1$ and

$$S = k\{x, y, z\} / (yz + bzy + cx^2, zx + bxz + cy^2, xy + byx + cz^2).$$

Take the ordering x < y < z. We first show that all ambiguities of S are resolvable. Secondly we compute a basis of irreducible monomials of S. Observe that $b^{-1} = b^2$ and $c^{-1} = c^2$. Now rewrite the relations of S as follows:

$$z^{2} = -bc^{2}yx - c^{2}xy,$$

$$zy = -b^{2}yz - b^{2}cx^{2},$$

$$zx = -bxz - cy^{2}.$$

There are three (overlap) ambiguities that we must resolve:

$$\begin{aligned} z^3 &= z^2 \cdot z = z \cdot z^2, \\ z^2 y &= z^2 \cdot y = z \cdot zy, \\ z^2 x &= z^2 \cdot x = z \cdot zx. \end{aligned}$$

Here are the computations for the first ambiguity:

$$z^2 \cdot z = (-bc^2yx - c^2xy) \cdot z = -bc^2yxz - c^2xyz$$

and

$$z \cdot z^{2} = z \cdot (-bc^{2}yx - c^{2}xy) = -bc^{2}(zy)x - c^{2}(zx)y$$
$$= c^{2}y(zx) + x^{3} + bc^{2}x(zy) + y^{3}$$
$$= -bc^{2}yxz - y^{3} + x^{3} - c^{2}xyz - x^{3} + y^{3}$$
$$= -bc^{2}yxz - c^{2}xyz.$$

Hence the ambiguity z^3 is resolvable.

Likewise $z^2 \cdot y = -bc^2yxy - c^2xy^2$ and

$$z \cdot zy = z \cdot (-b^2yz - b^2cx^2) = -b^2(zy)z - b^2c(zx)x$$

= $by(z^2) + bcx^2z + cx(zx) + b^2c^2y^2x$
= $-b^2c^2y^2x - bc^2yxy + bcx^2z - bcx^2z - c^2xy^2 + b^2c^2y^2x$
= $-bc^2yxy - c^2xy^2$.

Thus z^2y is resolvable. Similarly, we have that $z^2 \cdot x = z \cdot zx = -bc^2yx^2 - c^2xyx$.

Now by Theorem II.61, we can then conclude that S has a k-vector space of monomials $\langle V \rangle_{irred}$. Moreover we have that $\langle V \rangle_{irred}$ equals the set of monomials in $k\{x, y\} \oplus k\{x, y\} \cdot z$.

Example II.63. Given the dual $S^!$ of S in Example II.62, we show that all ambiguities are resolvable. Here $S^! = k\{u, v, w\}/(R)$ where R is the set of relations:

$$\{w^2 = cuv, wv = b^2vw, wu = buw, vw = c^2u^2, v^2 = bcuw, vu = b^2uv\}.$$

Take the ordering u < v < w, and recall that $b^3 = c^3 = 1$. There are twelve (overlap) ambiguities, which by the following calculations are resolvable:

$$\begin{split} w^{3} &= w^{2} \cdot w = w \cdot w^{2} = cu^{3}, & vw^{2} = vw \cdot w = v \cdot w^{2} = c^{2}u^{2}w, \\ w^{2}v &= w^{2} \cdot v = w \cdot wv = bc^{2}u^{2}w, & vwv = vw \cdot v = v \cdot wv = c^{2}u^{2}v, \\ w^{2}u &= w^{2} \cdot u = w \cdot wu = b^{2}cu^{2}v, & vwu = vw \cdot u = v \cdot wu = c^{2}u^{3}, \\ wvw &= wv \cdot w = w \cdot vw = b^{2}c^{2}u^{2}w, & v^{2}w = v^{2} \cdot w = v \cdot vw = bc^{2}u^{2}v, \\ wv^{2} &= wv \cdot v = w \cdot v^{2} = b^{2}c^{2}u^{2}v, & v^{3} = v^{2} \cdot v = v \cdot v^{2} = u^{3}, \\ wvu &= wv \cdot u = w \cdot vu = b^{2}c^{2}u^{3}, & v^{2}u = v^{2} \cdot u = v \cdot vu = b^{2}cu^{2}w. \end{split}$$

By Theorem II.61, S' has $\langle V \rangle_{irred}$ as a k-vector space basis of monomials, and this is precisely the set of the monomials in $k[u] \oplus k[u] \cdot v \oplus k[u] \cdot w$.

CHAPTER III

Degenerate Sklyanin algebras and generalized twisted homogeneous coordinate rings

A vital development in the field of noncommutative projective algebraic geometry is the investigation of connected graded noncommutative rings A with use of geometric data. In particular Artin-Tate-van den Bergh constructed twisted homogeneous coordinate rings B (Definition II.9), whose role is analogous to section rings in classical algebraic geometry. Often B is a well behaved ring and the algebra Awill share its properties.

For example consider $A = Skly_3$, the three-dimensional Sklyanin algebra for which we recall the definition for the reader's convenience.

Definition III.1. The **three-dimensional Sklyanin algebras**, denoted by S(a, b, c) or $Skly_3$, are generated by three noncommuting variables x, y, z, subject to three relations:

$$ayz + bzy + cx^{2} = 0$$

$$azx + bxz + cy^{2} = 0$$

$$axy + byx + cz^{2} = 0$$

for $[a:b:c] \in \mathbb{P}^2_k \smallsetminus \mathfrak{D}$ where

 $\mathfrak{D} = \{ [0:0:1], [0:1:0], [1:0:0] \} \cup \{ [a:b:c] \mid a^3 = b^3 = c^3 = 1 \}.$

The geometric data associated to $Skly_3$ is E = a cubic curve or \mathbb{P}^2 (Proposition II.15), from which we can build a twisted homogeneous coordinate ring B(E). Moreover there exists a ring surjection from $Skly_3$ onto B(E) with kernel generated by a regular, central element, homogeneous of degree 3 (Theorem II.16) in the case that E is smooth. Furthermore in this case, the behavior of $Skly_3$ reflects that of Bring-theoretically (Corollary II.18).

During the last few years, a number of examples of (even noetherian) algebras have appeared for which the techniques of Artin-Tate-van den Bergh are inapplicable. This is because the point modules for these algebras cannot by parameterized by a projective scheme of finite type (see for example [KRS05]). Consequently one cannot form a corresponding twisted homogeneous coordinate ring. In this chapter, we explore a recipe suggested in [ATVdB90, §3] for building a generalized twisted homogeneous coordinate ring for *any* connected graded ring. Note that such results appear in [Wal09]. In particular, we provide a geometric approach to examine degenerations of Sklyanin algebras.

Definition III.2. The rings S(a, b, c) from Definition III.1 with $[a : b : c] \in \mathfrak{D}$ are called **degenerate Sklyanin algebras**, denoted by S_{deg} .

We will see that the geometry of S_{deg} involves a degenerate cubic curve. On the other hand we establish algebraic properties of S_{deg} such as growth, dimensions, and other ring-theoretic characteristics in §3.1. These are listed in Theorem I.6.

Motivated by the methods of [ATVdB90], we make use of point modules over S_{deg} to construct the generalized coordinate ring. To do so, we first generate geometric data for S_{deg} in §3.2. We remind the reader of the definitions of (truncated) point modules in Definitions II.3 and II.2.

Theorem III.3. (Proposition III.26) For $d \ge 2$, the truncated point schemes $V_d \subset (\mathbb{P}^2)^{\times d}$ corresponding to S_{deg} are isomorphic to either:

(i) a union of three copies of (P¹)^{×^d/₂} and three copies of (P¹)^{×^d/₂}, if d is odd, or;
(ii) a union of six copies of (P¹)^{×^d/₂}, if d is even.

Observe that the point scheme of a degenerate Sklyanin algebra does not stabilize to produce a projective scheme of finite type. Hence we cannot mimic the approach of [ATVdB90] to construct a twisted homogeneous coordinate ring associated to S_{deg} . Instead we form a graded ring dependent on the truncated point schemes V_d of S_{deg} . Such a ring is called a *point parameter ring* $P(S_{deg})$ (Definition III.29), which have not appeared in the literature since their introduction in [ATVdB91]. The next results of §5.3 describe the behavior of the first non-trivial examples of these rings.

Theorem III.4. (Proposition III.31, Theorem III.36, Corollaries III.40 and III.41) The point parameter ring $P = P(S_{deg})$ is generated in degree one. Since $(S_{deg})_1 \cong (P(S_{deg}))_1$, we have that P is a factor of the corresponding S_{deg} . It has Hilbert series $H_P(t) = (1+t^2)(1+2t)[(1-2t^2)(1-t)]^{-1}$. Moreover P has exponential growth and is neither right noetherian, Koszul, nor a domain.

Thus Theorem III.4 yields a result surprisingly similar to Theorem II.16 (pertaining to $Skly_3 \twoheadrightarrow B(E)$), despite the fact that the rings $Skly_3$ and S_{deg} are entirely different as we will see in §3.1.

Moreover with the results of §3.1, we see that the behavior of $P(S_{deg})$ resembles that of S_{deg} and it is natural to ask if other noncommutative algebras can be analyzed in a similar fashion. Some further directions are discussed in §3.4; those particularly pertaining to the field of coherent noncommutative algebraic geometry are mentioned in §3.5.

3.1 Structure of degenerate Sklyanin algebras

In this section, we establish algebraic properties of degenerate Sklyanin algebras. We begin by considering the degenerate Sklyanin algebras $S(a, b, c)_{deg}$ with $a^3 = b^3 = c^3 = 1$ (Definition III.2) and the following definitions from [GW04].

Definition III.5. Let α be an endomorphism of a ring R. An α -derivation on R is any additive map $\delta : R \to R$ so that $\delta(rs) = \alpha(r)\delta(s) + \delta(r)s$ for all $r, s \in R$. The set of α -derivations of R is denoted α -Der(R).

We write $S = R[z; \alpha, \delta]$ provided S is isomorphic to the polynomial ring R[z] as a left R-module but with multiplication given by $zr = \alpha(r)z + \delta(r)$ for all $r \in R$. Such a ring S is called an Ore extension of R. More precisely:

Definition III.6. Let R be a ring, α a ring endomorphism of R, and δ an α derivation on R. We shall write $S = R[z; \alpha, \delta]$ provided

- (a) S is a ring, containing R as a subring;
- (b) z is an element of S;
- (c) S is a free left R-module with basis $\{1, z, z^2, \dots\}$;
- (d) $zr = \alpha(r)z + \delta(r)$ for all $r \in R$.

Such a ring S is called an **Ore extension of** R.

By generalizing the work of [BS] we see that most degenerate Sklyanin algebras are factors of Ore extensions of the free algebra on two variables. Note that if $a \neq 1$, then S(a, b, c) is isomorphic to $S\left(1, \frac{b}{a}, \frac{c}{a}\right)$. Hence whenever we consider such an S, we assume that a = 1. **Proposition III.7.** Let $a, b, c \in k$ so that a = 1 and $b^3 = c^3 = 1$, thus $[a : b : c] \in \mathfrak{D}$.

Then we get the ring isomorphism

(3.2)
$$S(1,b,c) \cong \frac{k\{x,y\}[z;\alpha,\delta]}{(\Omega)}$$

where

(a)
$$\alpha \in End(k\{x, y\})$$
 is defined by $\alpha(x) = -bx$, $\alpha(y) = -b^2y$,

- (b) $\delta \in \alpha$ -Der(k{x,y}) is given by $\delta(x) = -cy^2$, $\delta(y) = -b^2cx^2$,
- (c) $\Omega = xy + byx + cz^2$, which is a normal element of $k\{x, y\}[z, \alpha, \delta]$.

Proof. By direct computation α and δ are indeed an endomorphism and α -derivation of $k\{x, y\}$ respectively. Moreover $x \cdot \Omega = \Omega \cdot bx$, $y \cdot \Omega = \Omega \cdot by$, $z \cdot \Omega = \Omega \cdot z$ so Ω is a normal element of the Ore extension. Thus both rings of (3.2) have the same generators and relations.

Lemma III.8. The Hilbert series of S_{deg} is $H_{S_{deg}}(t) = (1+t)(1-2t)^{-1}$.

Proof. One can find a basis of irreducible monomials via Bergman's Diamond lemma (Theorem II.61) to imply that $\dim_k S_d = 2^{d-1}3$ for $d \ge 1$. For S(1, b, c), we have shown in Example II.62 that S is free with a basis $\{1, z\}$ as a left or right module over $k\{x, y\}$.

Now consider the monomial algebra $S(1,0,0) = k\{x,y,z\}/(yz,zx,xy)$. An irreducible word of S is precisely constructed by following the variable x with either x or z, by following the variable y with either x or y, and by following the variable z with either y or z. Here again $\dim_k S_d = 2^{d-1}3$ for $d \ge 1$. Similar results hold for the other two degenerate Sklyanin algebras that are monomial algebras.

Therefore in each case the algebra S_{deg} has Hilbert series,

$$H_{S_{deg}}(t) = 1 + \sum_{d \ge 1} 2^{d-1} \cdot 3 \cdot t^d = 1 + 3t \sum_{d \ge 0} (2t)^d = 1 + \frac{3t}{1 - 2t} = \frac{1 + t}{1 - 2t}.$$

Therefore by Proposition III.7 (for $a^3 = b^3 = c^3 = 1$) or Lemma III.8 (in general) we have the following consequence. Recall the notions of growth introduced in Chapter 2, section 2.

Corollary III.9. The degenerate Sklyanin algebras have exponential growth, infinite GK dimension, and are not left noetherian. Furthermore S_{deg} is not a domain.

Proof. For all values $[a:b:c] \in \mathfrak{D}$, we have by Lemma III.8 that

$$r_d \coloneqq \sum_{i \le d} \dim_k S_i = \sum_{i \le d} 2^{i-1} 3 = 3(2^{d+1} - 1).$$

Since $\overline{\lim}_d r_d^{1/d} = 2 > 1$, the algebras S_{deg} have exponential growth. Thus $\operatorname{GKdim}(S_{deg}) = \infty$ and so by Theorem II.30, S_{deg} is also not noetherian.

Now if $[a : b : c] \in \{[1 : 0 : 0], [0 : 1 : 0], [0 : 0 : 1]\}$, then the monomial algebra S(a, b, c) is obviously not a domain. On the other hand if [a : b : c] satisfies $a^3 = b^3 = c^3 = 1$, then assume without loss of generality that a = 1. As a result we have

$$f_1 + bf_2 + cf_3 = (x + by + bc^2 z)(cx + cy + b^2 z),$$

where $f_1 = yz + bzy + cx^2$, $f_2 = zx + bxz + cy^2$, and $f_3 = xy + byx + cz^2$ are the relations of S(1, b, c). Hence S is not a domain.

Now we verify homological properties of degenerate Sklyanin algebras.

Definition III.10. Let A be a connected graded algebra which is locally finite $(\dim_k A_i < \infty)$. We say that A is **Koszul** if its trivial module $A/\bigoplus_{i\geq 1} A_i \cong {}_Ak$ admits a minimal resolution P_{\bullet}^{\min} that is linear. In other words, the entries of the matrices that determine P_{\bullet}^{\min} belong to A_1 . Refer to [Kra, §4] for a discussion of basic properties and examples of such algebras.

Proposition III.11. The degenerate Sklyanin algebras are Koszul with infinite global dimension.

Proof. For S = S(a, b, c) with $a^3 = b^3 = c^3 = 1$, assume that a = 1 and consider the description of S in Proposition III.7. Since $k\{x, y\}$ is Koszul, the Ore extension $k\{x, y\}[z, \alpha, \delta]$ is also Koszul [CS08, Definition 1.1, Theorem 10.2]. By Proposition III.7, the element Ω is normal and regular in $k\{x, y\}[z; \alpha, \delta]$. Hence the factor S is Koszul by [ST01, Theorem 1.2].

To conclude $gl.dim(S) = \infty$, note that the Koszul dual of S is $S! = k\{u, v, w\}/(R)$ where R is the set of relations:

$$\{w^2 = cuv, wv = b^2vw, wu = buw, vw = c^2u^2, v^2 = bcuw, vu = b^2uv\}$$

Taking the ordering u < v < w, we have shown in Example II.63 that $S^!$ has a basis of irreducible monomials $\{u^i, u^j v, u^k w\}_{i,j,k \in \mathbb{N}}$. Hence $S^!$ is not a finite dimensional *k*-vector space and by [Kra, Corollary 5], *S* has infinite global dimension.

For S = S(a, b, c) with $[a : b : c] \in \{[1 : 0 : 0], [0 : 1 : 0], [0 : 0 : 1]\}$, note that S is Koszul as its ideal of relations is generated by quadratic monomials [PP05, Corollary 4.3]. Denote these monomials m_1 , m_2 , m_3 . The Koszul dual of S in this case is $S' = k\{u, v, w\}/(R)$ where (R) is the ideal of relations generated by the six monomials not equal to m_i (in variables u, v, w). Since S' is again a monomial algebra, it contains no hidden relations and has a nice basis of irreducible monomials. In particular, S' contains $\bigoplus_{i\geq 0} kp_i$ where p_i is the length i word:

$$p_{i} = \begin{cases} \underbrace{uvwuvwu\dots}_{i}, & \text{if } [a:b:c] = [1:0:0] \\ \underbrace{uwvuwvu\dots}_{i}, & \text{if } [a:b:c] = [0:1:0] \\ u^{i}, & \text{if } [a:b:c] = [0:0:1]. \end{cases}$$

Therefore $S^!$ is not a finite dimensional k-vector space. By [Kra, Corollary 5] these degenerate Sklyanin algebras have infinite global dimension.

We compute the center of the degenerate Sklyanin algebras.

Proposition III.12. The center of S_{deg} is equal to k.

Proof. For $a^3 = b^3 = c^3 = 1$, we assume that a = 1 as $S(a, b, c) \cong S(1, \frac{b}{a}, \frac{c}{a})$. By Example II.62, an element f of S(1, b, c) can be written uniquely as $f = t_1 + t_2 z$ with $t_i \in k\{x, y\}$. Suppose that f is central. Then fx = xf implies that $t_1x + t_2zx = xt_1 + xt_2z$. Since $zx + bxz + cy^2 = 0$ is a relation of S(1, b, c), we have that

$$t_1x + t_2(-bxz - cy^2) = xt_1 + xt_2z.$$

Considering the coefficients of z, we get that

$$t_1 x - ct_2 y^2 = xt_1$$
$$-bt_2 x = xt_2.$$

Therefore $t_2 = 0$ and $t_1 \in k[x]$. Likewise by using fy = yf, we conclude that $t_2 = 0$ and $t_1 \in k[y]$. Hence $f \in k[x] \cap k[y] = k$.

Now take the monomial algebra $S(1,0,0) = k\{x,y,z\}/(yz,zx,xy)$ whose irreducible basis of monomials is described in the proof of Lemma III.8. Take a central element f of S(1,0,0). Now fx = xf implies that either $f \in k[x]$ or $f = \sum \gamma_i m_i$ where $\gamma_i \in k$ and m_i is a monomial of the form $y \cdots z$. Likewise fy = yf implies that $f \in k[y]$ or $f = \sum \gamma_i n_i$ where n_i is a monomial of the form $z \cdots x$. Finally fz = zf yields $f \in k[z]$ or $f = \sum \gamma_i p_i$ where p_i is a monomial of the form $x \cdots y$. The simultaneous occurrence of the three outcomes above implies that $f \in k$. Thus Z(S(1,0,0)) = k. With similar reasoning, we conclude that Z(S(0,1,0)) = Z(S(0,0,1)) = k.
3.2 Truncated point schemes of S_{deg}

The goal of this section is to prove Theorem III.3, i.e. to construct the family of truncated point schemes $\{V_d \subseteq (\mathbb{P}^2)^{\times d}\}$ associated to the degenerate three-dimensional Sklyanin algebras S_{deg} (see Definition III.2). These schemes will be used in §3.3 for the construction of a generalized twisted homogeneous coordinate ring, namely the point parameter ring (Definition III.29).

Recall the definitions of truncated point modules of length d and point modules given respectively in Definitions II.3 and II.2. We will see in Lemma III.15 that the family $\{V_d\}$ plays a role in the study of point modules over S_{deg} . First we note the result below which is a special case of [ATVdB90, Corollary 3.13]. This result describes the relationship the isomorphism classes of truncated point modules and of point modules over S(a, b, c) for any $[a:b:c] \in \mathbb{P}^2$.

Lemma III.13. Let S = S(a, b, c) for any $[a : b : c] \in \mathbb{P}^2$. Denote by Γ the set of isomorphism classes of point modules over S and Γ_d the set of isomorphism classes of truncated point modules $M = \bigoplus_{i=0}^{d} M_i$ of length d + 1. With respect to the truncation function $\rho_d : \Gamma_d \to \Gamma_{d-1}$ given by $M \mapsto M/M_d$, we have that as a set Γ is the projective limit of $\{\Gamma_d\}$.

Now we proceed to construct schemes V_d that will parameterize length d truncated point modules. We remind the reader that length d truncated point schemes, V_d , were defined in Definition II.4. We now present the definition explicitly for the algebras S(a, b, c).

Definition III.14. [ATVdB90, §3] The truncated point scheme of length d, $V_d \subseteq (\mathbb{P}^2)^{\times d}$ for S(a, b, c), is the scheme defined by the multilinearizations of relations of S(a, b, c) from Definition III.1. More precisely $V_d = \mathbb{V}(f_i, g_i, h_i)_{0 \le i \le d-2}$ where

(3.3)

$$f_{i} = ay_{i+1}z_{i} + bz_{i+1}y_{i} + cx_{i+1}x_{i}$$

$$g_{i} = az_{i+1}x_{i} + bx_{i+1}z_{i} + cy_{i+1}y_{i}$$

$$h_{i} = ax_{i+1}y_{i} + by_{i+1}x_{i} + cz_{i+1}z_{i}.$$

For example, $V_1 = \mathbb{V}(0) \subseteq \mathbb{P}^2$ so $V_1 = \mathbb{P}^2$. Similarly, $V_2 = \mathbb{V}(f_0, g_0, h_0) \subseteq \mathbb{P}^2 \times \mathbb{P}^2$.

Lemma III.15. The set Γ_d is parameterized by the scheme V_d .

Proof. This is a special case of [ATVdB90, Proposition 3.9]. \Box

In short, to understand point modules over S(a, b, c) for any $[a : b : c] \in \mathbb{P}^2$, Lemmas III.13 and III.15 imply that we can now restrict our attention to truncated point schemes V_d . We point out another useful result pertaining to V_d associated to S(a, b, c) for any $[a : b : c] \in \mathbb{P}^2$.

Lemma III.16. The truncated point scheme V_d lies in d copies of $E \subseteq \mathbb{P}^2$ where E is the cubic curve $E: (a^3 + b^3 + c^3)xyz - (abc)(x^3 + y^3 + z^3) = 0.$

Proof. Let p_i denote the point $[x_i : y_i : z_i] \in \mathbb{P}^2$ and

(3.4)
$$\mathbb{M}_{abc,i} \coloneqq \begin{pmatrix} cx_i & az_i & by_i \\ bz_i & cy_i & ax_i \\ ay_i & bx_i & cz_i \end{pmatrix} \in \mathrm{Mat}_3(kx_i \oplus ky_i \oplus kz_i).$$

The point $p = (p_0, p_1, \dots, p_{d-1}) \in V_d \subseteq (\mathbb{P}^2)^{\times d}$ must satisfy $f_i = g_i = h_i = 0$ for $0 \le i \le d-2$ by definition of V_d . In other words, one is given $\mathbb{M}_{abc,j} \cdot (x_{j+1} \ y_{j+1} \ z_{j+1})^T = 0$ or equivalently $(x_j \ y_j \ z_j) \cdot \mathbb{M}_{abc,j+1} = 0$ for $0 \le j \le d-2$. Therefore for $0 \le j \le d-1$, $\det(\mathbb{M}_{abc,j}) = 0$. This implies that $p_j \in E$ for each j. Thus $p \in E^{\times d}$. \Box

3.2.1 On the truncated point schemes of some S_{deg}

We will show that to study the truncated point schemes V_d of degenerate Sklyanin algebras, it suffices to understand the schemes of specific four degenerate Sklyanin algebras. We use the following notion of a Zhang twist.

Definition III.17. Given a \mathbb{Z} -graded k-algebra $S = \bigoplus_{n \in \mathbb{Z}} S_n$ with graded automorphism σ of degree 0 on S, we form a **Zhang twist** S^{σ} of S by preserving the same additive structure on S, and defining multiplication * as follows: $a * b = ab^{\sigma^n}$ for $a \in S_n$.

The following is a special case of [Zha96, Theorem 1.2].

Theorem III.18. If S is connected graded and generated in degree one, then so is the algebra S^{σ} . We also have an equivalence of categories: S- $Gr \sim S^{\sigma}$ -Gr.

Realize \mathfrak{D} from Definition III.1 as the union of three point sets Z_i :

$$Z_{1} := \{ [1:1:1], [1:\zeta:\zeta^{2}], [1:\zeta^{2}:\zeta] \},$$

$$Z_{2} := \{ [1:1:\zeta], [1:\zeta:1], [1:\zeta^{2}:\zeta^{2}] \},$$

$$Z_{3} := \{ [1:\zeta:\zeta], [1:1:\zeta^{2}], [1:\zeta^{2}:1] \},$$

$$Z_{0} := \{ [1:0:0], [0:1:0], [0:0:1] \}.$$

where $\zeta = e^{2\pi i/3}$. Pick respective representatives [1:1:1], $[1:1:\zeta]$, $[1:\zeta:\zeta]$, and [1:0:0] of Z_1 , Z_2 , Z_3 , and Z_0 .

Lemma III.19. Every degenerate Sklyanin algebra is a Zhang twist of one the following algebras: S(1,1,1), $S(1,1,\zeta)$, $S(1,\zeta,\zeta)$, and S(1,0,0).

Proof. Consider the following graded automorphisms of the degenerate Sklyanin algebras S(a, b, c):

$$\sigma: \{x \mapsto \zeta x, \ y \mapsto \zeta^2 y, \ z \mapsto z\} \text{ and } \tau: \{x \mapsto y, \ y \mapsto z, \ z \mapsto x\}.$$

Now a routine computation shows that σ and τ yield the Zhang twists:

$$S(1,1,1)^{\sigma} = S(1,\zeta,\zeta^{2}), \qquad S(1,1,1)^{\sigma^{-1}} = S(1,\zeta^{2},\zeta) \qquad \text{for } Z_{1};$$

$$S(1,1,\zeta)^{\sigma} = S(1,\zeta,1), \qquad S(1,1,\zeta)^{\sigma^{-1}} = S(1,\zeta^{2},\zeta^{2}) \qquad \text{for } Z_{2};$$

$$S(1,\zeta,\zeta)^{\sigma} = S(1,\zeta^{2},1), \qquad S(1,\zeta,\zeta)^{\sigma^{-1}} = S(1,1,\zeta^{2}) \qquad \text{for } Z_{3};$$

$$S(1,0,0)^{\tau} = S(0,1,0), \qquad S(1,0,0)^{\tau^{-1}} = S(0,0,1) \qquad \text{for } Z_{0}.$$

For instance, consider the Zhang twist $S(1,1,1)^{\sigma}$. Now it has three quadratic relations of the form:

{
$$yz + \alpha zy + \beta x^2$$
, $zx + \alpha xz + \beta y^2$, $xy + \alpha yx + \beta z^2$ },

where $\alpha, \beta \in k$. Note that $yz + \alpha zy + \beta x^2$ is a relation of $S(1,1,1)^{\sigma}$ if and only if $y \cdot z^{\sigma} + \alpha z \cdot y^{\sigma} + \beta x \cdot x^{\sigma}$ is a relation of S(1,1,1). This is equivalent to $yz + \alpha \zeta^2 zy + \beta \zeta x^2$ equal to zero in S(1,1,1), which implies $\alpha = \zeta$ and $\beta = \zeta^2$. When we consider the other two relations of $S(1,1,1)^{\sigma}$, we also get that $\alpha = \zeta$ and $\beta = \zeta^2$. Thus $S(1,1,1)^{\sigma} = S(1,\zeta,\zeta^2)$.

By Theorem III.18, it suffices to study a representative of each of the four classes of degenerate three-dimensional Sklyanin algebras.

Correction: The point scheme of S(1,1,1) is actually

much larger than what is computed here. Please see the corrigendum to "Degenerate Sklyanin algebras and Generalized Twisted Homogeneous Coordinate rings" J. Alg (2009) We now compute the truncated point schemes of S(1, 1, 1) in detail. Calculations for the other three representative degenerate Sklyanin algebras, $S(1, 1, \zeta)$, $S(1, \zeta, \zeta)$, S(1, 0, 0), will follow with similar reasoning. We first discuss how to build a truncated point module M' of length d, when provided with a truncated point module M of length d-1.

3.2.2 Computation of V_d for S(1,1,1)

Let us explore the correspondence between truncated point modules and truncated point schemes. When given a truncated point module $M = \bigoplus_{i=0}^{d-1} M_i \in \Gamma_{d-1}$, multiplication from S = S(a, b, c) is determined by a point $p = (p_0, \ldots, p_{d-2}) \in V_{d-1}$ (Definition III.14, Equation (3.4)) in the following manner. As M is cyclic, M_i has basis say $\{m_i\}$. Furthermore for $x, y, z \in S$ with

 $p_i = [x_i : y_i : z_i] \in \mathbb{P}^2$, we get the left S-action on m_i determined by p_i :

(3.6)

$$x \cdot m_i = x_i m_{i+1}, \quad x \cdot m_{d-1} = 0;$$

 $y \cdot m_i = y_i m_{i+1}, \quad y \cdot m_{d-1} = 0;$
 $z \cdot m_i = z_i m_{i+1}, \quad z \cdot m_{d-1} = 0.$

Conversely given a point $p = (p_0, \ldots, p_{d-2}) \in V_{d-1}$, one can build a module $M \in \Gamma_{d-1}$ unique up to isomorphism by reversing the above process. We summarize this discussion in the following remark.

Remark III.20. Refer to notation from Lemma III.13. To construct $M' \in \Gamma_d$ from $M \in \Gamma_{d-1}$ associated to $p \in V_{d-1}$, we require $p_{d-1} \in \mathbb{P}^2$ such that $p' = (p, p_{d-1}) \in V_d$.

Now we begin to study the behavior of truncated point modules over S_{deg} through the examination of truncated point schemes in the next two lemmas.

Lemma III.21. Let $p = (p_0, \ldots, p_{d-2}) \in V_{d-1}$ with $p_{d-2} \notin \mathfrak{D}$ (refer to (3.5)). Then for each *i*, there exists a unique $p_{d-1} \in Z_i$ so that $p' := (p, p_{d-1}) \in V_d$.

Proof. For Z_1 , we study the representative algebra S(1, 1, 1). If such a p_{d-1} exists, then $f_{d-2} = g_{d-2} = h_{d-2} = 0$ so we would have

$$\mathbb{M}_{111,d-2} \cdot (x_{d-1} \ y_{d-1} \ z_{d-1})^T = 0$$

(Definition III.14, Equation (3.4)). Now we have that $\operatorname{rank}(\mathbb{M}_{111,d-2}) \leq 2$ as

$$(x_{d-3} \ y_{d-3} \ z_{d-3}) \cdot \mathbb{M}_{111,d-2} = 0.$$

By an argument in the proof of Proposition II.15, we see that if $\operatorname{rank}(\mathbb{M}_{111,d-2}) = 1$, then $p_{d-2} \in \mathfrak{D}$. Thus $\operatorname{rank}(\mathbb{M}_{111,d-2}) = 2$ when $p_{d-2} \notin \mathfrak{D}$ and the tuple $(x_{d-1}, y_{d-1}, z_{d-1})$ is unique up to scalar multiple. Thus the point p_{d-1} is projectively unique. To verify the existence of p_{d-1} , we solve the system of equations

$$f_{d-2}: y_{d-1}z_{d-2} + z_{d-1}y_{d-2} + x_{d-1}x_{d-2} = 0,$$

$$g_{d-2}: z_{d-1}x_{d-2} + x_{d-1}z_{d-2} + y_{d-1}y_{d-2} = 0,$$

$$(3.7) \qquad h_{d-2}: x_{d-1}y_{d-2} + y_{d-1}x_{d-2} + z_{d-1}z_{d-2} = 0,$$

$$x_{d-2}^3 + y_{d-2}^3 + z_{d-2}^3 = 3x_{d-2}y_{d-2}z_{d-2},$$

$$x_{d-1}^3 + y_{d-1}^3 + z_{d-1}^3 = 3x_{d-1}y_{d-1}z_{d-1}.$$

Here, the last two equations come from the fact that $p_{d-2}, p_{d-1} \in E$ (Lemma III.16).

Furthermore with $\zeta = e^{2\pi i/3}$, the curve $E = E_{111}$ is the union of three projective lines:

(3.8)
$$\mathbb{P}^1_A : x = -(y+z), \ \mathbb{P}^1_B : x = -(\zeta^2 y + \zeta z), \ \mathbb{P}^1_C : x = -(\zeta y + \zeta^2 z)$$



Figure 3.1: The curve $E = E_{111} \subseteq \mathbb{P}^2$: $x^3 + y^3 + z^3 - 3xyz = 0$

Using the algebra software Maple, we have the following three solutions to (3.7) with $(p_{d-2}, p_{d-1}) \in (E \setminus \mathfrak{D}) \times E$:

$$\left\{\begin{array}{l} (p_{d-2} \in \mathbb{P}^{1}_{A}, \ [1:1:1]), \\ (p_{d-2} \in \mathbb{P}^{1}_{B}, \ [1:\zeta:\zeta^{2}]), \\ (p_{d-2} \in \mathbb{P}^{1}_{C}, \ [1:\zeta^{2}:\zeta]) \end{array}\right\}.$$

Thus when $p_{d-2} \notin \mathfrak{D}$, there exists an unique point $p_{d-1} \in \mathbb{Z}_1$ so that

$$(p_0,\ldots,p_{d-2},p_{d-1})\in V_d.$$

Now consider the following notation.

$$S(1,1,\zeta) \qquad S(1,\zeta,\zeta)$$

$$\mathbb{P}_D^1 \quad x = -(y+\zeta^2 z) \qquad x = -(y+\zeta z)$$

$$\mathbb{P}_E^1 \quad x = -(\zeta^2 y+z) \qquad x = -(\zeta y+z)$$

$$\mathbb{P}_F^1 \quad x = -\zeta(y+z) \qquad x = -\zeta^2(y+z)$$

For $S(1,1,\zeta)$, the uniqueness of p_{d-1} still holds when $p_{d-2} \notin \mathfrak{D}$ and the following are the solutions $(p_{d-2}, p_{d-1}) \in (E \setminus \mathfrak{D}) \times E$ satisfying $f_{d-2} = g_{d-2} = h_{d-2} = 0$:

$$\left\{ \begin{array}{l} (p_{d-2} \in \mathbb{P}_D^1, \ [1:1:\zeta]), \\ (p_{d-2} \in \mathbb{P}_E^1, \ [1:\zeta:1]), \\ (p_{d-2} \in \mathbb{P}_F^1, \ [1:\zeta^2:\zeta^2]) \end{array} \right\}.$$

For $S(1,\zeta,\zeta)$, again the uniqueness of p_{d-1} holds when $p_{d-2} \notin \mathfrak{D}$ and we have similar set of solutions in $(E \setminus \mathfrak{D}) \times E$ for $f_{d-2} = g_{d-2} = h_{d-2} = 0$:

$$\left\{ \begin{array}{l} \left(p_{d-2} \in \mathbb{P}_{D}^{1}, [1:\zeta:\zeta] \right), \\ \left(p_{d-2} \in \mathbb{P}_{E}^{1}, [1:1:\zeta^{2}] \right), \\ \left(p_{d-2} \in \mathbb{P}_{F}^{1}, [1:\zeta^{2}:1] \right) \end{array} \right\}.$$

For Z_0 , we study the algebra S(1,0,0). Now if $p_i \in E = E_{1,0,0}$ satisfies $f_{d-2} = g_{d-2} = h_{d-2} = 0$, then we get the system of equations:

(3.9)
$$y_{d-1}z_{d-2} = z_{d-1}x_{d-2} = x_{d-1}y_{d-2} = 0$$
$$x_{d-2}y_{d-2}z_{d-2} = x_{d-1}y_{d-1}z_{d-1} = 0$$

If $p_{d-2} = [0: y_{d-2}: z_{d-2}] \notin \mathfrak{D}$, then both y_{d-2} and $z_{d-2} \neq 0$. Hence $x_{d-1} = y_{d-1} = 0$ and $p_{d-1} = [0:0:1] \in Z_0$. On the other hand if $p_{d-2} = [1: y_{d-2}: z_{d-2}]$, then $z_{d-1} = 0$, i.e. $p_{d-1} = [x_{d-1}: y_{d-1}:0]$. Now since $p_{d-2} \notin \mathfrak{D}$, we have that either y_{d-2} or $z_{d-2} \neq 0$. In the first case, if $y_{d-2} \neq 0$, then $x_{d-1} = 0$ and $p_{d-1} = [0:1:0] \in Z_0$. Otherwise $z_{d-2} \neq 0$ implies that $y_{d-1} = 0$ so $p_{d-1} = [1:0:0] \in Z_0$.

Thus we have verified the lemma.

The next result explores the case when $p_{d-2} \in Z_i$ for i = 1, 2, 3.

Lemma III.22. Let $p = (p_0, \ldots, p_{d-2}) \in V_{d-1}$ with $p_{d-2} \in Z_i$ for i = 1, 2, 3. Then for any $[y_{d-1} : z_{d-1}] \in \mathbb{P}^1$ there exists a function θ of two variables so that

$$p_{d-1} = \left[\theta(y_{d-1}, z_{d-1}) : y_{d-1} : z_{d-1}\right] \notin Z_i$$

which satisfies $(p_0, \ldots, p_{d-2}, p_{d-1}) \in V_d$.

Proof. The point $p' = (p, p_{d-1}) \in V_d$ needs to satisfy $f_i = g_i = h_i = 0$ for $0 \le i \le d-2$ (Definition III.14). Since $p \in V_{d-1}$, we need only to consider the equations $f_{d-2} = g_{d-2} = h_{d-2} = 0$ with $p_{d-2} \in Z_i$.

We study S(1,1,1) for Z_1 so the relevant system of equations is

$$f_{d-2}: y_{d-1}z_{d-2} + z_{d-1}y_{d-2} + x_{d-1}x_{d-2} = 0$$
$$g_{d-2}: z_{d-1}x_{d-2} + x_{d-1}z_{d-2} + y_{d-1}y_{d-2} = 0$$
$$h_{d-2}: x_{d-1}y_{d-2} + y_{d-1}x_{d-2} + z_{d-1}z_{d-2} = 0$$

If $p_{d-2} = [1:1:1] \in Z_1$, then $x_{d-1} = -(y_{d-1} + z_{d-1})$ is required. On the other hand, if $p_{d-2} = [1:\zeta:\zeta^2]$ or $[1:\zeta^2:\zeta]$, we require $x_{d-1} = -\zeta(y_{d-1} + \zeta z_{d-1})$ or $x_{d-1} = -\zeta(\zeta y_{d-1} + z_{d-1})$ respectively. Thus our function θ is defined as

$$\theta(y_{d-1}, z_{d-1}) = \begin{cases} -(y_{d-1} + z_{d-1}), & \text{if } p_{d-2} = [1:1:1] \\ -(\zeta^2 y_{d-1} + \zeta z_{d-1}), & \text{if } p_{d-2} = [1:\zeta:\zeta^2] \\ -(\zeta y_{d-1} + \zeta^2 z_{d-1}), & \text{if } p_{d-2} = [1:\zeta^2:\zeta]. \end{cases}$$

The arguments for $S(1,1,\zeta)$, $S(1,\zeta,\zeta)$, and S(1,0,0) yield similar results as we now describe.

For Z_2 , consider the representative algebra $S(1, 1, \zeta)$ with corresponding equations $f_{d-2} = g_{d-2} = h_{d-2} = 0$. In a similar fashion, we have that:

$$\theta(y_{d-1}, z_{d-1}) = \begin{cases} -(y_{d-1} + \zeta^2 z_{d-1}), & \text{if } p_{d-2} = [1:1:\zeta] \\ -(\zeta^2 y_{d-1} + z_{d-1}), & \text{if } p_{d-2} = [1:\zeta:1] \\ -\zeta(y_{d-1} + z_{d-1}), & \text{if } p_{d-2} = [1:\zeta^2:\zeta^2]. \end{cases}$$

For Z_3 , provided $S(1,\zeta,\zeta)$ as our representative algebra, we have that:

$$\theta(y_{d-1}, z_{d-1}) = \begin{cases} -(y_{d-1} + \zeta z_{d-1}), & \text{if } p_{d-2} = [1:\zeta:\zeta] \\ -(\zeta y_{d-1} + z_{d-1}), & \text{if } p_{d-2} = [1:1:\zeta^2] \\ -\zeta^2(y_{d-1} + z_{d-1}), & \text{if } p_{d-2} = [1:\zeta^2:1]. \end{cases}$$

Lemma III.23. Consider the algebra S(1,0,0). We have that solutions (p_{d-2}, p_{d-1}) to (3.9), with $p_{d-2} \in Z_0$, take the following form:

$$p_{d-1} = \begin{cases} [x_{d-1}:y_{d-1}:0], & \text{if } p_{d-2} = [1:0:0] \\ [0:y_{d-1}:z_{d-1}], & \text{if } p_{d-2} = [0:1:0] \\ [x_{d-1}:0:z_{d-1}], & \text{if } p_{d-2} = [0:0:1] \end{cases} \square$$

Fix a pair $(S_{deg}, Z_i(S_{deg}))$. We now know if $p_{d-2} \notin \mathfrak{D}$, then from every truncated point module of length d over S_{deg} we can produce a unique truncated point module of length d+1. Otherwise if $p_{d-2} \in Z_i$, we get a \mathbb{P}^1 worth of length d+1 modules. We summarize this in the following statement.

Proposition III.24. The parameter space of Γ_d over S_{deg} is isomorphic to the singular and nondisjoint union of

three copies of
$$(\mathbb{P}^1)^{\times \frac{d-1}{2}}$$
 and three copies of $(\mathbb{P}^1)^{\times \frac{d+1}{2}}$, for d odd;
six copies of $(\mathbb{P}^1)^{\times \frac{d}{2}}$, for d even.

The proof of this result will be given in Proposition III.26 below, which will also yield a more detailed statement. We restrict our attention to S(1,1,1) for reasoning mentioned in the proofs of Lemmas III.21 and III.22.

3.2.2.1. Parameterization of Γ_2

Recall that length 3 truncated point modules of Γ_2 are in bijective correspondence to points on $V_2 \subset \mathbb{P}^2 \times \mathbb{P}^2$ (Lemma III.15) and it is our goal to depict this truncated point scheme. By Lemma III.16, we know that $V_2 \subseteq E \times E$.

Now to calculate V_2 , recall that Γ_2 consists of length 3 truncated point modules $M_{(3)} := M_0 \oplus M_1 \oplus M_2$ where M_i is a 1-dimensional k-vector space say with basis m_i . The module $M_{(3)}$ has action determined by $(p_0, p_1) \in V_2$ (Equation (3.6)). Moreover Lemmas III.21 and III.22 provide the precise conditions for (p_0, p_1) to lie in $E \times E$. Namely,

Lemma III.25. Refer to (3.8) for notation. The set of length 3 truncated point modules Γ_2 is parametrized by the scheme $V_2 = \mathbb{V}(f_0, g_0, h_0)$ which is the union of the six subsets:

$$\begin{split} \mathbb{P}^1_A \times [1:1:1]; & [1:1:1] \times \mathbb{P}^1_A; \\ \mathbb{P}^1_B \times [1:\zeta:\zeta^2]; & [1:\zeta:\zeta^2] \times \mathbb{P}^1_B; \\ \mathbb{P}^1_C \times [1:\zeta^2:\zeta]; & [1:\zeta^2:\zeta] \times \mathbb{P}^1_C. \end{split}$$

of $E \times E$. Thus Γ_2 is isomorphic to 6 copies of \mathbb{P}^1 .

3.2.2.2. Parameterization of Γ_d for general d

To illustrate the parametrization of Γ_d , we begin with a truncated point module $M_{(d+1)}$ of length d+1 corresponding to $(p_0, p_1, \ldots, p_{d-1}) \in V_d \subseteq (\mathbb{P}^2)^{\times d}$. Due to Lemmas III.16, III.21, and III.22, we know that $(p_0, p_1, \ldots, p_{d-1})$ belongs to either

$$\underbrace{(E \smallsetminus \mathfrak{D}) \times Z_1 \times (E \smallsetminus \mathfrak{D}) \times Z_1 \times \dots}_{d} \text{ or } \underbrace{Z_1 \times (E \smallsetminus \mathfrak{D}) \times Z_1 \times (E \smallsetminus \mathfrak{D}) \times \dots}_{d}$$

where Z_1 is defined in (3.5).

By adapting the notation of Lemma III.22, we get in the first case that the point

 $(p_0, p_1, \ldots, p_{d-1})$ is of the form

$$([\theta(y_0, z_0) : y_0 : z_0], [1 : \omega : \omega^2], [\theta(y_2, z_2) : y_2 : z_2], [1 : \omega : \omega^2], \dots) \in (\mathbb{P}^2)^{\times d}$$

where $\omega^3 = 1$ and $\theta(y, z) = -(\omega y + \omega^2 z)$. Thus in this case, the set of length d truncated point modules is parameterized by three copies of $(\mathbb{P}^1)^{\times \lceil d/2 \rceil}$ with coordinates

 $([y_0:z_0], [y_2:z_2], \dots, [y_{2\lceil d/2\rceil-1}:z_{2\lceil d/2\rceil-1}]).$

In the second case $(p_0, p_1, \ldots, p_{d-1})$ takes the form

$$([1:\omega:\omega^2], \ [\theta(y_1, z_1):y_1:z_1], \ [1:\omega:\omega^2], [\theta(y_3, z_3):y_3:z_3], \dots) \in (\mathbb{P}^2)^{\times d}$$

and the set of truncated point modules is parameterized with three copies of $(\mathbb{P}^1)^{\times \lfloor d/2 \rfloor}$ with coordinates $([y_1 : z_1], [y_3 : z_3], \dots, [y_{2\lfloor d/2 \rfloor - 1} : z_{2\lfloor d/2 \rfloor - 1}]).$

In other words, we have now proved the next result.

Proposition III.26. Refer to (3.8) for notation. For $d \ge 2$ the truncated point scheme V_d for S(1,1,1) is equal to the union of the six subsets $\bigcup_{i=1}^6 W_{d,i}$ of $(\mathbb{P}^2)^{\times d}$ where

$$\begin{split} W_{d,1} &= \mathbb{P}_{A}^{1} \times [1:1:1] \times \mathbb{P}_{A}^{1} \times [1:1:1] \times \dots, \\ W_{d,2} &= [1:1:1] \times \mathbb{P}_{A}^{1} \times [1:1:1] \times \mathbb{P}_{A}^{1} \times \dots, \\ W_{d,3} &= \mathbb{P}_{B}^{1} \times [1:\zeta:\zeta^{2}] \times \mathbb{P}_{B}^{1} \times [1:\zeta:\zeta^{2}] \times \dots, \\ W_{d,4} &= [1:\zeta:\zeta^{2}] \times \mathbb{P}_{B}^{1} \times [1:\zeta:\zeta^{2}] \times \mathbb{P}_{C}^{1} \times \dots, \\ W_{d,5} &= \mathbb{P}_{C}^{1} \times [1:\zeta^{2}:\zeta] \times \mathbb{P}_{C}^{1} \times [1:\zeta^{2}:\zeta] \times \dots, \\ W_{d,6} &= [1:\zeta^{2}:\zeta] \times \mathbb{P}_{C}^{1} \times [1:\zeta^{2}:\zeta] \times \mathbb{P}_{C}^{1} \times \dots \end{split}$$

As a consequence, we obtain the proof of Proposition III.24 for S(1,1,1). By the next result, this holds for the remaining degenerate Sklyanin algebras.

Proposition III.27. Consider the following notation.

	$S(1,1,\zeta)$	$S(1,\zeta,\zeta)$	S(1, 0, 0)
\mathbb{P}^1_D	$x = -(y + \zeta^2 z)$	$x = -(y + \zeta z)$	z = 0
\mathbb{P}^1_E	$x = -(\zeta^2 y + z)$	$x = -(\zeta y + z)$	x = 0
\mathbb{P}^1_F	$x = -\zeta(y+z)$	$x = -\zeta^2(y+z)$	y = 0
q_1	$[1:1:\zeta]$	$[1:\zeta:\zeta]$	[1:0:0]
q_2	$[1:\zeta:1]$	$\begin{bmatrix}1:1:\zeta^2\end{bmatrix}$	[0:1:0]
q_3	$\left[1:\zeta^2:\zeta^2\right]$	$\left[1\!:\!\zeta^2\!:\!1\right]$	[0:0:1]

For $d \ge 2$, the d^{th} truncated point scheme V_d of the algebras $S(1, 1, \zeta)$ and of $S(1, \zeta, \zeta)$ is the union of the six subsets $\bigcup_{i=1}^{6} W_{d,i}$ of $(\mathbb{P}^2)^{\times d}$ where

$$W_{d,1} = \mathbb{P}_D^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \{q_1\} \times \dots,$$

$$W_{d,2} = \{q_1\} \times \mathbb{P}_D^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \dots,$$

$$W_{d,3} = \mathbb{P}_E^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \{q_2\} \times \dots,$$

$$W_{d,4} = \{q_2\} \times \mathbb{P}_E^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \dots,$$

$$W_{d,5} = \mathbb{P}_F^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \{q_3\} \times \dots,$$

$$W_{d,6} = \{q_3\} \times \mathbb{P}_F^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \dots.$$

Moreover the V_d for the algebra S(1,0,0) are given as the union of the following subsets of $(\mathbb{P}^2)^{\times d}$:

$$\begin{split} W_{d,1} &= \mathbb{P}_D^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \dots, \\ W_{d,2} &= \{q_2\} \times \mathbb{P}_E^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \{q_2\} \times \dots, \\ W_{d,3} &= \mathbb{P}_E^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \dots, \\ W_{d,4} &= \{q_3\} \times \mathbb{P}_F^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \{q_3\} \times \dots, \\ W_{d,5} &= \mathbb{P}_F^1 \times \{q_1\} \times \mathbb{P}_D^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \dots, \\ W_{d,6} &= \{q_1\} \times \mathbb{P}_D^1 \times \{q_2\} \times \mathbb{P}_E^1 \times \{q_3\} \times \mathbb{P}_F^1 \times \dots. \end{split}$$

Proof. This follows by a similar proof to that of Proposition III.26 and is left to the reader. $\hfill \square$

We thank Karen Smith for suggesting the following elegant way of interpreting the point scheme of S(1,1,1).

Remark III.28. We can provide an alternate geometric description of the point scheme of the Γ of S(1,1,1). Let $G := \mathbb{Z}_3 \rtimes \mathbb{Z}_2 = \langle \zeta, \sigma \rangle$ where $\zeta = e^{2\pi i/3}$ and $\sigma^2 = 1$. We define a *G*-action on $\mathbb{P}^2 \times \mathbb{P}^2$ as follows:

$$\begin{aligned} \zeta([x:y:z], [u:v:w]) &= ([x:\zeta^2 y:\zeta z], [u:\zeta v:\zeta^2 w]) \\ \sigma([x:y:z], [u:v:w]) &= ([u:v:w], [x:y:z]) \end{aligned}$$

Note that G stabilizes $E \times E$ and acts transitively on the $W_{2,i}$. We extend the action of G to $(\mathbb{P}^2 \times \mathbb{P}^2)^{\times \infty}$ diagonally. Now we interpret Γ as

$$\Gamma = \lim_{\longleftarrow} V_d = \lim_{\longleftarrow} V_{2d} = \lim_{\longleftarrow} \bigcup_i W_{2d,i} = G \cdot (\mathbb{P}^1_A \times [1:1:1])^{\times \infty},$$

as sets.

3.3 Point parameter ring of S(1,1,1)

We now construct a graded associative algebra P from truncated point schemes of the degenerate Sklyanin algebra S = S(1, 1, 1). The analogous result for the other degenerate Sklyanin algebras will follow in a similar fashion and we leave the details to the reader. As is true for the Sklyanin algebras themselves, it will be shown that this algebra P is a proper factor of S(1, 1, 1) and its properties closely reflect those of S(1, 1, 1). We will for example show that P is not right noetherian, nor a domain. In other words, we establish Theorem III.4.

The definition of the algebra P initially appears in [ATVdB90, §3]. In particular for P = P(S(a, b, c)), recall that we have projection maps $pr_{1,...,d-1}$ and $pr_{2,...,d}$ from $(\mathbb{P}^2)^{\times d}$ to $(\mathbb{P}^2)^{\times d-1}$. Restrictions of these maps to the truncated point schemes $V_d \subseteq$ $(\mathbb{P}^2)^{\times d}$ (Definition III.14) yield

$$pr_{1,\dots,d-1}(V_d) \subset V_{d-1}$$
 and $pr_{2,\dots,d}(V_d) \subset V_{d-1}$ for all d .

Definition III.29. Given the above data, we expand on the definition from the introduction. The **point parameter ring** P = P(S(a, b, c)) is an associative \mathbb{N} -graded ring defined as follows. First $P_d = H^0(V_d, \mathcal{L}_d)$ where \mathcal{L}_d is the restriction of invertible sheaf

$$pr_1^*\mathcal{O}_{\mathbb{P}^2}(1) \otimes \ldots \otimes pr_d^*\mathcal{O}_{\mathbb{P}^2}(1) \cong \mathcal{O}_{(\mathbb{P}^2)^{\times d}}(1,\ldots,1)$$

to V_d . The multiplication map $\mu_{i,j}: P_i \times P_j \to P_{i+j}$ is then defined by applying H^0 to the isomorphism

$$pr_{1,\ldots,i}^{*}(\mathcal{L}_{i}) \otimes_{\mathcal{O}_{V_{i+i}}} pr_{i+1,\ldots,i+j}^{*}(\mathcal{L}_{j}) \to \mathcal{L}_{i+j}.$$

We declare $P_0 = k$.

From now on, we restrict our attention to P(S(a, b, c)) with $[a : b : c] \in \mathfrak{D}$, or rather for S_{deg} . We will later see in Theorem III.36 that $P(S_{deg})$ is generated in degree one; thus S_{deg} surjects onto P as $(S_{deg})_1 \cong P_1$. To begin the analysis of P for S(1, 1, 1), recall that $V_1 = \mathbb{P}^2$ so

$$P_1 = H^0(V_1, pr_1^*\mathcal{O}_{\mathbb{P}^2}(1)) = kx \oplus ky \oplus kz$$

where [x : y : z] are the coordinates of \mathbb{P}^2 . For $d \ge 2$ we will compute $\dim_k P_d$ and then proceed to the more difficult task of identifying the multiplication maps $\mu_{i,j} : P_i \times P_j \to P_{i+j}$. Before we get to specific calculations for $d \ge 2$, let us recall that the schemes V_d are realized as the union of six subsets $\{W_{d,i}\}_{i=1}^6$ of $(\mathbb{P}^2)^{\times d}$ described in Proposition III.26 and Equation (3.8). These subsets intersect nontrivially so that each V_d for $d \ge 2$ is singular. More precisely, **Remark III.30.** A routine computation shows that the singular subset, $Sing(V_d)$, consists of six points:

$$\begin{array}{lll} v_{d,1} \coloneqq & ([1:1:1], \ [1:\zeta:\zeta^2], \ [1:1:1], \ [1:\zeta:\zeta^2], \dots) & \in W_{d,2} \cap W_{d,3}, \\ v_{d,2} \coloneqq & ([1:1:1], \ [1:\zeta^2:\zeta], \ [1:1:1], \ [1:\zeta^2:\zeta], \dots) & \in W_{d,2} \cap W_{d,5}, \\ v_{d,3} \coloneqq & ([1:\zeta:\zeta^2], \ [1:1:1], \ [1:\zeta:\zeta^2], \ [1:1:1], \dots) & \in W_{d,1} \cap W_{d,4}, \\ v_{d,4} \coloneqq & ([1:\zeta:\zeta^2], \ [1:\zeta:\zeta^2], \ [1:\zeta:\zeta^2], \ [1:\zeta:\zeta^2], \ [1:\zeta:\zeta^2], \dots) & \in W_{d,3} \cap W_{d,4}, \\ v_{d,5} \coloneqq & ([1:\zeta^2:\zeta], \ [1:1:1], \ [1:\zeta^2:\zeta], \ [1:1:1], \dots) & \in W_{d,1} \cap W_{d,6}, \\ v_{d,6} \coloneqq & ([1:\zeta^2:\zeta], \ [1:\zeta^2:\zeta], \ [1:\zeta^2:\zeta], \ [1:\zeta^2:\zeta], \dots) & \in W_{d,5} \cap W_{d,6}. \end{array}$$

where $\zeta = e^{2\pi i/3}$.

3.3.1 Computing the dimension of P_d

Our objective in this section is to prove

Proposition III.31. For $d \ge 1$, $\dim_k P_d = 3\left(2^{\lfloor \frac{d+1}{2} \rfloor} + 2^{\lceil \frac{d-1}{2} \rceil}\right) - 6$.

We first point out a result of Künneth that will be used several times in the rest of the chapter.

Theorem III.32. [BG06, A.10.37] (Künneth's Formula) Let X_1 and X_2 be varieties over k. Let \mathcal{E} be a locally free sheaf on X_1 and let \mathcal{F} be a coherent sheaf on X_2 . Then we have that

$$H^n(X_1 \times X_2, pr_1^* \mathcal{E} \otimes pr_2^* \mathcal{F}) \cong \bigoplus_{p+q=n} H^p(X_1, \mathcal{E}) \otimes H^q(X_2, \mathcal{F}).$$

For the rest of the section, let **1** denote a sequence of 1s of appropriate length. Now consider the normalization morphism $\pi : V'_d \to V_d$ where V'_d is the disjoint union of the six subsets $\{W_{d,i}\}_{i=1}^6$ mentioned in Proposition III.26. This map induces the following short exact sequence of sheaves on V_d :

$$(3.10) 0 \to \mathcal{O}_{V_d}(1) \to (\pi_*\mathcal{O}_{V'_d})(1) \to \mathcal{S}(1) \to 0,$$

where \mathcal{S} is the skyscraper sheaf whose support is $\operatorname{Sing}(V_d)$, that is $\mathcal{S} = \bigoplus_{k=1}^6 \mathcal{O}_{\{v_{d,k}\}}$.

Note that we have

(3.11)
$$H^{0}(V_{d}, (\pi_{*}\mathcal{O}_{V'_{d}})(\mathbf{1})) \cong_{k-\mathrm{v.s.}} H^{0}(V'_{d}, \mathcal{O}_{V'_{d}}(\mathbf{1}))$$

since the normalization morphism is a finite map, which in turn is an affine map [Har77, Exercises II.5.17(b), III.4.1]. To complete the proof of the proposition, we make the following assertion:

Sublemma III.33. $H^1(V_d, \mathcal{O}_{V_d}(1)) = 0.$

Assuming that Sublemma III.33 holds, we get from (3.10) the following long exact sequence of cohomology:

$$0 \to H^0(V_d, \mathcal{O}_{V_d}(\mathbf{1})) \to H^0(V_d, (\pi_*\mathcal{O}_{V'_d})(\mathbf{1}))$$
$$\to H^0(V_d, \mathcal{S}(\mathbf{1})) \to H^1(V_d, \mathcal{O}_{V_d}(\mathbf{1})) = 0$$

Thus, with writing $h^0(X, \mathcal{L}) = \dim_k H^0(X, \mathcal{L})$, (3.11) implies that

$$\dim_k P_d = h^0(\mathcal{O}_{V_d}(\mathbf{1})) = h^0((\pi_*\mathcal{O}_{V'_d})(\mathbf{1})) - h^0(\mathcal{S}(\mathbf{1}))$$
$$= h^0(\mathcal{O}_{V'_d}(\mathbf{1})) - h^0(\mathcal{S}(\mathbf{1}))$$
$$= \sum_{i=1}^6 h^0(\mathcal{O}_{W_{d,i}}(\mathbf{1})) - 6.$$

Therefore applying Proposition III.24 and Theorem III.32 completes the proof of Proposition III.31. It now remains to verify Sublemma III.33.

Proof of Sublemma III.33: By the discussion above, it suffices to show that

$$\delta_d: H^0(V'_d, \mathcal{O}_{V'_d}(\mathbf{1})) \to H^0\left(\bigcup_{k=1}^6 \{v_{d,k}\}, \ \mathcal{S}(\mathbf{1})\right)$$

is surjective. Referring to the notation of Proposition III.26 and Remark III.30, we choose $v_{d,i} \in \text{Supp}(\mathcal{S}(1))$ and W_{d,k_i} containing $v_{d,i}$. This W_{d,k_i} contains precisely two points of $\text{Supp}(\mathcal{S}(1))$ and say the other is $v_{d,j}$ for $j \neq i$. After choosing a basis $\{t_i\}_{i=1}^6$

for the six-dimensional vector space $H^0(\mathcal{S}(\mathbf{1}))$ where $t_i(v_{d,j}) = \delta_{ij}$, we construct a preimage of each t_i . Since $\mathcal{O}_{W_{d,k_i}}(\mathbf{1})$ is a very ample sheaf, it separates points. In other words there exists $\tilde{s}_i \in H^0(\mathcal{O}_{W_{d,k_i}}(\mathbf{1}))$ such that $\tilde{s}_i(v_{d,j}) = \delta_{ij}$. Extend this section \tilde{s}_i to $s_i \in H^0(\mathcal{O}_{V'_d}(\mathbf{1}))$ by declaring $s_i = \tilde{s}_i$ on W_{d,k_i} and $s_i = 0$ elsewhere. Thus $\delta_d(s_i) = t_i$ for all i and the map δ_d is surjective as desired. \Box

This concludes the proof of Proposition III.31.

Corollary III.34. We have that P has exponential growth hence infinite GK dimension. We also get that P is not left or right noetherian.

Proof. Recall from Proposition III.31 that $\dim_k P_d = 3\left(2^{\lceil \frac{d-1}{2}\rceil} + 2^{\lfloor \frac{d+1}{2}\rfloor} - 2\right)$, which is greater than $3 \cdot 2^{d/2}$ for d >> 0. Moreover

$$\sum_{i \le d} 3 \cdot 2^{i/2} \ge 3 \left(2^{\lceil d/2 \rceil} - 1 \right) =: s_d.$$

Since $\overline{\lim}_d s_d^{1/d} = \sqrt{2}$, we have that *P* has exponential growth. By Theorem II.30, the ring *P* is not left or right noetherian.

We can also determine the Hilbert series of P.

Proposition III.35. $H_P(t) = \frac{(1+t^2)(1+2t)}{(1-2t^2)(1-t)}.$

Proof. Recall from Proposition III.31 that $\dim_k P_d = 3\left(2^{\lceil \frac{d-1}{2}\rceil} + 2^{\lfloor \frac{d+1}{2}\rfloor}\right) - 6$ for $d \ge 1$ and that $\dim_k P_0 = 1$. Thus

$$H_P(t) = 1 + 3\left(\sum_{d\geq 1} 2^{\lceil \frac{d-1}{2} \rceil} t^d + \sum_{d\geq 1} 2^{\lfloor \frac{d+1}{2} \rfloor} t^d - 2\sum_{d\geq 1} t^d\right)$$

= 1 + 3\left(t \sum_{d\geq 0} 2^{\lceil \frac{d}{2} \rceil} t^d + 2t \sum_{d\geq 0} 2^{\lfloor \frac{d}{2} \rfloor} t^d - 2t \sum_{d\geq 0} t^d\right)

Consider generating functions $a(t) = \sum_{d \ge 0} a_d t^d$ and $b(t) = \sum_{d \ge 0} b_d t^d$ for the respective sequences $a_d = 2^{\lceil d/2 \rceil}$ and $b_d = 2^{\lfloor d/2 \rfloor}$. Elementary operations result in $a(t) = \frac{1+2t}{1-2t^2}$ and $b(t) = \frac{1+t}{1-2t^2}$. Hence

$$H_P(t) = 1 + 3\left[t\left(\frac{1+2t}{1-2t^2}\right) + 2t\left(\frac{1+t}{1-2t^2}\right) - 2t\left(\frac{1}{1-t}\right)\right] = \frac{(1+t^2)(1+2t)}{(1-2t^2)(1-t)}.$$

3.3.2 The multiplication maps $\mu_{ij}: P_i \times P_j \rightarrow P_{i+j}$

In this section we examine the multiplication of the point parameter ring P of S(1,1,1). In particular, we show that the multiplication maps are surjective which results in the following theorem.

Theorem III.36. The point parameter ring P of S(1,1,1) is generated in degree one.

With similar reasoning, the ring $P = P(S_{deg})$ is generated in degree one for all S_{deg} . We leave the details to the reader.

Proof. It suffices to prove that the multiplication maps $\mu_{d,1} : P_d \times P_1 \to P_{d+1}$ are surjective for $d \ge 1$. Recall from Definition III.29 that $\mu_{d,1} = H^0(m_d)$ where m_d is the isomorphism

$$m_d: \mathcal{O}_{V_d \times \mathbb{P}^2}(1, \dots, 1, 0) \otimes_{\mathcal{O}_{V_{d+1}}} \mathcal{O}_{(\mathbb{P}^2)^{\times d}}(0, \dots, 0, 1) \to \mathcal{O}_{V_{d+1}}(1, \dots, 1).$$

To use the isomorphism m_d , we employ the following commutative diagram:

The source of t_d is isomorphic to $\mathcal{O}_{V_d \times \mathbb{P}^2}(1, \ldots, 1)$ and the map t_d is given by restriction to V_{d+1} . Hence we have the short exact sequence

(3.13)
$$0 \longrightarrow \mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(1) \longrightarrow \mathcal{O}_{V_d \times \mathbb{P}^2}(1) \xrightarrow{t_d} \mathcal{O}_{V_{d+1}}(1) \longrightarrow 0,$$

where $\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}$ is the ideal sheaf of V_{d+1} defined in $V_d \times \mathbb{P}^2$. Since Theorem III.32 and Sublemma III.33 implies that $H^1(\mathcal{O}_{V_d \times \mathbb{P}^2}(\mathbf{1})) = 0$, the cokernel of $H^0(t_d)$ is $H^1(\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(\mathbf{1}))$. Now we assert:

Proposition III.37. $H^1\left(\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(1)\right) = 0 \text{ for } d \ge 1.$

By assuming that Proposition III.37 holds, we get the surjectivity of $H^0(t_d)$ for $d \ge 1$. Now by applying the global section functor to Diagram (3.12), we have that $H^0(m_d) = \mu_{d,1}$ is surjective for $d \ge 1$. This concludes the proof of Theorem III.36. \Box

Proof of III.37. Consider the case d = 1. We study the ideal sheaf $\mathcal{I}_{\frac{V_2}{\mathbb{P}^2 \times \mathbb{P}^2}} \coloneqq \mathcal{I}_{V_2}$ by using the resolution of the ideal of defining relations (f_0, g_0, h_0) for V_2 (Equations (3.3)) in the \mathbb{N}^2 -graded ring $R = k[x_0, y_0, z_0, x_1, y_1, z_1]$. Note that each of the defining equations have bidegree (1,1) in R and we get the following resolution:

$$0 \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-3, -3) \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-2, -2)^{\oplus 3} \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-1, -1)^{\oplus 3} \to \mathcal{I}_{V_2} \to 0.$$

Twisting the above sequence with $\mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(1,1)$ we get

$$0 \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-2, -2) \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-1, -1)^{\oplus 3} \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}^{\oplus 3} \xrightarrow{f} \mathcal{I}_{V_2}(1, 1) \to 0.$$

Let $\mathcal{K} = \ker(f)$. Then $h^0(\mathcal{I}_{V_2}(1,1)) = 3 - h^0(\mathcal{K}) + h^1(\mathcal{K})$. On the other hand, $H^1(\mathcal{O}_{\mathbb{P}^2}(j)) = H^2(\mathcal{O}_{\mathbb{P}^2}(j)) = 0$ for j = -1, -2. Thus Theorem III.32 applied the cohomology of the short exact sequence

$$0 \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-2, -2) \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(-1, -1)^{\oplus 3} \to \mathcal{K} \to 0$$

results in $h^0(\mathcal{K}) = h^1(\mathcal{K}) = 0$. Hence $h^0(\mathcal{I}_{V_2}(1,1)) = 3$.

Now using the long exact sequence of cohomology arising from the short exact sequence

$$0 \to \mathcal{I}_{V_2}(1,1) \to \mathcal{O}_{\mathbb{P}^2 \times \mathbb{P}^2}(1,1) \to \mathcal{O}_{V_2}(1,1) \to 0,$$

and the facts:

$$h^{0}(\mathcal{I}_{V_{2}}(1,1)) = 3, \qquad h^{0}(\mathcal{O}_{\mathbb{P}^{2} \times \mathbb{P}^{2}}(1,1)) = 9$$
$$h^{0}(\mathcal{O}_{V_{2}}(1,1)) = \dim_{k} P_{2} = 6, \qquad h^{1}(\mathcal{O}_{\mathbb{P}^{2} \times \mathbb{P}^{2}}(1,1)) = 0$$

we conclude that $H^1(\mathcal{I}_{V_2}(1,1)) = 0$.

For $d \ge 2$ we will construct a commutative diagram to assist with the study of the cohomology of the ideal sheaf $\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(\mathbf{1})$. Recall from (3.10) that we have the following normalization sequence for V_d :

$$0 \longrightarrow \mathcal{O}_{V_d} \longrightarrow \bigoplus_{i=1}^6 \mathcal{O}_{W_{d,i}} \longrightarrow \bigoplus_{k=1}^6 \mathcal{O}_{\{v_{d,k}\}} \longrightarrow 0.$$
 (†_d)

Consider the sequence

$$pr_{1,\dots,d}^{*}\left(\left(\dagger_{d}\right)\otimes\mathcal{O}_{(\mathbb{P}^{2})^{\times d}}(1)\right)\otimes_{\mathcal{O}_{(\mathbb{P}^{2})^{\times d+1}}}pr_{d+1}^{*}\mathcal{O}_{\mathbb{P}^{2}}(1)$$

and its induced sequence of restrictions to V_{d+1} , namely

$$(3.14) \qquad 0 \to \mathcal{O}_{V_d \times \mathbb{P}^2}(\mathbf{1})\big|_{V_{d+1}} \to \bigoplus_{i=1}^6 \mathcal{O}_{W_{d,i} \times \mathbb{P}^2}(\mathbf{1})\big|_{V_{d+1}} \to \bigoplus_{k=1}^6 \mathcal{O}_{\{v_{d,k}\} \times \mathbb{P}^2}(\mathbf{1})\big|_{V_{d+1}} \to 0.$$

Now $V_{d+1} \subseteq V_d \times \mathbb{P}^2$ and $(W_{d,i} \times \mathbb{P}^2) \cap V_{d+1} = W_{d+1,i}$ due to Proposition III.26 and Remark III.30. We also have that $(\{v_{d,k}\} \times \mathbb{P}^2) \cap V_{d+1} = \{v_{d+1,k}\}$ for all i,k. Therefore the sequence (3.14) is equal to $(\dagger_{d+1}) \otimes \mathcal{O}_{(\mathbb{P}^2)^{\times d+1}}(1)$. In other words, we are given the commutative diagram:

Diagram 1: Understanding $\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(1, \dots, 1)$

where the vertical maps are given by restriction to V_{d+1} . Observe that the kernels of the vertical maps (from left to right) are respectively $\mathcal{I}_{\frac{V_{d+1}}{V_d \times \mathbb{P}^2}}(\mathbf{1}), \bigoplus_i \mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(\mathbf{1})$, and $\bigoplus_k \mathcal{I}_{\frac{\{v_{d+1,k}\}}{\{v_{d,k}\} \times \mathbb{P}^2}}(\mathbf{1})$, and the cokernels are all 0.

By the Sublemma III.33 and Theorem III.32, we have that

$$H^1(\mathcal{O}_{V_d \times \mathbb{P}^2}(1)) = H^1(\mathcal{O}_{V_{d+1}}(1)) = 0.$$

Hence the application of the global section functor to Diagram 1 yields Diagram 2 below.



Diagram 2: Induced Cohomology from Diagram 1

Now by the Snake Lemma, we get the following sequence:

$$\dots \longrightarrow \bigoplus_{i=1}^{6} H^{0}\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^{2}}}(1)\right) \xrightarrow{\psi} \bigoplus_{k=1}^{6} H^{0}\left(\mathcal{I}_{\frac{\{v_{d+1,k}\}}{\{v_{d,k}\} \times \mathbb{P}^{2}}}(1)\right)$$
$$\longrightarrow H^{1}\left(\mathcal{I}_{\frac{V_{d+1}}{V_{d} \times \mathbb{P}^{2}}}(1)\right) \longrightarrow \bigoplus_{i=1}^{6} H^{1}\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^{2}}}(1)\right) \to \dots$$

In Lemma III.38, we will show that $\bigoplus_{i} H^1\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(\mathbf{1})\right) = 0$ for $d \ge 2$. Furthermore the surjectivity of the map ψ will follow from Lemma III.39. This will complete the proof of Proposition III.37.

Lemma III.38.
$$\bigoplus_{i=1}^{6} H^1\left(W_{d,i} \times \mathbb{P}^2, \ \mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(1)\right) = 0 \text{ for } d \ge 2.$$

Proof. We consider the different parities of d and i separately. For d even and i odd,

$$\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i}\times\mathbb{P}^2}} \cong \mathcal{O}_{W_{d,i}\times\mathbb{P}^2}(0,\ldots,0,-1)$$

because $W_{d+1,i}$ is defined in $W_{d,i} \times \mathbb{P}^2$ by one equation of degree $(0, \ldots, 0, 1)$ (Proposition III.26). Twisting by $\mathcal{O}_{(\mathbb{P}^2)^{\times d+1}}(1, \ldots, 1)$ results in

(3.15)
$$H^{1}\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i}\times\mathbb{P}^{2}}}(1,\ldots,1)\right) \cong H^{1}\left(\mathcal{O}_{W_{d,i}\times\mathbb{P}^{2}}(1,\ldots,1,0)\right).$$

Since $W_{d,i}$ is the product of \mathbb{P}^1 and points lying in \mathbb{P}^2 and $H^1(\mathcal{O}_{\mathbb{P}^1}(1)) = H^1(\mathcal{O}_{\{pt\}}(1)) = H^1(\mathcal{O}_{\mathbb{P}^2}) = 0$, Theorem III.32 implies that the right hand side of (3.15) is equal to zero.

Consider the case of d and i even. As $pr_{1,\ldots,d}(W_{d+1,i}) = W_{d,i}$ and $pr_{d+1}(W_{d+1,i}) = [1 : \omega : \omega^2]$ for $\omega = \omega_{d,i}$ a third of unity, we have that $W_{d+1,i}$ is defined in $W_{d,i} \times \mathbb{P}^2$ by two equations of degree $(0,\ldots,0,1)$. The defining equations (in variables x, y, z) of $[1 : \omega : \omega^2]$ form a k[x, y, z]-regular sequence and so we have

the Koszul resolution of $\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}} \otimes \mathcal{O}_{(\mathbb{P}^2)^{\times d+1}}(1,\ldots,1)$:

$$(3.16) \qquad 0 \to \mathcal{O}_{W_{d,i} \times \mathbb{P}^2}(1, \dots, 1, -1) \to \mathcal{O}_{W_{d,i} \times \mathbb{P}^2}(1, \dots, 1, 0)^{\oplus 2} \\ \to \mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(1, \dots, 1) \to 0.$$

Now apply the global section functor to sequence (3.16) and note that

$$H^{j}(\mathcal{O}_{W_{d,i}}(1,\ldots,1)) = H^{j}(\mathcal{O}_{\mathbb{P}^{2}}) = H^{j}(\mathcal{O}_{\mathbb{P}^{2}}(-1)) = 0 \text{ for } j = 1,2.$$

Hence Theorem III.32 yields

$$H^1(\mathcal{O}_{W_{d,i}\times\mathbb{P}^2}(1,\ldots,1,0))^{\oplus 2} = H^2(\mathcal{O}_{W_{d,i}\times\mathbb{P}^2}(1,\ldots,1,-1)) = 0.$$

Therefore $H^1\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i}\times\mathbb{P}^2}}(\mathbf{1})\right) = 0$ for d and i even.

We conclude that for d even, we know $\bigoplus_{i=1}^{6} H^1\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(\mathbf{1})\right) = 0$. For d odd, the same conclusion is drawn by swapping the arguments for the i even and i odd subcases. \Box

Lemma III.39. The map ψ is surjective for $d \ge 2$.

Proof. Refer to the notation from Diagram 2. To show ψ is onto, here is our plan of attack.

- 1. Choose a basis of $\bigoplus_{k} H^{0}\left(\mathcal{I}_{\frac{\{v_{d+1,k}\}}{\{v_{d,k}\}\times\mathbb{P}^{2}}}(\mathbf{1})\right)$ so that each basis element t lies in $H^{0}\left(\mathcal{I}_{\frac{\{v_{d+1,k_{0}}\}}{\{v_{d,k_{0}}\}\times\mathbb{P}^{2}}}(\mathbf{1})\right)$ for some $k = k_{0}$. For such a basis element t, identify its image under λ in $\bigoplus_{k} H^{0}\left(\mathcal{O}_{\{v_{d,k}\}\times\mathbb{P}^{2}}(\mathbf{1})\right)$.
- 2. Construct for $\lambda(t)$ a suitable preimage $s \in \nu^{-1}(\lambda(t))$.
- 3. Prove $s \in \ker(\beta)$.

As a consequence, s lies in $\bigoplus_{i} H^0\left(\mathcal{I}_{\frac{W_{d+1,i}}{W_{d,i} \times \mathbb{P}^2}}(\mathbf{1})\right)$ and serves as a preimage to t under ψ . In other words, ψ is surjective. To begin, fix such a basis element t and integer k_0 .

Step 1: Observe that $pr_{1,\dots,d}(\{v_{d+1,k_0}\}) = \{v_{d,k_0}\}$ and $pr_{d+1}(\{v_{d+1,k_0}\}) = [1:\omega:\omega^2]$ for some ω , a third root of unity (Remark III.30). Thus our basis element $t \in \bigoplus_k H^0(\mathcal{I}_{\frac{\{v_{d+1,k}\}}{\{v_{d,k}\}\times\mathbb{P}^2}}(1))$ is of the form

(3.17)
$$t = a(\omega x_d - y_d) + b(\omega^2 x_d - z_d)$$

for some $a, b \in k$, with $\{\omega x_d - y_d, \omega^2 x_d - z_d\}$ defining $[1 : \omega : \omega^2]$ in the $(d+1)^{st}$ copy of \mathbb{P}^2 . Note that λ is the inclusion map so we may refer to $\lambda(t)$ as t. This concludes Step 1.

Step 2: Next we construct a suitable preimage $s \in \nu^{-1}(\lambda(t))$. Referring to Remark III.30, let us observe that for all k, there is an unique even integer $:= i''_k$ and unique odd integer $:= i'_k$ so that $v_{d,k} \in W_{d,i''_k} \cap W_{d,i'_k}$ for all k = 1, ..., 6. For instance with $k_0 = 1$, we consider the membership $v_{d,1} \in W_{d,2} \cap W_{d,3}$; hence $i''_1 = 2$ and $i'_1 = 3$.

As a consequence, $\lambda(t)$ has preimages under ν in

$$H^0\left(W_{d,i_{k_0}''}\times\mathbb{P}^2,\mathcal{O}_{W_{d,i_{k_0}''}\times\mathbb{P}^2}(\mathbf{1})\right)\oplus\ H^0\left(W_{d,i_{k_0}'}\times\mathbb{P}^2,\mathcal{O}_{W_{d,i_{k_0}'}\times\mathbb{P}^2}(\mathbf{1})\right).$$

For d even (respectively odd) we write $i_{k_0} := i''_{k_0}$ (respectively $i_{k_0} := i'_{k_0}$). Therefore we intend to construct $s \in \nu^{-1}(t)$ belonging to $H^0(\mathcal{O}_{W_{d,i_{k_0}} \times \mathbb{P}^2}(\mathbf{1}))$. However this $W_{d,i_{k_0}}$ will also contain another point $v_{d,j}$ for some $j \neq k_0$. Let us define the global section $\tilde{s} \in H^0(\mathcal{O}_{W_{d,i_{k_0}} \times \mathbb{P}^2}(\mathbf{1}))$ as follows. Since $\mathcal{O}_{W_{d,i_{k_0}}}(\mathbf{1})$ is a very ample sheaf, we have a global section \tilde{s}_{k_0} separating the points v_{d,k_0} and $v_{d,j}$; say $\tilde{s}_{k_0}(v_{d,k}) = \delta_{k_0,k}$. We then use (3.17) to define \tilde{s} by

$$\tilde{s} = \tilde{s}_{k_0} \cdot [a(\omega x_d - y_d) + b(\omega^2 x_d - z_d)].$$

where $[1:\omega:\omega^2] = pr_{d+1}(\{v_{d+1,k_0}\})$. We now extend this section \tilde{s} to

$$s \in \bigoplus_{i=1}^{6} H^{0}\left(\mathcal{O}_{W_{d,i} \times \mathbb{P}^{2}}(\mathbf{1})\right) \cong \left(\bigoplus_{i=1}^{6} H^{0}\left(\mathcal{O}_{W_{d,i}}(\mathbf{1})\right)\right) \otimes H^{0}(\mathcal{O}_{\mathbb{P}^{2}}(\mathbf{1})).$$

This is achieved by setting $s = \tilde{s}$ on $W_{d,i_{k_0}} \times \mathbb{P}^2$ and 0 elsewhere. To check that $\nu(s) = t$, note

(3.18)
$$s = \bigoplus_{i=1}^{6} s_i \text{ where } s_i \in H^0\left(\mathcal{O}_{W_{d,i} \times \mathbb{P}^2}(\mathbf{1})\right), \ s_i = \begin{cases} \tilde{s}, & i = i_{k_0}, \\ 0, & i \neq i_{k_0}; \end{cases}$$

Therefore by the construction of \tilde{s} , we have $\nu(\tilde{s}) = t|_{\{v_{d,k_0}\}\times\mathbb{P}^2}$. Hence we have built our desired preimage $s \in \nu^{-1}(t)$ and this concludes Step 2.

Step 3: Recall the structure of s from (3.18). By definition of β , we have that $\beta(s) = \beta\left(\bigoplus_{i=1}^{6} s_i\right)$ is equal to $\bigoplus_{i=1}^{6} \left(s_i\Big|_{W_{d+1,i}}\right)$.

For $i \neq i_{k_0}$, we clearly get that $s_i \Big|_{W_{d+1,i}} = 0$. On the other hand, the key point of our construction is that $W_{d+1,i_{k_0}} = W_{d,i_{k_0}} \times [1 : \epsilon : \epsilon^2]$ for some $\epsilon^3 = 1$ as i_{k_0} is chosen to be even (respectively odd) when d is even (respectively odd) (Proposition III.26). Moreover $v_{d+1,k_0} \in W_{d+1,i_{k_0}}$ and

$$pr_{d+1}(W_{d+1,i_{k_0}}) = pr_{d+1}(\{v_{d+1,k_0}\}) = [1:\omega:\omega^2]$$

where ω is defined by Step 1 and Remark III.30. Thus $\epsilon = \omega$. Now we have

$$s_{i_{k_0}}\Big|_{W_{d+1,i_{k_0}}} = \tilde{s}_{k_0} \cdot \left[a(\omega x_d - y_d) + b(\omega^2 x_d - z_d)\right]\Big|_{[1:\omega:\omega^2]} = 0.$$

Therefore $s_i|_{W_{d+1,i}} = 0$ for all i = 1, ..., 6. Hence $\beta(s) = 0$.

Hence Steps 1-3 are complete which concludes the proof of Lemma III.39. $\hfill \Box$

Consequently, we have verified Proposition III.37.

One of the main results why twisted homogeneous coordinate rings are so useful for studying Sklyanin algebras is that the former are factors of their corresponding Sklyanin algebra (by some homogeneous element; refer to Theorem II.16). The following corollaries to Theorem III.36 illustrate an analogous result for S_{deg} . **Corollary III.40.** Let P be the point parameter ring of a degenerate Sklyanin algebra S_{deg} . Then $P \cong S_{deg}/K$ for some ideal K of S_{deg} that has six generators of degree 4 and possibly higher degree generators.

Proof. Since $(S_{deg})_1 \cong P_1$, we have by Theorem III.36 that S_{deg} surjects onto P say with kernel K. By Lemma III.8 we have that $\dim_k S_4 = 57$, yet we know $\dim_k P_4 =$ 63 by Proposition III.31. Hence $\dim_k K_4 = 6$. The same results also imply that $\dim_k S_d = \dim_k P_d$ for $d \leq 3$.

Corollary III.41. The ring $P = P(S_{deg})$ is neither a domain or Koszul.

Proof. By Corollary III.9, there exist linear nonzero elements $u, v \in S$ with uv = 0. The image of u and v are nonzero, hence P is not a domain due to Corollary III.40. Since P has degree 4 relations, it does not possess the Koszul property.

3.4 Open problems pertaining to point parameter rings

This section is dedicated to furthering the study of noncommutative coordinate rings in noncommutative projective algebraic geometry. In particular, we outline open problems pertaining to the point parameter rings introduced in Definition I.9 (or in Definition III.29 for three-dimensional Sklyanin algebras). This is with a view towards a theory that encompasses all coordinate rings arising from the geometric data of noncommutative connected graded algebras.

The problems discussed in the section are prompted by the observation that one can produce point scheme data (Definition-Lemma II.6) and subsequently a point parameter ring for *any* connected graded ring. Recall that point parameter rings are thought of as generalizations of twisted homogeneous coordinate rings. Furthermore the catalyst for the first task is the relationship between three-dimensional Sklyanin algebras, $Skly_3$, and twisted homogeneous coordinate rings. Namely Theorem II.16 states that the point scheme data of a generic $Skly_3$ stabilizes to a smooth cubic curve E, and there is a ring surjection from this $Skly_3$ to a twisted homogeneous coordinate ring B = B(E) with kernel generated by the degree 3 element g. See Figure 2.8 for a depiction.

Question III.42. Let A be a connected graded ring for which there does not exist a corresponding twisted homogeneous coordinate ring. In other words, the point scheme data of A does not stabilize to a projective scheme. Determine conditions on A so that there is a ring surjection from A onto its point parameter ring P(A).

Now the techniques of [ATVdB90], specifically the construction of point scheme data and noncommutative coordinate rings, is typically applied to strongly noetherian rings.

Definition III.43. A k-algebra A is said to be strongly (left) noetherian if $A \otimes_k C$ is a left noetherian ring for all commutative noetherian k-algebras C.

In fact, we have a result reminiscent of Theorem II.16 for all strongly noetherian connected graded k-algebras.

Theorem III.44. [AZ01, Corollaries E4.11,E4.12] [RZ08, Theorem 1.1] Let A be a strongly noetherian connected graded k-algebra, which is generated in degree one. Then, we have that the following statements hold.

- 1. The point scheme X of A exists.
- There is a ring homomorphism from A to a canonically constructed twisted homogeneous coordinate ring B(X, L, σ), which is surjective in high degree. Moreover, the sheaf L is σ-ample and the kernel of this map is known.

Only recently have these methods been extended to a class of non-strongly noetherian rings, so-called *naïve blowups* [KRS05]. Such rings are in fact noetherian and serve as examples in the ongoing program of classifying noncommutative surfaces. Meanwhile, more peculiar algebras have recently been discovered in [RS] and essentially nothing is known about their geometry. This prompts the following project.

Question III.45. (Toby Stafford) Formulate a theory of noncommutative coordinate rings that encapsulates results on twisted homogeneous coordinate rings, point parameter rings, and naïve blowup rings constructed from the geometric data of a given connected graded noetherian algebra. Secondly, apply this theory to establish geometric structure for the rings of [RS].

This question is partially answered in recent work of Nevins and Sierra.

Theorem III.46. [NSa] Under several technical conditions we have that if A is a noetherian connected graded k-algebra, generated in degree one, then there exists a naïve blowup algebra S for which A surjects onto S in high degree.

It is also conjectured that Question III.42 should speak to a class of noncommutative graded rings properly containing the noetherian structures. Namely we consider the class of *coherent* rings, which is motivated by the recent work on *coherent noncommutative algebraic geometry* [Pio08, Pol05]. Their results thus far include the classification of noncommutative projective lines and the definition of a coordinate ring of a noncommutative coherent projective scheme (Definition III.51, Theorem III.54). Details are found in the next section.

Question III.47. (Lance Small, Paul Smith) Determine if the rings S_{deg} and $P(S_{deg})$ are coherent. If so, then determine whether $P(S_{deg})$ arises as a coordinate ring of a noncommutative coherent projective scheme in the sense of [Pio08].

We show in the next section that the degenerate Sklyanin algebras are indeed coherent, yet it is not known if this is true for $P(S_{deg})$.

3.5 On coherent noncommutative algebraic geometry

This section further addresses Question III.47 in the previous section. First we introduce Piontkovskii-Polishchuk's theory of *coherent noncommutative algebraic geometry*, beginning with the definition, some properties, and examples of coherent (graded) algebras. We then discuss the notion of projective schemes in terms of coherent noncommutative algebraic geometry. Lastly we prove that degenerate Sklyanin algebras (Definition III.2) are coherent (Proposition III.55), and provide some consequences afterward.

Definition III.48. [Lam99, §4G]

- A finitely generated left module M is coherent if every finitely generated submodule is finitely presented. Analogously a finitely generated graded module is graded coherent if every graded finitely generated submodule is finitely presented.
- A (graded) algebra A is left (graded) coherent if every (homogeneous) finitely generated left-sided ideal in A is finitely presented, i.e. A is a (graded) coherent as a left module over itself.
- We have analogous definitions for right coherence. Moreover an algebra is coherent if it is both left and right coherent.

Examples of (left) coherent rings include (left) noetherian rings and (left) semihereditary rings [Lam99, Example 4.46]. Moreover we have that the free algebras (on finitely many generators) are coherent [Pol05, Corollary 3.2].

Here are a couple of results that will help determine the coherence of a ring.

Lemma III.49. [Lam99, Example 4.61(c)] Let I is a homogeneous ideal of a ring A,

which is finitely generated as a left (right) ideal. If A is left (right) graded coherent, then A/I is left (right) coherent.

Lemma III.50. [Pio08, Proposition 3.2] Let B = A/I where the algebra B is left (right) Noetherian and the ideal I is free as a right (left) A-module. Then the algebra A is left (right) graded coherent.

Motivated by Artin and Zhang's approach to (noetherian) noncommutative algebraic geometry [AZ94], namely their construction of noncommutative schemes (see §2.1.5), Piontkovskii developed a version of a noncommutative projective scheme for the coherent setting [Pio08].

Definition III.51. [Pio08, $\S1$] Provided a coherent algebra A, a **coherent projective scheme** is given by the abstract triple

$$(A - c.qgr, \mathcal{A}, s),$$

where A-c.qgr is the quotient category of finitely presented (=graded coherent) left graded A-modules by finite-dimensional modules, \mathcal{A} is the image of A in A-c.qgr, and s is the autoequivalence of A-c.qgr induced by shift of grading.

The first non-trivial examples of this theory, namely coherent noncommutative projective lines, are given below.

Definition III.52. A graded algebra A is **regular** if it has finite global dimension d, and $\operatorname{Ext}^{q}(k, A) = \delta_{d,q} \cdot k$.

Theorem III.53. [Zha98, Theorem 0.1] A connected graded algebra is regular of global dimension 2 if and only if $A \cong k\{x_1, \ldots, x_n\}/(b)$ with:

1. $n \ge 2;$

- 2. If the x_i 's are labeled so that $1 \leq \deg x_1 \leq \cdots \leq \deg x_n$, then $\deg x_i + \deg x_{n-i}$ is a constant for all i,
- 3. There is a graded algebra automorphism α of $k\{x_1, \ldots, x_n\}$ so that

$$b = \sum_{i=1}^{n} x_i \alpha(x_{n-i}).$$

In fact, such algebras are noetherian if and only if n = 2 [Zha98, Corollary 1.2]. Piontkovskii proves that all of these algebras are graded coherent [Pio08, Theorem 4.3]. Now we consider the algebras of Theorem III.53 that are generated in degree one.

Theorem III.54. [Pio08, Theorem 1.4, Proposition 1.5] The coherent noncommutative \mathbb{P}^1 are classified. These are precisely the regular algebras, finitely generated in degree one, of global dimension 2.

We aim to include degenerate Sklyanin algebras and eventually point parameter rings P(A) (for arbitrary connected graded algebras A) into the theory of coherent noncommutative algebraic geometry. The following result is the first step towards this goal.

Proposition III.55. The degenerate Sklyanin algebras are graded coherent.

Proof. Recall that S(a, b, c) from Definition III.1 is a degenerate Sklyanin algebra when

$$[a:b:c] \in \{[0:0:1], [0:1:0], [1:0:0]\} \cup \{[a:b:c] \mid a^3 = b^3 = c^3 = 1\}.$$

In the first three cases, S is a monomial algebra defined by finitely many homogeneous relations. Thus by [Pio96], we have the result for these cases.

Now assume that $a^3 = b^3 = c^3 = 1$. Since $S(a, b, c) \cong S(1, \frac{b}{a}, \frac{c}{a})$, we take a = 1 without loss of generality. Furthermore recall from Proposition III.7 that

$$S(1,b,c) \cong \frac{k\{x,y\}[z;\alpha,\delta]}{(\Omega)}$$

where Ω is a normal element of $k\{x, y\}[z; \alpha, \delta]$. Now by Lemma III.49, it suffices to show that $k\{x, y\}[z; \alpha, \delta]$ is (left) graded coherent.

Let $A := k\{x, y\}[z; \alpha, \delta]$, let I = AxA + AyA, and B := k[z]. Note that B is left noetherian. Moreover as

$$zx = -bxz - cy^2$$
 and $zy = -b^2yz - b^2cx^2$,

one sees I = xA + yA which is free as a right A-module. Therefore A is left graded coherent due to Lemma III.50. A similar argument also yields right coherence. \Box

To illustrate further consequences, we refer to [Lam99, §4F-4H] for results about the flatness of modules over S_{deg} . In particular, we have the following result of Chase.

Theorem III.56. [Cha60, Theorem 2.1] A ring A is left coherent if and only if the direct product of any family of flat right A-modules is flat.

One may compare this to a result of Papp and Bass: A ring A is left noetherian if and only if the direct sum of any family of injective left A-modules is injective [GW04, Theorem 5.23].

We also point out that there is work in progress with S. Paul Smith in realizing the category S_{deg} -qgr as equivalent to a module category over a direct limit of finitedimensional algebras [SW].

CHAPTER IV

Representation theory of deformed Sklyanin algebras On the three-dimensional Sklyanin algebras $Skly_3$

let us remind the reader of the definition of three-dimensional Sklyanin algebras. in particular that deformed Sklyanin algebras arise as superpotential algebras. Now We begin by introducing the physics background relevant to this thesis. We will see the study of deformations of the N = 4 super Yang-Mills theory in four dimensions. I.12=IV.2). Such a task is motivated by current research in string theory, namely finite-dimensional representations of so-called deformed Sklyanin algebras (Definition This chapter and Chapter 5 present results towards the classification of irreducible

three relations: S(a, b, c) or $Skly_3$, are generated by three noncommuting variables x, y, z, subject to Definition IV.1. The three-dimensional Sklyanin algebras, denoted by

$$ayz + bzy + cx^{2} = 0$$
$$azx + bxz + cy^{2} = 0$$
$$axy + byx + cz^{2} = 0$$

for $[a:b:c] \in \mathbb{P}^2_k \smallsetminus \mathfrak{D}$ where

$$\mathfrak{D} = \{ [0:0:1], [0:1:0], [1:0:0] \} \cup \{ [a:b:c] \mid a^3 = b^3 = c^3 = 1 \}.$$

We then define the deformations of $Skly_3$ that are the focus of this chapter.

Definition IV.2. For i = 1, 2, 3, let a, b, c, d_i, e_i be scalars in k with $[a:b:c] \notin \mathfrak{D}$ (Definition IV.1). The **deformed Sklyanin algebras**, S_{def} , are generated by three noncommuting variables x, y, z, subject to three relations:

$$ayz + bzy + cx^{2} + d_{1}x + e_{1} = 0$$
$$azx + bxz + cy^{2} + d_{2}y + e_{2} = 0$$
$$axy + byx + cz^{2} + d_{3}z + e_{3} = 0.$$

In fact the name is motivated by the observation that, for (almost all) parameters (a, b, c), the ring S_{def} is a PBW-deformation is $Skly_3$ [BT07][EG, §3].

As we will explain in §4.2, the study of of the representation theory of S_{def} falls into two tasks, one of which is such a study for $Skly_3$. Hence the last section of this chapter is dedicated to the classification of irreducible finite-dimensional representations of three-dimensional Sklyanin algebras.

4.1 Physical Motivation

Here we discuss the physical motivation behind the task of classifying simple finite-dimensional S_{def} -modules. In particular we point out connections to work of Berenstein, Jejjala, and Leigh in the field of supersymmetric string theory. To introduce string theory in general, we use Zwiebach's description of the investigation of open strings in a system of *M D-branes* [Zwi09, §§1.3, 15.5].

"Just as the strings of a violin are held stretched by pegs, the D-branes hold fixed the endpoints of the open strings whose lowest vibrational modes could represent the particles of the *Standard Model*."

Furthermore, we see that the interaction of open strings in the 'world' of M D-branes is in accordance with a certain Yang-Mills theory: "On the world volume of M coincident D-branes, there are U(M) gauge fields [...] whose low energy dynamics is governed by a Yang-Mills theory with gauge group U(M)."

Now the work in this chapter is prompted by the article [BJL00], in which the authors study various deformations of a *super* Yang-Mills theory with gauge group U(M). More precisely the vacua, or solutions of *F*-term constraints, of these deformed theories are desired. Algebraically, this boils down to finding $M \times M$ matrix solutions to cyclic derivatives of a superpotential.

Definition IV.3. [EG, §3.1] Let V be a k-vector space with basis x_1, \ldots, x_n and let $F = T(V) = k\{x_1, \ldots, x_n\}$ be the corresponding free algebra. The commutator quotient space $F_{cyc} = F/[F, F]$ is a k-vector space with the natural basis formed by cyclic words in the alphabet x_1, \ldots, x_n . Elements of F_{cyc} are referred to as **superpotentials** (or as **potentials** in some articles).

Let $\Phi = \sum_{\{i_1, i_2, \dots, i_r\} \in I} x_{i_1} x_{i_2} \cdots x_{i_r} \in F_{cyc}$ for some indexing set *I*. For each $j = 1, \dots, n$, one defines $\partial_j \Phi \in F$, the corresponding partial derivative of Φ given by:

$$\partial_j \Phi = \sum_{\{s \mid i_s = j\}} x_{i_s+1} x_{i_s+2} \cdots x_{i_r} x_{i_1} x_{i_2} \cdots x_{i_s-1} \ \in F.$$

The elements $\partial_j \Phi$ are called **cyclic derivatives** of Φ .

The algebra $F/(\partial_j \Phi)_{j=1,...,n}$ is called a **superpotential algebra**.

Example IV.4. Consider $F = k\{x, y, z\}$ with the superpotential $\Phi = xyz - xzy$. We get that

$$\partial_x \Phi = yz - zy, \ \partial_y \Phi = zx - xz, \ \partial_z \Phi = xy - yx.$$

Hence the polynomial algebra k[x, y, z] arises as the superpotential algebra $F/(\partial \Phi)$.

We return to string theory with the following remark.
Remark IV.5. The superpotential relevant to the work of Berenstein et al. (after rescaling parameters) is $\Phi = \Phi_{marg} + \Phi_{rel}$ where:

$$\Phi_{marg} = axyz + byxz + \frac{c}{3}(x^3 + y^3 + z^3),$$

$$\Phi_{rel} = \frac{d_1}{2}x^2 + \frac{d_2}{2}(y^2 + z^2) + (e_1x + e_2y + e_3z).$$

Here the two superpotentials respectively correspond to the moduli spaces of marginal and relevant deformations of a N=4 super Yang-Mills theory.

Observe that if $[a:b:c] \notin \mathfrak{D}$, then the corresponding superpotential algebra is precisely a deformed Sklyanin algebra with parameters $d_2 = d_3$. Hence the classification of *M*-dimensional representations of S_{def} , or more specifically the study of simple finite-dimensional S_{def} -modules, has physical implications.

4.2 Strategy to classify simple finite-dimensional modules over S_{def}

Let us consider the following notation.

Notation. Given an algebra A, let $\operatorname{Repr}_{<\infty}(A)$ be the set of isomorphism classes of finite-dimensional left A-modules and $\operatorname{Repr}_m(A)$ be the objects of dimension m. We denote by $\operatorname{Simp}_{<\infty}(A)$ and $\operatorname{Simp}_m(A)$, the respective subsets of simple modules.

We then formulate our aim as follows.

Classify the simple finite-dimensional modules over S_{def} .

The strategy to achieving the goal is as follows. First, homogenize the relations of S_{def} with a central element w. Then, study the representation theory of the resulting *central extension* D of $Skly_3$. Reasons for this construction are made clear in Proposition IV.7 below.

Definition IV.6. For i = 1, 2, 3, let a, b, c, d_i, e_i be scalars in k with $[a : b : c] \notin \mathfrak{D}$ (Definition IV.1). The **central extension** D of $Skly_3$ is generated by three variables x, y, z and a central element w, subject to the relations:

$$ayz + bzy + cx^{2} + d_{1}xw + e_{1}w^{2} = 0$$

$$azx + bxz + cy^{2} + d_{2}yw + e_{2}w^{2} = 0$$

$$axy + byx + cz^{2} + d_{3}zw + e_{3}w^{2} = 0$$

$$xw - wx = yw - wy = zw - wz = 0.$$

Proposition IV.7. Simple finite-dimensional modules over $D(a, b, c, d_i, e_i)$ are precisely those over $Skly_3(a, b, c)$ or over $S_{def}(a, b, c, d_i, e_i)$.

In other words, our objective can be reformulated in terms of studying the simple finite-dimensional modules over the graded algebras $Skly_3$ and D.

Proof. For one direction, consider the surjection $\pi : D \twoheadrightarrow D/D(w - \lambda)$ for $\lambda \in k$. If $\lambda = 0$, then $D/D(w - \lambda) \cong Skly_3$. Otherwise we can rescale to assume that $\lambda = 1$ and so $D/D(w - \lambda) \cong D/D(w - 1) \cong S_{def}$. Let S denote either $Skly_3$ or S_{def} , and let M be a simple finite-dimensional left S-module. Then we can construct from M a left D-module with D-action given by $d * m = \pi(d) \cdot m$. This module remains simple and finite-dimensional.

Conversely, given a simple finite-dimensional left D-module M, we will show that it belongs to either $\operatorname{Simp}_{<\infty}(Skly_3)$ or $\operatorname{Simp}_{<\infty}(S_{def})$. Since k[w] embeds into D, we can consider $_{k[w]}M$, a finite-dimensional module over the PID, k[w]. Hence $_{k[w]}M$ is torsion and there exists a nonzero polynomial f(w) in $\operatorname{ann}_{k[w]}M$. Now k[w]/(f(w))embeds in $\operatorname{End}_D M$, a division ring by Schur's lemma. Thus k[w]/(f(w)) embeds into a domain, and f(w) is irreducible and of the form $w - \lambda$ where $\lambda \in k$. If $\lambda = 0$, then $M \in \operatorname{Simp}_{<\infty}(Skly_3)$. Otherwise $D/D(w - \lambda) \cong D/D(w - 1)$ by rescaling, and we have that M belongs to $\operatorname{Simp}_{<\infty}(S_{def})$.

Hence we proceed to study the simple finite-dimensional modules of $Skly_3$ in the

next section. Results on the analysis of $\operatorname{Simp}_{<\infty}D$ is presented in Chapter 5.

4.3 Irreducible finite dimensional representations of Sklyanin algebras

We study irreducible finite-dimensional representations of the three-dimensional Sklyanin algebra (Definition IV.1), denoted by $Skly_3$, S(a, b, c), or S. Before we begin, we point out that the representation theory of $Skly_3$ has also physical consequences.

Remark IV.8. We see specifically from Remark IV.5 that the algebra S(a, b, c) arises as the superpotential algebra, $k\{x, y, z\}/(\partial \Phi_{marg})$. The corresponding marginal deformed theories are studied in [BJL00, §4.6.1], and in our language, the authors sought to understand the set $\operatorname{Simp}_{<\infty}S(a, b, c)$. This task was not achieved.

Recall from §2.1.3 that $Skly_3$ is associated to the geometric data (E, \mathcal{L}, σ) where

$$E = \mathbb{V}((abc)(x^3 + y^3 + z^3) - (a^3 + b^3 + c^3)(xyz)) \stackrel{i}{\subset} \mathbb{P}^2_k,$$

 \mathcal{L} is the invertible sheaf $i^*\mathcal{O}_{\mathbb{P}^2}(1)$ on E, and σ is the automorphism of E induced by the shift functor on point modules of S.

Assumption IV.9. We restrict our attention to the case that E is smooth curve, to say that the parameters a, b, c of S satisfy $abc \neq 0$ with $(3abc)^3 \neq (a^3 + b^3 + c^3)^3$. Furthermore we assume that $\sigma \in \operatorname{Aut}(E)$ is given by translation by a point so that our definition of S(a, b, c) corresponds to the standard definition of a three-dimensional Sklyanin algebra, i.e. S is a type A Artin-Schelter regular algebra of dimension 3 [ATVdB90, 4.13].

Remark IV.10. Now the sheaf $\mathcal{L} = i^* \mathcal{O}_{\mathbb{P}^2}(1)$ on E has degree 3 due to a version of the Riemann-Roch theorem. Namely if \mathcal{L} is any line bundle on an irreducible,

reduced projective curve C, then

$$\dim_k H^0(C,\mathcal{L}) - \dim_k H^1(C,\mathcal{L}) = \deg \mathcal{L} + 1 - \dim_k H^1(C,\mathcal{O}_C).$$

We also remind the reader that there exists a central, regular element g, homogeneous of degree 3 so that S/Sg is isomorphic to a twisted homogeneous coordinate ring $B = B(E, \mathcal{L}, \sigma)$ (Theorem II.16).

The results of this section are given as follows. The simple 1-dimensional Smodules are determined in Lemma IV.11. We then consider the order of σ , and investigate $\operatorname{Simp}_{<\infty}S$ when $|\sigma| = \infty$ (Proposition IV.13), and when $|\sigma| < \infty$ afterward. For the $|\sigma| < \infty$ case in particular, we also take into consideration the subcases where simple finite-dimensional S-modules are either g-torsionfree or g-torsion. These results are reported in Propositions IV.16 and IV.18, respectively.

Recall the terminology from Definition II.31.

Lemma IV.11. The set $Simp_1(S)$ solely consists of the trivial module: $\{S/S_+\}$.

Proof. We know that:

$$\operatorname{Simp}_{1}(S) = \begin{cases} S \xrightarrow{\phi} k & a\beta\gamma + b\gamma\beta + c\alpha^{2} = 0, \\ x \mapsto \alpha & a\gamma\alpha + b\alpha\gamma + c\beta^{2} = 0, \\ \vdots & \\ y \mapsto \beta & a\alpha\beta + b\beta\alpha + c\gamma^{2} = 0, \\ z \mapsto \gamma & [a:b:c] \in \mathbb{P}^{2}_{k} \smallsetminus \mathfrak{D} \end{cases}$$

By Assumption IV.9, $abc \neq 0$ so let $\lambda := -(a+b)/c$. Now the conditions on (α, β, γ) above are equivalent to:

$$\{\lambda\beta\gamma = \alpha^2, \ \lambda\alpha\gamma = \beta^2, \ \lambda\alpha\beta = \gamma^2\},\$$

in which we have that $(\lambda^3 - 1)\alpha^2\beta = 0$. If $\lambda^3 \neq 1$, then we get that $\alpha = \beta = \gamma = 0$ and Simp₁(S) = {S/S₊}. Else if $\lambda^3 = 1$, then $a^3 + b^3 + c^3 = -3ab(a + b)$. This implies that $(a^3 + b^3 + c^3)^3 = (3abc)^3$, which contradicts Assumption IV.9.

The next result describes $\operatorname{Simp}_{<\infty} S$ in the case that $|\sigma| = \infty$. First we require the following lemma.

Lemma IV.12. Let A be a finitely generated, locally finite, connected graded kalgebra. Take $M \in Simp_{<\infty}A$ and let P be the largest graded ideal contained in $ann_A(M)$. Then we have that GKdim(A/P) = 0 or 1. In particular, GKdim(A/P) = 0if and only if M is the trivial module A/A_+ .

Proof. Let $\overline{A} := A/P$. Note that $M \in \operatorname{Simp}_{<\infty}(\overline{A})$ so $\operatorname{ann}_{\overline{A}}(M)$ contains no homogeneous elements. Suppose by way of contradiction that $\operatorname{GKdim}(\overline{A}) > 1$. Then by Proposition II.27(5), we have that $\lim_{i\to\infty} \dim_k \overline{A}_i = \infty$. Since $\overline{A}/\operatorname{ann}_{\overline{A}}(M)$ is finite-dimensional by the Density Theorem [MR01, Theorem 0.3.6], we get that $\operatorname{ann}_{\overline{A}} M \cap \overline{A}_i \neq 0$ for all i > 0. Hence there exists a homogeneous element in $\operatorname{ann}_{\overline{A}}(M)$, which contradicts the maximality of P. Now $\operatorname{GKdim}(\overline{A}) = 0$ or 1 by Proposition II.27(4).

If $M = A/A_+$, then $P = A_+$, and $\operatorname{GKdim}(A/P)=0$ as A/P is finite-dimensional (Proposition II.27(3)). Conversely, $\operatorname{GKdim}(A/P) = 0$ implies that $\overline{A} = A/P$ is finitedimensional and connected graded. Since $\bigcap_{n \in \mathbb{N}} (\overline{A}_+)^n = 0$, we get that $(\overline{A}_+)^n = 0$ for some $n \in N$. Hence $\overline{A}_+ = 0$ by primality, and so $P = A_+$ with M being the trivial module.

Proposition IV.13. The set $Simp_{<\infty}(S)$ equals $\{S/S_+\}$ when $|\sigma| = \infty$.

Proof. According to Lemma II.32, we have that simple finite dimensional modules over S are quotients of irreducible objects in S-qgr. For $|\sigma| = \infty$, the set of nontrivial,

g-torsionfree quotients of S-qgr is empty due to [ATVdB91, Propositions 7.5, 7.9].

On the other hand, the set of g-torsion irreducible objects of S-qgr equals the set of irreducible objects of B-qgr where B is the twisted homogeneous coordinate ring $B(E, \mathcal{L}, \sigma)$ (Theorem II.16). This is precisely the set of B-point modules (Theorem II.14). Take $M \in \operatorname{Simp}_{<\infty} B$ and let P be the largest graded ideal contained in $\operatorname{ann}_B M$. By Lemma IV.12, we have that $\operatorname{GKdim}(B/P) \leq 1$. If equal to 1, then $\operatorname{Kdim}(B/P)=1$, which is a contradiction as $|\sigma| = \infty$ and B is projectively simple in this case [RRZ06]. Thus $\operatorname{GKdim}(B/P)=0$, and again by Lemma IV.12 we know that $M = B/B_+$. \Box

We now determine the dimensions of simple finite-dimensional left S-modules in terms of the PI degree of S. We require the following preliminary result.

Notation. Let g denote the central homogeneous degree 3 element of S such that $S/Sg \cong B$ (Theorem II.16). Put $\Lambda \coloneqq S[g^{-1}]$. We have that S is module finite over its center precisely when $|\sigma| = n < \infty$ (Theorem I.18). Thus S is PI and let p denote the PI degree of S in this case.

Note that the result below is mentioned in [LB95a, §3], yet the details are omitted. We provide the full proof here.

Lemma IV.14. In the case of $|\sigma| < \infty$, the $PIdeg(\Lambda_0)$ equals p/d where $d = gcd(3, |\sigma|)$. *Proof.* The center of S, denoted by Z(S), has four generators: g of degree 3 and three others of degree $|\sigma|$ [ST94, Theorem 3.7]. Let u be any one of the degree $|\sigma|$ elements.

In the case of d = 1, recall that $\Lambda \coloneqq S[g^{-1}]$ and consider the following notation:

$$\Lambda_0' \coloneqq \Lambda_0[g^{\pm 1}], \qquad \Lambda_0'' \coloneqq \Lambda_0'[u^{\pm 1}], \qquad \Lambda' \coloneqq \Lambda[u^{\pm 1}].$$

Note that

Adjoining a central element does not alter PI degree so we have that:

(†)
$$\operatorname{PIdeg}(\Lambda_0) = \operatorname{PIdeg}(\Lambda'_0) = \operatorname{PIdeg}(\Lambda''_0);$$

(‡)
$$\operatorname{PIdeg}(\Lambda') = \operatorname{PIdeg}(S) =: p.$$

We now show that $\Lambda_0'' = \Lambda'$. Take $a \in \Lambda'$ of degree $m \in \mathbb{Z}$. Then there exists $m_1 \in \mathbb{Z}_{\geq 0}$ so that $au^{m_1} \in \Lambda$. As gcd(3, deg(u)) = 1, we have that $au^{m_1+m_2} \in \Lambda_{3\mathbb{Z}}$ for $m_2 = 0, 1$, or 2. Now $au^{m_1+m_2}g^{-t} \in \Lambda_0$ for some $t \in \mathbb{Z}_{\geq 0}$. Thus $a = (au^{m_1+m_2}g^{-t})(u^{-(m_1+m_2)}g^t) \in \Lambda_0''$. Thus with (\dagger, \ddagger) above, we conclude that:

$$\operatorname{PIdeg}(\Lambda_0) = \operatorname{PIdeg}(\Lambda_0'') = \operatorname{PIdeg}(\Lambda') = p$$

as desired.

On the other hand, for d = 3 we show that:

$$\frac{p}{3} = \operatorname{PIdeg}(S_{3\mathbb{Z}}) = \operatorname{PIdeg}(\Lambda_0).$$

The second equality follows from two observations:

(*)
$$\Lambda_0 = \{sg^{-m} : m \in \mathbb{Z}, s \in S_{3m}\} = S_{3\mathbb{Z}}[g^{-1}]_0,$$

(**) $S_{3\mathbb{Z}}[g^{-1}] = \{sg^{-m} : s \in S_{3d}; d, m \in \mathbb{Z}\} = S_{3\mathbb{Z}}[g^{-1}]_0[g^{\pm 1}].$

Now we have that:

$$\operatorname{PIdeg}(S_{3\mathbb{Z}}) \stackrel{(**)}{=} \operatorname{PIdeg}(S_{3\mathbb{Z}}[g^{-1}]_0[g^{\pm 1}]) = \operatorname{PIdeg}(S_{3\mathbb{Z}}[g^{-1}]_0) \stackrel{(*)}{=} \operatorname{PIdeg}(\Lambda_0).$$

It suffices to show $p = 3 \cdot \operatorname{PIdeg}(S_{3\mathbb{Z}})$ on the level of fraction fields, i.e. we must show that $\operatorname{PIdeg}(FS) = 3(\operatorname{PIdeg}(FS_{3\mathbb{Z}}))$ for $F = \operatorname{Frac}(Z(S)) = Z(S)[C^{-1}]$ with $C = Z(S) \setminus \{0\}$. We also note that

$$F \subseteq FS_{3\mathbb{Z}} = S_{3\mathbb{Z}}[C^{-1}] \subseteq FS = S[C^{-1}].$$

Since S and $S_{3\mathbb{Z}}$ are domains, by Posner's theorem (Theorem II.42) we have that $FS_{3\mathbb{Z}}$ and FS are division rings. Let r denote the PIdegree of $FS_{3\mathbb{Z}}$. By Posner's

theorem and [MR01, Lemma 13.3.4], we know that there exists a maximal subfield K of $FS_{3\mathbb{Z}}$ so that $\dim_K(FS_{3\mathbb{Z}}) = r$. Since $\dim_{S_{3\mathbb{Z}}} S = 3$, we have that $\dim_K(FS) = 3r$. Using the regular representation,

$$FS \hookrightarrow \operatorname{End}_K({}_KFS) \cong \operatorname{Mat}_{3r}(K),$$

we get that $\operatorname{PIdeg}(FS) \leq 3r$.

For the other inequality, pick a maximal subfield L of FS containing K. Then

$$3r = \dim_K FS = (\dim_K L)(\dim_L FS) = (\dim_K L)(\operatorname{PIdeg} FS).$$

The last equality is again due to Posner's theorem and [MR01, Lemma 13.3.4]. Therefore, PIdeg(FS) divides 3r.

Since the degree of central elements of Z(S) are in $3\mathbb{Z}$, then

$$F = Z(S)[C^{-1}] \subseteq Z(FS_{3\mathbb{Z}}) = Z(S_{3\mathbb{Z}}[C^{-1}]).$$

Now we know that

$$(\dim_F FS_{3\mathbb{Z}})^{1/2} \geq (\dim_{Z(FS_{3\mathbb{Z}})} FS_{3\mathbb{Z}})^{1/2} = \operatorname{PIdeg} FS_{3\mathbb{Z}}$$

Therefore

$$\begin{aligned} 3^{1/2} (\operatorname{PIdeg} FS_{3\mathbb{Z}}) &\leq (\dim_F FS_{3\mathbb{Z}})^{1/2} \cdot 3^{1/2} \\ &= (\dim_F FS_{3\mathbb{Z}})^{1/2} (\dim_{FS_{3\mathbb{Z}}} FS)^{1/2} \\ &= \operatorname{PIdeg} FS. \end{aligned}$$

In other words, $3^{1/2} \cdot r \leq \text{PIdeg}FS$. Since PIdegFS divides 3r and PIdegFS > r, we have that PIdegFS = 3r. Thus

$$\operatorname{PIdeg} S = \operatorname{PIdeg} FS = 3 \cdot (\operatorname{PIdeg} FS_{3\mathbb{Z}}) = 3 \cdot (\operatorname{PIdeg} S_{3\mathbb{Z}}),$$

and $\operatorname{PIdeg} S = 3 \cdot \operatorname{PIdeg} \Lambda_0$ in the case of $(3, |\sigma|) \neq 1$.

Corollary IV.15. In this case $|\sigma| < \infty$, we have that the PI degree of S(a, b, c) is equal to $|\sigma_{abc}|$.

Proof. Let s denote the smallest integer such that σ^s fixes $[\mathcal{L}]$ in PicE. We know by [ATVdB91, Theorem 7.3] that the ring Λ_0 is Azumaya (Definition II.46) and s is its PI degree. Now suppose that $(3, |\sigma|) = 1$. Then $s = |\sigma|$ by [Art92, §5]. Therefore with Lemma IV.14, we have that PIdeg $S = \text{PIdeg } \Lambda_0 = s = |\sigma|$.

On the other hand, suppose that $|\sigma|$ is divisible by 3. We have that s is also the order of the automorphism η introduced in [ATVdB91, §5], due to [ATVdB91, Theorem 7.3] (or more explicitly by [AdJ, Lemma 5.5.5(i)]). Since \mathcal{L} has degree 3 (Remark IV.10), we know that η is σ^3 [AdJ, Lemma 5.3.6]. Thus

$$\operatorname{PIdeg} S = 3 \cdot \operatorname{PIdeg} \Lambda_0 = 3s = 3|\eta| = 3|\sigma^3| = 3 \cdot \frac{|\sigma|}{(3,|\sigma|)} = |\sigma|.$$

Now we consider two subcases of the classification of $\operatorname{Simp}_{<\infty}S$ for $|\sigma| < \infty$: first consisting of *g*-torsionfree modules in Proposition IV.16, and secondly consisting of *g*-torsion modules in Proposition IV.18.

Proposition IV.16. Take $|\sigma| < \infty$, and let M be a g-torsionfree simple finitedimensional left S-module. Then we have that $\dim_k M = p$ for $(3, |\sigma|) = 1$, and $p/3 \le \dim_k M \le p$ for $|\sigma|$ divisible by 3.

Proof. Let M be a g-torsionfree simple S-module of dimension $m < \infty$. By Lemma II.32 and Remark II.33, M is the quotient of some 1-critical graded S-module N with $\operatorname{mult}(N) \leq m$. Note that N is an irreducible object in S-qgr, which is also g-torsionfree. The equivalence of categories, $\operatorname{Irred}(S - \operatorname{qgr}) \sim \operatorname{Simp}_{<\infty}(\Lambda_0)$, from g-torsionfree [ATVdB91, Theorem 7.5] implies that N yields the object $N[g^{-1}]_0$ in $\operatorname{Simp}_{\infty}(\Lambda_0)$. Furthermore we have that $\dim_k N[g^{-1}]_0 = \operatorname{mult}(N)$ as follows.

Since N is 1-critical, by Remark II.23 and Definition II.24 we know that $\dim_k(N_j) = \operatorname{mult}(N)$ for j >> 0. Recall that g is homogeneous of degree 3. Note that $N[g^{-1}]_0 \cdot g^i \subseteq N_{3i}$ and moreover that $\dim_k(N_{3i}) = \operatorname{mult}(N)$ for i >> 0. Hence such an i, we have that:

$$\dim_k N[g^{-1}]_0 = \dim_k (N[g^{-1}]_0 \cdot g^i) \leq \dim_k N_{3i} = \operatorname{mult}(N).$$

Conversely $N_{3i} \cdot g^{-i} \subseteq N[g^{-1}]_0$. Hence for i >> 0 we have that:

$$\operatorname{mult}(N) = \dim_k(N_{3i}) = \dim_k(N_{3i} \cdot g^{-i}) \leq \dim_k N[g^{-1}]_0.$$

Thus $\dim_k N[g^{-1}]_0 = \operatorname{mult}(N)$ as desired.

Since Λ_0 is Azumaya, the Λ_0 -module $N[g^{-1}]_0$ has dimension equal to $\operatorname{PIdeg}(\Lambda_0)$ (Theorem II.51). Therefore $\operatorname{PIdeg}(\Lambda_0) = \operatorname{mult}(N) \leq m$. On the other hand, we know by Proposition II.52 that $m \leq p$. Hence

$$\operatorname{PIdeg}(\Lambda_0) \leq \dim_k M \leq p.$$

In the case of $(3, |\sigma|) = 1$, Lemma IV.14 implies that $\operatorname{PIdeg} \Lambda_0 = p$. Thus $\dim_k M = p$ in this case. For $|\sigma|$ divisible by 3, we can conclude from Lemma IV.14 that $p/3 \leq \dim_k M \leq p$.

Corollary IV.17. In the case of $|\sigma| < \infty$, a generic finite-dimensional simple, gtorsionfree, left S-module M has dimension equal to $|\sigma|$.

Proof. We have an inclusion of sets $\underset{g-\text{torsionfree}}{\text{simp}_{<\infty}S} \subset \underset{g-\text{torsionfree}}{\text{Simp}_{<\infty}S[g^{-1}]}$ given by $M \mapsto M[g^{-1}]$, where $M[g^{-1}]$ remains simple by [GW04, Corollary 10.16] and $\dim_k M[g^{-1}] = \dim_k M$ by [GW04, Exercise 10K]. Since inverting central elements does not alter PI degree, we have that $\text{PIdeg}(S[g^{-1}]) = \text{PIdeg}(S)$. Now we can conclude the result by applying Corollary IV.15 and Propositions II.52, II.54. □ To study g-torsion S-modules, let us consider the following notation.

Notation. Recall that S/Sg is isomorphic to a twisted homogeneous coordinate ring *B* (Theorem II.16). Since *B* is a homomorphic image of *S* which is PI for $|\sigma| < \infty$, we have that *B* is also PI in this case. Let *q* denote the PI degree of *B*.

Proposition IV.18. For the case of $|\sigma| = n < \infty$, let M be a non-trivial g-torsion simple finite-dimensional left S-module. Then $\dim_k M = q$.

Proof. For a graded ring $A = \bigoplus_{i \in \mathbb{N}} A_i$, let $\operatorname{Simp}_{<\infty}^o A$ denote the set of simple finitedimensional left A-modules that are not annihilated by the irrelevant ideal $A_+ = \bigoplus_{i \ge 1} A_i$. Notice that

$$\underset{g-\text{torsion}}{\text{Simp}_{<\infty}S} = \text{Simp}_{<\infty}B.$$

Thus it suffices to show that all modules in $\operatorname{Simp}_{<\infty}^{o} B$ have maximal dimension, which is equal to q by Proposition II.52. We proceed by establishing the following claims.

<u>Claim 1:</u> We can reduce the task to studying $\operatorname{Simp}_{<\infty}^{o}\overline{B}$ for \overline{B} some factor of B. Furthermore $\operatorname{Simp}_{<\infty}^{o}\overline{B} = \operatorname{Simp}_{<\infty}^{o}C$ for $C = B(Y_n, \mathcal{L}|_{Y_n}, \sigma|_{Y_n})$, the twisted homogeneous coordinate ring of an irreducible $|\sigma|$ -orbit of E.

<u>Claim 2:</u> The ring C is isomorphic to a graded matrix ring.

<u>Claim 3:</u> The modules of $\operatorname{Simp}_{<\infty}^{o}C$ all have maximal dimension which is equal to $\operatorname{PIdeg}(C) = |\sigma|.$

Proof of Claim 1: Take $M \in \operatorname{Simp}_{<\infty}^{o} B$ and let P be the largest graded ideal contained in $\operatorname{ann}_{B}M$. Note that P is a prime ideal as follows. If not, then there exists graded ideals I, J strictly containing P, with $IJ \subseteq P$. Now IJM = 0, yet $JM \neq 0$. Hence by the simplicity of M, we know that JM = M. So IM = 0, which prompts a contradiction. Set $\overline{B} = B/P$. We have by Lemma IV.12 that $\operatorname{GKdim}(\overline{B})=1$, so $\operatorname{Kdim}(\overline{B})=1$ [MR01, 8.3.18]. We also have by [AS95, Lemma 4.4] that the height one prime P of B corresponds to an σ -irreducible maximal closed subset of E. Namely as σ is given by translation, P is associated to a σ -orbit of $|\sigma|$ points of E, an orbit denoted by Y_n . Here $n := |\sigma|$. Now by [BRS10, proof of Proposition 3.8(1)], we have a natural homomorphism:

$$\phi: B \longrightarrow B(Y_n, \mathcal{L}|_{Y_n}, \sigma|_{Y_n}) =: C$$

given by restrictions of sections, with $\ker(\phi) = P$. The map ϕ is also surjective in high degree, whence \overline{B} is isomorphic to C in high degree.

Since B and C are PI, we have by Kaplansky's theorem [MR01, Theorem 13.3.8] that for each of these rings: the maximal spectrum equals the primitive spectrum. Thus it suffices to verify the following subclaim.

<u>Subclaim</u>: For a graded k-algebra $A = \bigoplus_{i \ge 0} A_i$, let max^oA denote the set of maximal ideals of A not containing the irrelevant ideal A_+ . Then there is a bijective correspondence between max^oC and max^o \overline{B} , given by $I \mapsto \overline{B} \cap I$. Moreover $\overline{B}/(\overline{B} \cap I) \cong C/I$. *Proof of Subclaim*: We know that $\overline{B}_{\ge m} = C_{\ge m}$ for m >> 0. Therefore as B is generated in degree 1, for any such m:

$$J := (\overline{B}_+)^m = \overline{B}_{\geq m} = C_{\geq m} \supseteq (C_+)^m$$

is a common ideal of \overline{B} and C. (The last inclusion may be strict as C need not be generated in degree 1.)

For one inclusion, let $I \in \max^{o} C$. We want to show that $\overline{B} \cap I \in \max^{o} \overline{B}$. Note that $\overline{B}/(\overline{B} \cap I) \cong (\overline{B} + I)/I$ as rings. Furthermore

$$C \supseteq \overline{B} + I \supseteq J + I \supseteq (C_+)^m + I = C.$$

The last equality is due to C_+ and I being comaximal in C. Hence $C = \overline{B} + I$ and $\overline{B}/(\overline{B} \cap I) \cong C/I$ is a simple ring. Consequently $\overline{B} \cap I \in \max^o \overline{B}$.

Conversely, take $M \in \max^{o} \overline{B}$ and we want to show that $M = \overline{B} \cap Q$ for some $Q \in \max^{o} C$. We show that $CMC \neq C$. Suppose not, then

$$(\overline{B}_+)^{2m} = J^2 = JCJ = JCMCJ = JMJ \subseteq M$$

which implies that $\overline{B}_+ \subseteq M$ as M is prime. This is a contradiction. Now CMC is contained in some maximal ideal Q of C. Since $M \in \max^o \overline{B}$, we have that $Q \in \max^o C$; else $M \subseteq Q \cap \overline{B} = \overline{B}_+$. By the last paragraph, we know that $\overline{B} \cap Q \in \max^o \overline{B}$ with $\overline{B} \cap Q = M \in \max \overline{B}$.

Thus Claim 1 is verified and so it suffices to examine $\operatorname{Simp}_{<\infty}^{o}C$.

Proof of Claim 2: Recall that $n := |\sigma|$. We will show that C is isomorphic to the graded matrix ring:

$$R \coloneqq \left(\begin{array}{cccccc} T & x^{n-1}T & x^{n-2}T & \dots & xT \\ xT & T & x^{n-1}T & \dots & x^2T \\ x^2T & xT & T & \dots & x^3T \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ x^{n-1}T & x^{n-2}T & x^{n-3}T & \dots & T \end{array} \right)$$

with $T = k[x^n]$. Say $Y_n := \{p_1, \dots, p_n\}$, the $|\sigma|$ -orbit of n distinct points p_i . Let $\mathcal{L}' \coloneqq \mathcal{L}|_{Y_n}$ and $\sigma' \coloneqq \sigma|_{Y_n}$. Reorder the $\{p_i\}$ to assume that $\sigma'(p_i) = p_{i+1}$ and $\sigma'(p_n) = p_1$ for $1 \le i \le n-1$. Now by Definition II.9, $C = \bigoplus_{d \ge 0} H^0(Y_n, \mathcal{L}'_d)$ where $\mathcal{L}'_0 = \bigoplus_{i=1}^n \mathcal{O}_{p_i}$, and $\mathcal{L}'_1 = \bigoplus_{i=1}^n \mathcal{O}_{p_i}(1)$, and $\mathcal{L}'_d = \mathcal{L}' \otimes_{\mathcal{O}_{Y_n}} (\mathcal{L}')^{\sigma'} \otimes_{\mathcal{O}_{Y_n}} \dots \otimes_{\mathcal{O}_{Y_n}} (\mathcal{L}')^{\sigma'^{d-1}}$.

Note that

$$(\mathcal{L}')^{\sigma} = \mathcal{O}_{p_n}(1) \oplus \mathcal{O}_{p_1}(1) \oplus \cdots \oplus \mathcal{O}_{p_{n-1}}(1) \cong \mathcal{L}'.$$

Therefore $\mathcal{L}'_d \cong (\mathcal{L}')^{\otimes d}$. If i = j, then $\mathcal{O}_{p_i}(1) \otimes \mathcal{O}_{p_j}(1) \cong \mathcal{O}_{p_i}(2)$. Otherwise the sheaf $\mathcal{O}_{p_i}(1) \otimes \mathcal{O}_{p_j}(1)$ has empty support. Hence $(\mathcal{L}')^{\otimes d} \cong \bigoplus_{i=1}^n \mathcal{O}_{p_i}(d)$. So C shares the same k-vector space structure as a sum of n polynomial rings in one variable, say $k[u_1] \oplus \cdots \oplus k[u_n]$.

We next study the multiplication of C; write C_i as $k \cdot u_1^i \oplus \cdots \oplus k \cdot u_n^i$. We define the multiplication $C_i \times C_j \to C_{i+j}$ on basis elements (u_1^k, \ldots, u_n^k) for k = i, j, which will extend to C by linearity. Moreover all of the following sums in indices are taken modulo n.

Observe that

$$C_i \times C_j = H^0(Y_n, \mathcal{L}'_i) \otimes H^0(Y_n, \mathcal{L}'_j)$$
$$= H^0(Y_n, \mathcal{L}'_i) \otimes H^0(Y_n, (\mathcal{L}'_j)^{(\sigma')^i})$$

Recall the multiplicative structure of twisted homogeneous coordinate rings (Definition II.9). Hence the multiplication C is defined as

$$(u_1^i, \ldots, u_n^i) * (u_1^j, \ldots, u_n^j) = (u_1^i u_{1-i}^j, \ldots, u_n^i u_{n-i}^j).$$

We now show that C is isomorphic to the graded matrix ring R. Define a map $\phi: C \to R$ by

$$(u_1^i, \dots, u_n^i) \mapsto \begin{pmatrix} \operatorname{Row}(u_1^i) \\ \vdots \\ \operatorname{Row}(u_n^i) \end{pmatrix} =: M_i.$$

Here $\operatorname{Row}(u_l^i)$ denotes the row with the entry u_l^i in column l-i modulo n, and zeros entries elsewhere. In other words, the matrix M_i has a degree i entry in positions (l, l-i) for $l = 1, \ldots, n$ and zeros elsewhere. We see that $M_i \in R$.

The map ϕ is the ring homomorphism as follows. First note that

$$\phi((u_{1}^{i},\ldots,u_{n}^{i})*(u_{1}^{j},\ldots,u_{n}^{j})) = \phi((u_{1}^{i}u_{1-i}^{j},\ldots,u_{n}^{i}u_{n-i}^{j})) = \begin{pmatrix} \operatorname{Row}(u_{1}^{i}u_{1-i}^{j}) \\ \vdots \\ \operatorname{Row}(u_{n}^{i}u_{n-i}^{j}) \end{pmatrix}.$$

Here the entry $u_l^i u_{l-i}^j$ appears in positions (l, l - (i + j)) for l = 1, ..., n.

On the other hand,

$$\phi((u_1^i, \dots, u_n^i)) \cdot \phi((u_1^j, \dots, u_n^j)) = \begin{pmatrix} \operatorname{Row}(u_1^i) \\ \vdots \\ \operatorname{Row}(u_n^i) \end{pmatrix} \cdot \begin{pmatrix} \operatorname{Row}(u_1^j) \\ \vdots \\ \operatorname{Row}(u_n^j) \end{pmatrix} =: \operatorname{M}_{\mathbf{i}} \cdot \operatorname{M}_{\mathbf{j}}.$$

The nonzero entries of M_i , namely u_l^i , appear in positions (l, l-i) for l = 1, ..., n, whereas the nonzero entries of M_j , the u_l^j , are in positions (l, l-j) = (l-i, l-(i+j)) for l, i = 1, ..., n. Therefore the product $M_i \cdot M_j$ has nonzero entries in positions (l, l-(i+j))and these entries are $u_l^i u_{l-i}^j$ for l = 1, ..., n. Thus we have a ring homomorphism ϕ between C and R which is clearly bijective. This concludes the proof of Claim 2.

Proof of Claim 3: We aim to show that $M \in \operatorname{Simp}_{<\infty}^o R$ has maximal k-vector space dimension, which is equal to n. Let s denote the diagonal matrix, $\operatorname{diag}(x^n)$ in R, and note that $sR = R_+$. Consider the ring $R[s^{-1}]$. For $M \in \operatorname{Simp}_{<\infty}^o R$, construct the module $0 \neq M[s^{-1}] = R[s^{-1}] \otimes_R M \in R[s^{-1}]$ -mod. Since M is simple, $M[s^{-1}]$ is also simple by [GW04, Corollary 10.16]. Furthermore $\dim_k M[s^{-1}] = \dim_k M$ as M is s-torsionfree [GW04, Exercise 10K]. Thus we have an inclusion of sets $\operatorname{Simp}_{<\infty}^o R \subseteq$ $\operatorname{Simp}_{<\infty} R[s^{-1}]$ given by $M \mapsto M[s^{-1}]$. Now observe that:

$$R[s^{-1}] = \begin{pmatrix} T' & x^{n-1}T' & xT' \\ xT' & T' & x^2T' \\ \vdots & \vdots & \ddots & \vdots \\ x^{n-1}T' & x^{n-2}T' & T' \end{pmatrix}$$

for $T' = k[(x^n)^{\pm 1}]$. So $R[s^{-1}] \subseteq \operatorname{Mat}_n(T')$ as rings, to say the rings have the same multiplicative structure. The rings are also equal as sets. Hence $R[s^{-1}] = \operatorname{Mat}_n(T')$ as rings. Since all simple modules of $\operatorname{Mat}_n(T')$ have dimension n, we have now verified Claim 3.

Thus we conclude the proof of the proposition.

Corollary IV.19. Given a three-dimensional Sklyanin algebra S(a, b, c) where $|\sigma_{abc}| < \infty$ with $S/Sg \cong B$, we have that $PIdeg(S) = PIdeg(B) = |\sigma|$.

Proof. In the proof of the preceding proposition, we have that $|\sigma|$ is equal to the maximal dimension of the simple finite-dimensional modules of the rings R and C. Since simple finite-dimensional modules of R and C correspond to modules in $\operatorname{Simp}^{o}_{<\infty}B$ with the same dimension by Claim 1, we have that $|\sigma| = \operatorname{PIdeg}(B)$ (Proposition II.52). Moreover $\operatorname{PIdeg}(S) = |\sigma|$ by Corollary IV.15.

Corollary IV.20. (We remind the reader that we have restricted our attention to E smooth.) The smooth locus and Azumaya locus of B coincide when $|\sigma| < \infty$.

Proof. By [ST94, Corollary 2.8], the center of B is equal to the twisted homogeneous coordinate ring associated to the triple $(E', \mathcal{L}|_{E'}, \sigma|_{E'})$ where $E' = \frac{E}{\langle \sigma \rangle}$, which is also a smooth elliptic curve. Hence max Z(B) is the affine cone over $E' \subseteq \mathbb{P}^2$, and so the smooth locus of max Z(B) is equal to max $Z(B) \smallsetminus \{0\}$. This set is precisely the Azumaya locus as proved in Proposition IV.18.

CHAPTER V

Representation theory of deformed Sklyanin algebras II: On the central extension D of $Skly_3$

Recall from Chapter 4 that our main objective is to classify irreducible finitedimensional representations of the deformed Sklyanin algebra S_{def} (Definition IV.2), a task which has implications in string theory (see §4.1). This problem boils down to studying such modules over the three-dimensional Sklyanin algebra $Skly_3$ (Definition IV.1), and over the central extension D of $Skly_3$. We are only able to give partial results towards this classification, and these are collected in this chapter. For instance, we are able to determine the 1-dimensional representations of D (Lemmas V.5 and V.6). For the reader's convenience, we restate the definition of D below.

Definition V.1. For i = 1, 2, 3, let a, b, c, d_i, e_i be scalars in k with $[a : b : c] \notin \mathfrak{D}$ (Definition IV.1). The **central extension** D of $Skly_3$ is generated by three variables x, y, z and a central element w, subject to the relations:

$$ayz + bzy + cx^{2} + d_{1}xw + e_{1}w^{2} = 0$$

$$azx + bxz + cy^{2} + d_{2}yw + e_{2}w^{2} = 0$$

$$axy + byx + cz^{2} + d_{3}zw + e_{3}w^{2} = 0$$

$$xw - wx = yw - wy = zw - wz = 0.$$

Notation. Let $\operatorname{Simp}_{<\infty} A$ denote the set of isomorphism classes of simple finitedimensional left A-modules, and let $\operatorname{Simp}_m A$ denote those of dimension m. Now the set $\operatorname{Simp}_{<\infty}Skly_3$ was investigated in Chapter 4 and hence this chapter is dedicated to the study of $\operatorname{Simp}_{<\infty}D$.

In the first section, we present results on the noncommutative geometry of D and we also describe the set $\operatorname{Simp}_1 D$ in terms of D-point modules. In certain cases (of physical significance), we provide the explicit descriptions of $\operatorname{Simp}_1 D$ and $\operatorname{Simp}_1 S_{def}$. The last two sections discuss partial results in determining the center of D and the fat point modules over D, with a view towards classifying higher dimensional irreducible representations of D. These techniques are motivated by the work of [LB95b, SS93] and their analysis of conformal \mathfrak{sl}_2 enveloping algebras and four-dimensional Sklyanin algebras respectively.

5.1 Background on the geometry of D, and computation of $Simp_1D$

Fortunately much is known about the noncommutative geometry of D, to say its point scheme and line scheme (Definition II.19) are known, and we present these results below. As a consequence, we can describe the 1-dimensional representations of D in terms of D-point modules.

To describe the geometric data of D, we remind the reader of the constructions of a point scheme with automorphism σ induced by the shift functor and of a line scheme (Definitions II.19, II.8). Recall in particular that the point scheme of $Skly_3$ is generically an elliptic curve $E \subseteq \mathbb{P}^2$ (Proposition II.15). In fact, we make the following assumption.

Assumption V.2. Let us restrict our attention to the case that the point scheme of $Skly_3$ is smooth, i.e. $E = \mathbb{P}^2$ or a smooth cubic curve. In the case that E is a smooth curve, we invoke Assumption IV.9 from Chapter 4.

Proposition V.3. [LBSVdB96, Theorem 4.1.11, Theorem 4.2.2(2), Proposition

4.3.7] Let D be a central extension of a generic three-dimensional Sklyanin algebra as in Definition V.1.

- Point modules M(p) of D corresponding to p ∈ P_D are the form of
 D/Dy₁ + Dy₂ + Dy₃ where y_i ∈ D₁. We also have that the point p ∈ P_D is
 V(y₁, y₂, y₃) ⊆ P³.
- 2. The point scheme of D, denoted P_D , is the union of E and a set of r points $S = \{s_i\}_{i=1}^r$. In fact, r = 8 for generic central extensions D of $Skly_3$ and this is the maximum cardinality of S. Here $E \subseteq \mathbb{V}(w) = \mathbb{P}^2 \subseteq \mathbb{P}^3_{[x:y:z:w]}$ and $S \subseteq \mathbb{P}^3 \setminus \mathbb{V}(w)$.
- The automorphism σ_D of P_D, induced by the shift functor on point modules of D, is given by σ on E and the identity on S.

Observe that by the last part of this result, we get that $|\sigma_D| = |\sigma|$.

Proposition V.4. [LBSVdB96, Example 5.2.5] The line scheme of D lies in $\mathbb{P}^3_{[x:y:z:w]}$ and is the union of the line scheme of Skly₃ and two lines passing through p for each point $p \in E \subseteq \mathbb{V}(w) \subseteq \mathbb{P}^2$.

We will use the line scheme later to understand fat point modules of D. On the other hand, since we understand point modules over D, we can now study the 1-dimensional representations of D and those of S_{def} .

Lemma V.5. Let D be a central extension of a three-dimensional Sklyanin algebra. Then the simple 1-dimensional D-modules arise as simple quotients of point modules over D.

Proof. By Lemma II.32, a simple 1-dimensional *D*-module *M* is the quotient of some 1-critical graded *D*-module *N*. Moreover by Remark II.33, $\operatorname{mult}(N) \leq \dim_k M = 1$, so $\operatorname{mult}(N) = 1$. Thus simple 1-dimensional *D*-modules are quotients of *D*-point modules.

Before we continue to study the sets $\operatorname{Simp}_1 D$ and $\operatorname{Simp}_1 S_{def}$, we point out specific algebras, D and S_{def} , that are of physical importance, namely relevant to $[\operatorname{BJL00}, \S4]$. The authors in this work study *q*-deformed theories or rather the set $\operatorname{Simp}_{<\infty} S_{def}(a, b, c, d_i, e_i)$ with the parameter c = 0. They also focus on the theories equipped with either: (i) a single mass term; (ii) a mass term and a linear term; (iii) three arbitrary linear terms; or (iv) three arbitrary mass terms. In our language, this is equivalent to classifying $\operatorname{Simp}_{<\infty} S_{def}$ with the respective conditions that:

- (i) d_1 is arbitrary, $d_2 = d_3 = e_i = 0$;
- (ii) d_1 , e_1 are arbitrary, $d_2 = d_3 = e_2 = e_3 = 0$;
- (iii) $d_i = 0$, or;
- (iv) $e_i = 0$.

Thus we describe the 1-dimensional representations of S_{def} for each of the cases: $d_i = 0$, $e_i = 0$, and $d_2 = d_3 = e_2 = e_3 = 0$.

Lemma V.6. For each of the cases $d_i = 0$, $e_i = 0$, and $d_2 = d_3 = e_2 = e_3 = 0$, the 1dimensional representations of $S_{def}(a, b, c, d_i, e_i)$ can be explicitly described. In fact, there is one such representation if $d_i = 0$, five representations if $e_i = 0$, and three representations if $d_2 = d_3 = e_2 = e_3 = 0$.

Now we compare these 1-dimensional representations of S_{def} with the following results on point modules of D.

Proposition V.7. Let D be a central extension of a generic Sklyanin algebra. Then the non-trivial simple 1-dimensional modules over S_{def} are simple quotients of the point modules

$$\{M(s_i) \mid s_i \in \mathcal{S}\} = \left\{\frac{D}{Dy_1 + Dy_2 + Dy_3} \mid \mathbb{V}(y_1, y_2, y_3) = s_i \in P_D\right\}.$$

Proof. See Routine A.4.

The set $S = \{s_i\}$ is explicitly described in the appendix for the cases: $d_i = 0$, $e_i = 0$, and $d_2 = d_3 = e_2 = e_3 = 0$. In fact for these cases, S consists of 1 point, 5 points, and 3 points respectively.

Proof. By Lemma IV.7, we know that $\operatorname{Simp}_1 S_{def} = \operatorname{Simp}_1 D \setminus \operatorname{Simp}_1 Skly_3$. Hence by Lemma IV.11, we have that $\operatorname{Simp}_1 S_{def} = \operatorname{Simp}_1 D \setminus \{D/D_+\}$.

For generic D, we know that $|\sigma_D| = |\sigma| = \infty$, and so the point modules M(p) for $p \in E$ only yield trivial simple quotients [LS93, Lemma 5.8(d)]. Thus the nontrivial simple 1-dimensional S_{def} -modules are simple quotients of the point modules $M(s_i)$ for $s_i \in S$.

Given the parameters $d_i = 0$, $e_i = 0$, or $d_2 = d_3 = e_2 = e_3 = 0$, refer to Routine A.5 for the computation of (the point scheme P_D , and particularly) the set S.

5.2 On the computation of $Simp_{>1}D$

Since we studied 1-dimensional representations of D (and of S_{def}) in the previous section, we now analyze simple D-modules of higher dimension. We are not able to provide classification results in general, but we proceed by listing partial results towards this goal. We begin with the following result.

Lemma V.8. If $M \in Simp_{<\infty}D$ arises as a simple quotient of a fat point module, then $\dim_k M > 1$.

Proof. Apply Remark II.33. \Box

In fact for generic D, we would like to establish the nonexistence of D-fat point modules.

On the other hand for certain D, namely some for which $|\sigma_D| = |\sigma| < \infty$, we aim to verify that D is module-finite over its center. In this case, D would have many higher

dimensional representations, particularly many of which have dimension equal to the PI degree of D. This is due to the Azumaya locus of D (Definition II.53) being dense in maxZ(D) (Propositions II.52, II.54). This leads us to the following claim.

Conjecture V.9. A central extension D of $Skly_3$ is either PI or all simple finitedimensional D-modules are 1-dimensional.

In §5.3, we discuss results on the center of D. We show that for generic D(for $|\sigma| = \infty$), the center of D is generated by the central element w and a homogeneous degree 3 element \hat{g}_3 (Proposition V.10). Moreover D is not PI in this case. We also investigate the center of D when $|\sigma| < \infty$, in which we conjectured that D is PI if $|\sigma| = 3, 6$ and not PI if $|\sigma| = 1, 2$ (Conjectures V.12, V.13, V.15, Remark V.16).

The final section, §5.4, is dedicated to the study of fat point modules of D for the reasons mentioned above. We aim to present fat point modules of D as quotients of line modules of D by shifted line modules of D, a strategy prompted by the methods in [LB95b] and [SS93]. We show that this presentation of D-fat point modules holds in the case that $|\sigma| = 2$ (Proposition V.29). We then aim to employ the geometry of line modules to draw conclusions about either the existence or structure of fat point modules. Partial results are reported below.

5.3 On the center of D

In this section we discuss the center of D and whether D satisfies a polynomial identity. Recall from the definition of D that D has one central element w of degree 1. For generic D (with $|\sigma_D| = |\sigma| = \infty$), we also have another central element \hat{g}_3 homogeneous of degree 3. This element is computed with the noncommutative algebra package Affine of the computer algebra system Maxima, and it is also presented in [EG, §9]. Here we assume (by rescaling) that the parameter a of D is equal to 1:

$$(5.1)$$

$$\hat{g}_{3} = c(c^{3} - b^{3})y^{3} + b(c^{3} - 1)yxz + (b^{3} - c^{3})xyz + c(1 - c^{3})x^{3}$$

$$+ (-c^{2}d_{1})wyz$$

$$+ [(-b^{4}d_{2} + b^{3}d_{2} - b^{2}d_{2} + 2bc^{3}d_{2} - c^{3}d_{2})/(b - 1)]wy^{2}$$

$$+ [(-b^{3}d_{3} + bc^{3}d_{3})/(-c(b - 1))]wyx$$

$$+ (bc^{2}d_{2})wxz$$

$$+ [(-b^{2}d_{3} + bc^{3}d_{3})/(-c(b - 1))]wxy$$

$$+ [(-b^{2}d_{1} - bc^{3}d_{1} + bd_{1} + 2c^{3}d_{1} - d_{1})/(b - 1)]wx^{2}$$

$$+ [(-b^{3}ce_{3} + 2b^{2}ce_{3} - b^{2}d_{3}^{2} + bc^{2}d_{1}d_{2} - bce_{3}/(-c(b - 1)))]w^{2}z$$

$$+ [(-b^{4}e_{2} + 2b^{3}e_{2} - 2b^{2}e_{2} + bc^{3}e_{2} + bc^{2}d_{2}^{2} - bcd_{1}d_{3} + be_{2} - c^{3}e_{2})/(b - 1)]w^{2}y$$

$$+ [(b^{3}e_{1} + 2b^{2}e_{1} - bc^{3}e_{1} - bcd_{2}d_{3} + 2be_{1} + c^{3}e_{1} + c^{2}d_{1}^{2} - e_{1})/(b - 1)]w^{2}x.$$

Considering the surjection $D \twoheadrightarrow D/Dw \cong Skly_3$, we have that the image of \hat{g}_3 is the element g from Remark II.17. With this observation, we now show that \hat{g}_3 and w generate the center of D in the case where $|\sigma| = \infty$.

Proposition V.10. Given a central extension D of $Skly_3$ for which $|\sigma| = \infty$, we have that $Z(D) = k[w, \hat{g}_3]$. In this case, D is not PI.

Proof. Let C be the graded subalgebra of Z(D) generated by \hat{g}_3 . Consider the surjection $\pi: D \twoheadrightarrow D/Dw = Skly_3$. Since $\pi(C) = k[g]$, and Z(S) = k[g] for $|\sigma| = \infty$ [Smi94, §13], we have by [ST94, Lemma 3.6] that $Z(D) = k[\hat{g}_3, w]$.

Furthermore S is not PI as $|\sigma| = \infty$ (Theorem I.18). So by considering the homomorphism π above, we have that D is also not PI (Proposition II.38).

Hence according to Conjecture V.9, we claim that $\operatorname{Simp}_{<\infty} D = \operatorname{Simp}_1 D$ for $|\sigma| = \infty$. This problem is addressed in the next section. Now we study the center of D when $|\sigma_D| = |\sigma| < \infty$. The task of finding such D is equivalent to classifying parameters (a, b, c) for which $\sigma = \sigma_{abc}$ has finite order. Here, by rescaling parameters, we assume that a = 1. Partial results on this problem are reported in Proposition A.1 in the appendix. In fact for $n = 1, \ldots, 6$, we know the parameters (a, b, c) for which $\sigma = \sigma_{abc}$ has order n. Hence we study the central extensions D of S(a, b, c) where $|\sigma_{abc}| = 1, \ldots, 6$.

Case: $|\sigma| = 1$

First, we have that $|\sigma| = 1$ if and only if $Skly_3$ is the commutative polynomial ring S(1, -1, 0) = k[x, y, z]. Contrary to the behavior of k[x, y, z], for (generic) parameters d_i , e_i , we have the following computations for the center of a central extension of k[x, y, z].

Lemma V.11. 1. Let D be a generic central extension of k[x, y, z]. Then for $1 \le n \le 3$, the k-vector spaces $Z(D)_n$ are generated by w and

$$q = d_1 x^2 + d_2 y^2 + d_3 z^2 + 2e_1 w x + 2e_2 w y + 2e_3 w z.$$

Let D be the central extension of a k[x, y, z] given in the table below. Then for 1 ≤ n ≤ 6, the k-vector spaces Z(D)_n are generated by w and the element q given as follows:

D	q
$D(1, -1, 0, 0, 0, 0, e_1, e_2, e_3)$	$e_1x + e_2y + e_3z$
$D(1, -1, 0, d_1, d_2, d_3, 0, 0, 0)$	$d_1x^2 + d_2y^2 + d_3z^2$
$D(1, -1, 0, d_1, 0, 0, e_1, 0, 0)$	x

This leads us to the following claim.

Conjecture V.12. The center of $D(1, -1, 0, d_i, e_i)$ (respectively of D with $d_i = 0$, with $e_i = 0$, and with $d_2 = d_3 = e_2 = e_3 = 0$) is generated by the elements w and q given in Lemma V.11. Thus D is not PI.

If this conjecture holds, then we see that a result such as Theorem I.18 clearly does not apply. In other words, the structure of Z(D) does not necessarily depend on the order of σ_D (= $|\sigma|$). However we will see that this relationship does hold for other cases, such as $|\sigma| = 3$.

We proceed to study the center of D with σ of higher order. Since the algebra program Affine does not accommodate for complex parameters and parameters with fractional powers, according to Proposition A.1 we restrict our attention to $|\sigma| =$ 2,3,6.

Case: $|\sigma| = 2$

Here we have that $|\sigma| = 2$ if and only if the parameters (a, b, c) equal (1, 1, c) with $c \neq 0$ and $c^3 \neq 1$ by Proposition A.1. Moreover by Affine, we have a central element \hat{g}_2 of D of degree 2:

(5.2)
$$\hat{g}_{2} = d_{2}(c^{4} - c)y^{2} + d_{3}(-c^{3} + 1)yx + d_{3}(-c^{3} + 1)xy + d_{1}(c^{4} - c)x^{2} + (d_{3}^{2} - c^{2}d_{1}d_{2}) \cdot wz + (-c^{2}d_{1}d_{3} + c^{3}d_{2}^{2}) \cdot wy + (-c^{2}d_{2}d_{3} + c^{3}d_{1}^{2}) \cdot wx$$

Now for various parameters c, d_i , e_i , we have computed the center of D up to degree 12. For these examples, the k-vector spaces $Z(D)_n$ for $1 \le n \le 12$, are generated by w and \hat{g}_2 . Hence we make the following claim.

Conjecture V.13. For generic parameters d_i and e_i , the center of the central extension $D(1, 1, c, d_i, e_i)$ is $k[\hat{g}_2, w]$. Moreover D is not PI.

Case: $|\sigma| = 3$

By Proposition A.1, we know that $|\sigma_{abc}| = 3$ if and only if:

$$[a:b:c] = \begin{cases} [1:0:\omega], & for \ \omega = -1, e^{\pi i/3}, e^{5\pi i/3}; \\ [1:\omega:0], & for \ \omega = e^{\pi i/3}, e^{5\pi i/3}. \end{cases}$$

For computational reasons mentioned above, we will focus on the case where (a, b, c) = (1, 0, -1).

Lemma V.14. The central extension $D(1, 0, -1, d_1, d_2, d_3, e_1, e_2, e_3)$ is module-finite over its center, and thus is PI. Furthermore Z(D) is generated by four elements, $\{\rho_i\}_{i=1}^4$, of degree 3, and one element w of degree 1.

Proof. Note that $S(1,0,-1) \cong D(1,0,-1,d_1,d_2,d_3,e_1,e_2,e_3)/(w)$ is module-finite over its center as $|\sigma_{1,0,-1}| = 3$ (Theorem I.18). Following [ST94, Lemma 3.6], we show that there exists a graded subalgebra C of Z(D) so that the image of C in S = S(1,0,-1)is Z(S). By [ST94, Theorem 3.7], Z(S) is generated by four elements of degree 3. In fact by Affine, we know that these generators are:

$$\begin{array}{ll} \rho_{1} &= zyx + yxz + xzy, \\ \rho_{2} &= y^{2}x + xy^{2} + x^{2}z, \\ \rho_{3} &= yx^{2} + xyx + x^{2}y, \\ \rho_{4} &= x^{3}. \end{array}$$

Furthermore, we have by Affine, four degree 3 central elements of D:

$$\begin{aligned} \hat{\rho_1} &= \rho_1 - d_1(wzy) - d_2(wy^2) - d_3(wyx) - d_2(wxz) - d_3(wxy) - d_1(wx^2) \\ &+ (e_3 + d_1d_2)(w^2z) + (e_2 + d_1d_3 + d_2^2)(w^2y) + (e_1 + d_2d_3 + d_1^2)(w^2x), \\ \hat{\rho_2} &= \rho_2 - d_1(wy^2) - d_2(wyx) - d_1(wxz) - d_2(wxy) - d_3(wx^2) - e_1(w^2z) \\ &+ d_1d_2(w^2y) + (-e_2 + d_1d_3)(w^2x), \\ \hat{\rho_3} &= \rho_3 - d_1(wyx) - d_1(wxy) - d_2(wx^2) - e_1(w^2y) + (e_3 + d_1d_2)(w^2x), \\ \hat{\rho_4} &= \rho_4 - d_1(wx^2) - e_1(w^2x). \end{aligned}$$

Let *C* be the graded subalgebra of Z(D) generated by the $\hat{\rho}_i$. Since the image of $\hat{\rho}_i$ in *S* is ρ_i , we have by [ST94, Lemma 3.6] that both $Z(D) = k[\hat{\rho}_1, \hat{\rho}_2, \hat{\rho}_3, \hat{\rho}_4, w]$ and *D* is module-finite over this center.

Thus Conjecture V.9 holds for a central extension D with $|\sigma| = 3$. For the other four central extensions D where $|\sigma| = 3$, we believe that Z(D) is generated by w, and elements $\{\hat{\rho}_{i,\omega}\}_{i=1}^{4}$ which are the elements $\hat{\rho}_{i}$ with coefficients involving ω and ω^{2} . This prompts the following claim.

Conjecture V.15. Let D be a central extension of Skly₃ for which $|\sigma|=3$. Then the center of D is generated by w and four elements of degree 3, and D is module-finite over its center. In other words, Conjecture V.9 holds for $|\sigma|=3$.

Cases: $|\sigma| = 4, 5$

If the automorphism σ_{abc} has order 4 or 5, then one of the parameters, b or c, is a complex number or involves a fractional power (Proposition A.1). For the computational reasons mentioned above, we will omit these cases for now.

Case: $|\sigma| = 6$

By Proposition A.1, we know that $|\sigma_{abc}| = 6$ if and only if:

$$[a:b:c] = \begin{cases} [1:b:\zeta], [1:b:b\zeta], \text{ for } b \neq 0, b^3 \neq 1, \\ [1:\zeta:c], [1:c\zeta:c], \text{ for } c \neq 0, c^3 \neq 1, \zeta \neq 1 \end{cases}$$

where ζ is a third root of 1. For computational purposes, we will only study the cases where $\zeta = 1$. Our approach to study Z(D) is given in the following remark.

Remark V.16. We believe that we can employ the approach of the $|\sigma| = 3$ case. Namely, compute the center of D in degree 3 and 6, then show that the images of these elements under the surjection $D \twoheadrightarrow Skly_3$ are generators of $Z(Skly_3)$. However due to memory constraints, the computations with arbitrary parameters d_i , e_i cannot be executed at this time.

5.4 On the fat point modules of D

In the spirit of [Art92], [LB95b], and [SS93], we study fat point modules over D by presenting them as a quotient of a line module by a shifted line module. Although for generic D, we will see that we have to take other "curve modules" into consideration.

We restrict our attention to the cases where $|\sigma| = \infty$ (to say $D = D(1, b, c, d_i, e_i)$) and $|\sigma| = 2$ (to say $D = D(1, 1, c, d_i, e_i)$) as we have shown/ conjectured that such Dare not PI (Proposition V.10, Conjecture V.13). (Here we assume by rescaling that the parameter a equals 1.) In fact, we have respective central subalgebras $k[\hat{g}_3, w]$ and $k[\hat{g}_2, w]$ for these central extensions, where $\deg(\hat{g}_d) = d$. We will use this data towards our aim to verify the nonexistence of fat point modules of D with $|\sigma| = 2, \infty$, or rather towards our aim to prove Conjecture V.9 for these cases.

Now given a fat point module F of D, our strategy to study F breaks down into four tasks:

- **I.** Compute line modules M_l of D.
- II. Compute central annihilators Ω_l for the line modules M_l .

In fact, we want our fat point module F to a share a central annihilator with some line module M_l . By [LB95b, proof of Proposition 10], the intersection of $\operatorname{ann}_D F$ and the graded central subalgebra $k[\hat{g}_d, w^d]$ is nonzero. Hence for every point $[\alpha : \beta] \in \mathbb{P}^1$, we require a line module $M_{\alpha\beta}$ with corresponding central annihilator $\Omega_{\alpha\beta} = \alpha w^d + \beta \hat{g}_d$.

Continuing with our strategy, we have the final two tasks:

III. Present F (say of multiplicity $\epsilon > 1$) by line modules. In other words, verify the existence of an exact sequence in D-qgr:

$$0 \to M_{l'}[-\epsilon] \to M_l \to F \to 0,$$

for line modules M_l and $M_{l'}$ of D.

IV. Use the geometry of the line scheme of D (Proposition V.4) to yield results about the structure/existence of the fat point module F.

Remark V.17. Karen Smith suggested that presenting fat point modules of D with plane modules (rather than line modules) may be useful. In particular, does there exists a sequence

$$0 \longrightarrow \bigoplus_{j=1}^{n-1} M_{H''_j} \longrightarrow \bigoplus_{i=1}^n M_{H'_i} \longrightarrow M_H \longrightarrow F \longrightarrow 0$$

for plane modules M_H , M_{H_i} , M_{H_j} of D? For now, we leave it to the reader to explore such an approach.

Case: $|\sigma| = \infty$

I. We begin by constructing a family of line modules for $D = D(1, b, c, d_i, e_i)$.

Lemma V.18. Given a central extension $D = D(1, b, c, d_i, e_i)$, we have a 5 parameter family of line modules of D:

$$\left\{M_{l} = \frac{D}{D(x+u_{2}y+u_{3}z+u_{4}w)+D(v_{1}x+u_{2}v_{1}y+u_{3}v_{1}z+v_{4}w)}\right\},\$$

where $u_2, u_3, u_4, v_1, v_4 \in k$ and $(v_1, v_4) \neq (0, 0)$.

Proof. By [LBSVdB96, Proposition 4.1], there is a bijective correspondence between line modules of D:

$$\{D/(Du+Dv) \mid u, v \in D_1\},\$$

and rank two tensors of D:

$$\{r \otimes u - sv \in R_D \mid r, s, u, v \in D_1\}$$

where R_D is the ideal of relations of D. With Affine, we have constructed a 5 parameter family of rank two tensors given by:

$$r = (b_1v_1) \cdot x + (b_1u_2v_1) \cdot y + (b_1u_3v_1) \cdot z + (b_1v_4) \cdot w,$$

$$s = b_1 \cdot x + (b_1u_2) \cdot y + (b_1u_3) \cdot z + (b_1u_4) \cdot w,$$

$$u = x + u_2 \cdot y + u_3 \cdot z + u_4 \cdot w,$$

$$v = v_1 \cdot x + (u_2v_1) \cdot y + (u_3v_1) \cdot z + v_4 \cdot w,$$

with $u_i, b_i, v_i \in k$ for i = 1, 2, 3, 4. The verification computation is Routine A.6 in the appendix.

II. Next, we need to compute the central annihilators Ω_l for the line modules $M_l = D/Du + Dv$ given in the lemma above, aiming to get a \mathbb{P}^1 's worth of pairs (M_l, Ω_l) with $\Omega_l \cdot M_l = 0$. Note that $\dim_k D_2 = 10$. Thus for i = 1, 2, we require scalars $\{a_{i,j}\}$ so that

$$f_i = a_{i,1}x^2 + a_{i,2}xy + a_{i,3}xz + \dots + a_{i,10}w^2 \in D_2 \text{ with } f_1u + f_2v = \alpha w^3 + \beta \hat{g}_3$$

for some $[\alpha : \beta] \in \mathbb{P}^1$. This computation is unattainable at the moment due to memory restrictions, yet we proceed to step III assuming that this task has been achieved; see Assumption V.20 below.

III. Along with line modules, we also consider the following geometric modules.

Definition V.19. A plane degree d curve module of D is a cyclic graded left D-module M with Hilbert series:

$$H_M(t) = \frac{1+t+\dots+t^{d-1}}{(1-t)^2}.$$

Assumption V.20. Recall that for $|\sigma| = \infty$, we aim to prove that fat point modules over D do not exist via the use of the presentation of D-line modules. Moreover recall that if fat point modules of D, then we get that $\operatorname{ann}_D F \cap k[w^3, \hat{g}_3] \neq 0$. Hence there will be a central annihilator $\Omega_{\alpha\beta} = \alpha w^3 + \beta \hat{g}_3$ of F, and we need $\Omega_{\alpha\beta} \cdot M_l = 0$ for some D-line module M_l .

- 1. We assume that we have a pair $(M_{\alpha\beta}, \Omega_{\alpha\beta})$ of a line module with corresponding central annihilator for every point $[\alpha : \beta] \in \mathbb{P}^1$.
- 2. We assume that $\Omega_{\alpha\beta} \neq \Omega_{10} = w^3$. Otherwise *F* is a fat point module over $Skly_3$ which does not exist as $|\sigma| = \infty$ [ATVdB91, Theorem 7.5, Corollary 7.9]. Hence we are done in this case, i.e. we have verified Conjecture V.9.

As the central extension D is quadratic Auslander regular of global dimension 4 with Hilbert series $(1 - t)^{-4}$ [LBSVdB96, Corollary 2.7], we employ the following result of LeBruyn relevant to D.

Proposition V.21. [LB95b, Proposition 10] For a central extension D of $Skly_3$, assume that the following conditions hold.

- (H1) D has two central elements w^d and \hat{g}_d of degree d.
- (H2) For every $\Omega = \alpha w^d + \beta \hat{g_d}$ with $[\alpha : \beta] \in \mathbb{P}^1$, there is a line module $M_{\Omega} = D/DU$ with U a 2-dimensional subspace of D_1 such that:
 - $(H2a)\ \Omega\cdot M_{\Omega}=0;$
 - (H2b) For every $u \in U$, $\Omega \cdot (D/Du) \neq 0$;
 - (H2c) M_{Ω} has a point module as a quotient.

Then every fat point module of D is either a quotient of a line module or of a plane degree d-1 curve module of D.

We proceed by showing that the hypothesis of this proposition hold for D with $|\sigma| = \infty$.

Lemma V.22. The hypotheses of Proposition V.21 hold for D with $|\sigma| = \infty$.

Proof. Now (H1) holds by taking d = 3 (Equation (5.1)). Considering the family of line modules constructed in Lemma V.18, Assumption V.20 implies that (H2a) holds. By [LBSVdB96, Proposition 5.1.1(2)], we get that line modules M = D/Du + Dv with $u, v \in D_1$ correspond to lines $l = \mathbb{V}(u, v) \subseteq \mathbb{P}^3$ in the line scheme of D. Recalling the structure of the line scheme (Proposition V.4), we know that M/Mw (corresponds to the point $\mathbb{V}(u, v, w) \in E$ and) is a point module of $Skly_3$, and thus of D. Hence condition (H2c) holds.

Finally we have that condition (H2b) holds as follows. Suppose not, i.e. that there exists $u \in U$ with $\Omega \cdot (D/Du) = 0$. Then $\Omega = du$ for some $d \in D_2$. In the domain, $B \cong D/(Dw + Dg)$, we have that $\overline{du} = \overline{\Omega} = 0$. So either \overline{d} or \overline{u} equals 0. In the first case d = aw for some $a \in D_1$, and in the second case u = bw for some $b \in k$. Both cases imply that w divides Ω , which contradicts Assumption V.20(2).

Therefore given a fat point module F of multiplicity $\epsilon > 1$, we have one of the presentations of F in D-qgr:

$$0 \to K \to M_l \to F \to 0$$
 or $0 \to K \to N \to F \to 0$,

where M_l is a line module of D and N is a plane curve degree 2 module of D. Since we do not have results on the kernel of $N \twoheadrightarrow F$ at this time, we make the following assumption.

Assumption V.23. Given a fat point module F of D, we assume that F is the quotient of a line module M_l of D.

Now we study the kernel of $M_l \twoheadrightarrow F$. Before we do so, consider the following terminology.

Definition-Lemma V.24. [LBSVdB96, Corollary 2.7] Given a central extension D of Skly₃, let M be a finitely generated D-module.

1. The grade of M is the quantity $j(M) \in \mathbb{N} \cup \{+\infty\}$ defined by

$$j(M) \coloneqq \inf\{i \mid Ext_D^i(M, D) \neq 0\}.$$

2. D is a Cohen-Macaulay (CM) ring in the sense that

$$GKdim(M) + j(M) = 4 \ (= gldim(D)).$$

 The module M is called a Cohen Macaulay (CM) module if pd(M)=j(M), to say Extⁱ_D(M,D) = 0 if i ≠ j(M).

Proposition V.25. Let F be a fat point module of D of multiplicity $\epsilon > 1$, which is the quotient of a line module M_l . Then we have that the kernel of $M_l \twoheadrightarrow F$ is isomorphic to a shifted line module $M_{l'}[-\epsilon]$.

Proof. We assume that

$$(5.3) 0 \longrightarrow K \longrightarrow M_l \longrightarrow F \longrightarrow 0,$$

exact in *D*-qgr. Without loss of generality, we can assume that this sequence is exact in *D*-gr as *F* is equivalent to a submodule for which this condition holds. We now have to show that *K* is a shifted line module. Apply $\text{Hom}_D(-, D)$ to sequence (5.3) and take cohomology. Let $E^i(P) := \text{Ext}_D^i(P, D)$ for a *D*-module *P*. We get the following long exact sequence:

$$0 \longrightarrow \operatorname{Hom}_{D}(F, D) \longrightarrow \operatorname{Hom}_{D}(M_{l}, D) \longrightarrow \operatorname{Hom}_{D}(K, D)$$
$$E^{1}(F) \xrightarrow{\checkmark} E^{1}(M_{l}) \longrightarrow E^{1}(K) \dots$$

Since D is a Cohen Macaulay ring, we have that:

$$j(F) = GKdim(D) - GKdim(F) = 4 - 1 = 3.$$

Moreover line modules have GK-dimension 2, so we have that:

$$j(M_l) = GKdim(D) - GKdim(M_l) = 4 - 2 = 2$$

Since $2 = j(M_l) = \inf\{j(K), j(F) = 3\}$ [LS93, remarks on p.43], we get that j(K) = 2. Now $E^i(M_l) = E^{i+1}(F) = 0$ for $i \neq 2$; hence $E^i(K) = 0$ for $i \neq 2 = j(K)$. Thus K is CM of GK-dimension 2 (which equals GKdimD - j(K)). Moreover we get that $\operatorname{mult}(K) = 1$ as $\operatorname{mult}(M_l) = 1$.

It now remains to verify the following statement.

<u>Sublemma</u>. If K is a finitely generated graded D-module, CM, of GK-dimension 2 and multiplicity 1, then K is a shifted line module.

Proof of Sublemma. This proof is based on techniques of [LS93, §2]. We first shift the grading of K to assume that

$$K = K_0 \oplus K_1 \oplus K_2 \oplus \ldots$$

with $K_0 \neq 0$. Moreover K is 2-critical due to [LS93, Lemma 1.11].

If w is a zero divisor of K, then we get that wK = 0 by [LS93, Lemma 2.10]. Thus K is a module over $Skly_3$ and is isomorphic to a shifted line module by [ATVdB91, Proposition 6.2].

If w is a nonzero divisor of K, then form $\overline{K} = K/wK$. Since $H_{\overline{K}}(t) = (1-t)H_K(t)$, we have that $\operatorname{GKdim}(\overline{K})=1$ and $\operatorname{mult}(\overline{K})=1$. Considering the sequence

$$0 \longrightarrow K \xrightarrow{\cdot w} K[1] \longrightarrow \overline{K}[1] \longrightarrow 0,$$

we have that the first two terms are CM, thus \overline{K} is also CM [LS93, Lemma 1.12]. Hence by [LS93, Proposition 2.6],

$$H_{\overline{K}}(t) = t^p (1-t)^{-1} = t^p + t^{p+1} + t^{p+2} + \dots$$

for some $p \in \mathbb{Z}$. For $i \leq 0$:

$$\dim \overline{K}_i = \dim K_i - \dim(wK)_i$$
$$= \dim K_i - \dim K_{i-1} = \dim K_i$$

so p = 0. Thus $H_{\overline{K}}(t) = (1 - t)^{-1}$ and so $H_K(t) = (1 - t)^{-2}$. Now we must show that K is cyclic.

Let $K^{\vee} := E^2(K)$, the dual of K. Since $K \cong (K^{\vee})^{\vee}$, it suffices to prove that K^{\vee} is cyclic. By [LS93, Remark 1.12], if $P_{\bullet} \xrightarrow{\partial} K \to 0$ is a minimal resolution of K, then $P_{\bullet}^{\vee} \xrightarrow{\partial^{\vee}} K^{\vee} \to 0$ is a minimal resolution of K^{\vee} . More precisely, consider the following minimal resolution of K:

$$0 \longrightarrow \bigoplus_{i} D[-i]^{p_i} \xrightarrow{\partial_1} \bigoplus_{i} D[-i]^{m_i} \xrightarrow{\partial_0} \bigoplus_{i} D[-i]^{n_i} \longrightarrow K \longrightarrow 0$$

where n_i is the number of generators of K of degree i. Then, the sequence below is a minimal resolution of K^{\vee} :

$$0 \longrightarrow \bigoplus_{i} D[i]^{n_{i}} \xrightarrow{\partial_{0}^{\vee}} \bigoplus_{i} D[i]^{m_{i}} \xrightarrow{\partial_{1}^{\vee}} \bigoplus_{i} D[i]^{p_{i}} \longrightarrow K \longrightarrow 0$$

where p_i is the number of generators of K^{\vee} of degree -i.

Since $K = \bigoplus_{i \ge 0} K_i$, we have that $n_i = 0$ for $i \le -1$. Therefore $m_i = 0$ for $i \le 0$ and $p_i = 0$ for $i \le 1$. Now by [LS93, Proposition 1.10], the Hilbert series of K^{\vee} is

$$H_{K^{\vee}}(t) = t^{-4}H_K(t^{-1}) = t^{-2} + 2t^{-1} + 3 + 4t + 5t^2 + \dots$$

Thus $p_2 = 1$ and $p_i = 0$ for $i \ge 3$. Hence K^{\vee} is cyclic with a generator in degree -2, and so K is cyclic as desired. This concludes the proof of the sublemma.

Thus K is a shifted line module over D with $H_K(t) = t^d(1-t)^{-2}$. This implies that

$$H_F(t) = H_{M_l}(t) - H_K(t) = (1 + t + \dots + t^{d-1})(1 - t)^{-1}$$

Hence $\epsilon = \operatorname{mult}(F) = d$ and $K \cong M_{l'}[-\epsilon]$ for some line module $M_{l'}$ over D.

IV. Although we do not have a conclusion about the existence/ structure of fat point modules for D with $|\sigma| = \infty$, we do have the following geometric result involving the line modules of D, which is an adaptation of [ATVdB91, Propositions 6.23, 6.24].

Proposition V.26. Let $D = D(1, b, c, d_i, e_i)$ be a central extension of $Skly_3$. For a fat point module F of multiplicity $\epsilon > 1$, consider the exact sequence

$$0 \to M_{l'}[-\epsilon] \to M_l \to F \to 0$$

resulting from Proposition V.25, where M_l and $M_{l'}$ are line modules over D. Then we have that $l' \cap E = \sigma^{-\epsilon}(l \cap E)$.

Proof. We use techniques from [ATVdB91, §6]. Say $M_l = D/(Du + Dv)$ for some $u, v \in D_1$ and so $l = \mathbb{V}(u, v) \subseteq \mathbb{P}^3$ [LBSVdB96, Proposition 5.1.1]. Let P be the scheme theoretic intersection of l and E which is nonempty due to [LBSVdB96, page 204]. Let $M(P) = (\Gamma_*(\mathcal{O}_P))_{\geq 0}$ where $\Gamma_n(\mathcal{O}_P) = H^0(E, \mathcal{O}_P \otimes \mathcal{L}_n)$. This is a B-module. Taking Γ_* of $\mathcal{O}_E \to \mathcal{O}_P$ yields $B \to M(P)$, surjective in high degree. Now we get the map ϕ as in the following diagram:



with the horizontal sequence exact in *D*-qgr. Thus M(P) is equivalent to D/K(P)in *D*-qgr. Since *P* is a 0-dimensional subscheme of *E*, let *P*=Spec *R* for some finite
k-algebra R. Hence

$$R \cong H^0(E, \mathcal{O}_{\mathrm{Spec}R}) = H^0(E, \mathcal{O}_P \otimes \mathcal{L}_0) = M(P)_0.$$

This implies that $\phi(1_D) = 1_R$. Now

$$P \subseteq l = \mathbb{V}(u, v) \iff u \cdot 1_R = v \cdot 1_R = 0 \text{ in } M(P) \iff u, v \in K(P).$$

Hence $\phi: D \to M(P)$ factors through the *D*-gr surjection $D \twoheadrightarrow M_l$ to get the map ψ below:



As ϕ is surjective in *D*-qgr, the cokernel of ϕ is finite dimensional. Thus coker ψ is also finite dimensional. Consider the following diagram.

$$0 \longrightarrow M_{l'}[-\epsilon] \longrightarrow M_l \longrightarrow F \longrightarrow 0$$

$$\downarrow^{\psi} M(P)$$

Now $\ker \psi \notin M_{\ell'}[-\epsilon]$; else we get the induced map $F \to M(P)$ which implies that $F/(\ker(F \to M(P)))$, equivalent to a submodule of M(P). So the fat point module F would be a B-module, a contradiction. Note that $M_l/(\ker \psi + M_{\ell'}[-\epsilon])$ is finite dimensional due to F being 1-critical. Consider the map $\psi|_{M_{\ell'}[-\epsilon]}$ and we get that:

$$\operatorname{coker}(\psi|_{M_{l'}[-\epsilon]}) \cong \frac{M(P)}{\operatorname{im}(\psi|_{M_{l'}[-\epsilon]})}$$
$$\cong \frac{M(P)}{M_{l'}[-\epsilon]/(\operatorname{ker}\psi \cap M_{l'}[-\epsilon])}$$
$$\cong \left(\frac{M_l}{\operatorname{ker}\psi}\right) / \left(\frac{\operatorname{ker}\psi + M_{l'}[-\epsilon]}{\operatorname{ker}\psi}\right),$$

and the latter is finite dimensional. Therefore $\psi|_{M_{l'}[-\epsilon]}: M_{l'}[-\epsilon] \to M(P)$ has finite cokernel. By shifting we get that $M_{l'} \to M(P)[\epsilon] \cong M(\sigma^{-\epsilon}P)$ and this map has a finite cokernel. Thus l' contains $\sigma^{-\epsilon}P$ and $\sigma^{-\epsilon}P = (\sigma^{-\epsilon}P) \cap E \subseteq l' \cap E$, which implies that scheme-theoretically: $l' \cap E = \sigma^{-\epsilon}P$.

Case: $|\sigma| = 2$

Now we analyze fat point modules of $D(1, 1, c, d_i, e_i)$ via the steps I - IV stated at the beginning of the section. Most results will follow in a similar fashion to the $|\sigma| = \infty$ case. However a conclusion about the existence/ structure of fat point modules for these central extensions D remains unknown.

I. We begin by constructing a family of line modules.

Lemma V.27. Given a central extension $D(1, 1, c, d_i, e_i)$, we have a 5 parameter family of line modules of D:

$$\left\{M_{l} = \frac{D}{D(x+u_{2}y+u_{3}z+u_{4}w)+D(v_{1}x+u_{2}v_{1}y+u_{3}v_{1}z+v_{4}w)}\right\},\$$

where $u_2, u_3, u_4, v_1, v_4 \in k$.

Proof. Referring to the proof of Lemma V.18, note that the line modules do not depend on the parameter *b*. Hence we obtain the same family of line modules $\{M_{u_2u_3u_4v_1v_4}\}$ for $D(1, 1, c, d_1, d_2, d_3, e_1, e_2, e_3)$.

II. For D with $|\sigma| = 2$, Assumption V.20 is not needed as we have the following result.

Lemma V.28. For $D = D(1, 1, c, d_i, e_i)$, the line modules $\{M_{u_2u_3u_4v_1v_4}\}$ have central annihilators $\Omega_l = \hat{g}_2 + \gamma w^2$ where γ depends on scalars u_4 , v_1 , v_4 . Thus we get a pair $(M_{\alpha\beta}, \Omega_{\alpha\beta})$ of a line module and corresponding central annihilator for each $[\alpha : \beta] \in \mathbb{P}^1$. Proof. If $[\alpha : \beta] = [1:0]$, then $\Omega_{\alpha\beta} = w^3$. Moreover we have line modules M over $Skly_3$, which exist by Example II.20, so that $w^3 \cdot M = 0$. On the other hand, if $[\alpha : \beta] \neq [1:0]$, then $[\alpha : \beta] = [\gamma : 1]$ for some $\gamma \in k$ and $\Omega_{\alpha\beta} =: \Omega_{\gamma} = \gamma w^2 + \hat{g}_2$. See Routine A.7 in the appendix for the proof that for all $\gamma \in k$, we have that Ω_{γ} annihilates some line module $M_{u_2u_3u_4v_1v_4}$.

III. Now we have that fat point modules of $D = D(1, 1, c, d_i, e_i)$ are presented by line modules of D by the following argument.

Proposition V.29. Let $|\sigma| = 2$, so we consider the central extension

 $D = D(1, 1, c, d_i, e_i)$ over S(1, 1, c). Take F a fat point module of D of multiplicity $\epsilon > 1$. Then there exist line modules M_l and $M_{l'}$ for which $0 \to M_{l'}[-\epsilon] \to M_l \to F \to 0$ is exact in D-qgr.

Proof. First note that we do not require Assumption V.23 as Proposition V.21 holds for d = 2. In other words, we always get that F is the quotient of a line module (as a plane curve degree 1 module is a line module by definition). The rest of the proof follows from the proof for Proposition V.25.

IV. Moreover we have the same geometric result (Proposition V.26) from the $|\sigma| = \infty$ case, and its proof follows in the same manner.

Proposition V.30. Let $D = D(1, 1, c, d_i, e_i)$ be a central extension of S(1, 1, c). For a fat point module F of D of multiplicity $\epsilon > 1$, consider the exact sequence $0 \rightarrow M_{l'}[-\epsilon] \rightarrow M_l \rightarrow F \rightarrow 0$ resulting from Proposition V.29, where M_l and $M_{l'}$ are line modules over D. Then we have that $l' \cap E = \sigma^{-\epsilon}(l \cap E)$.

Thus for $|\sigma| = 2$, we understand *D*-fat point modules in terms of *D*-line modules and have some results about the geometry of such modules.

APPENDIX

APPENDIX A

Computational results, Affine and Maple routines

This appendix contains computational results and computer routines pertaining to the order of σ for the three-dimensional Sklyanin algebras $Skly_3 = S(a, b, c)$ (Definition I.2), to the 1-dimensional representations of deformed Sklyanin algebras S_{def} (Definition I.12), and to the line modules over central extensions D of $Skly_3$ (Definition IV.6).

Note: We assume that the base field k is $mathbb{C}$ for these computations.

The automorphisms σ of finite order

Recall from §2.1.3 that the three-dimensional Sklyanin algebras S = S(a, b, c) come equipped with geometric data (E, \mathcal{L}, σ) where $E = E_{abc} \subseteq \mathbb{P}^2$ is given by:

$$E = E_{abc} : \mathbb{V}\left((a^3 + b^3 + c^3) xyz - (abc)(x^3 + y^3 + z^3) \right) \stackrel{i}{\subset} \mathbb{P}^2,$$

and \mathcal{L} is the invertible sheaf $i^*\mathcal{O}_{\mathbb{P}^2}(1)$ on E. Here we assume that E is smooth, i.e. E is either \mathbb{P}^2 or an elliptic cubic curve. Moreover $\sigma = \sigma_{abc}$ is an automorphism of Einduced by the shift functor on point modules of S whose image is the cross product of any two rows of the matrix M_i from Equation (2.1) [ATVdB90, §1]. In other words, σ can be explicitly given as follows:

(A.1)
$$\sigma([x:y:z]) = [acy^2 - b^2xz : bcx^2 - a^2yz : abz^2 - c^2xy].$$

Now the behavior of S(a, b, c) varies according to $|\sigma_{abc}|$. More precisely, S(a, b, c) is module-finite over its center if and only if $|\sigma_{abc}| < \infty$ [ATVdB91, Theorem 7.1]. Hence the following proposition yields the first steps in the classification of the parameters (a, b, c) of $Skly_3$ for which the automorphism σ_{abc} of smooth E_{abc} have finite order. Since $S(a, b, c) \cong S\left(1, \frac{b}{a}, \frac{c}{a}\right)$, we assume that a = 1.

Proposition A.1. Let ζ be a third root of unity. Then we have the following results.

• $|\sigma| = 1 \iff [a:b:c] = [1:-1:0]$, the origin of E. • $|\sigma| = 2 \iff [a:b:c] = [1:1:c]$, for $c \neq 0$, $c^3 \neq 1$.

Correction (pointed out by Daniel Reich):

For |sigma|=n |leq 3, only the implications <= hold. We actually have more $\bullet |\sigma| = 3 \iff [a:b:c] = \begin{cases} [1:0:\omega], \text{ for } \omega = -1, e^{\pi i/3}, e^{5\pi i/3}; \\ [1:\omega:0], \text{ for } \omega = e^{\pi i/3}, e^{5\pi i/3}. \end{cases}$.

•
$$|\sigma| = 4 \iff [a:b:c] = \begin{cases} \left[1:b:\left(\frac{b(b^2+1)}{b+1}\right)^{1/3}\zeta\right], & for \ b \neq 0, -1, \pm i, and \ b^3 \neq 1; \\ \left[1:\left(\frac{f(c)^{2/3}-12(1-c^3)}{6f(c)^{1/3}}\right)\zeta:c\right], & for \ c \neq 0, c^3 \neq 1. \end{cases}$$

where $f(c) = 108c^3 + 12(12 - 36c^3 + 117c^6 - 12c^9)^{1/3}$.

•
$$|\sigma| = 5 \iff [a:b:c] = \begin{cases} \left[1:b:\left(\frac{r\pm s^{1/2}}{2b}\right)^{1/3}\zeta\right], & for \ b \neq 0, \ b^3 \neq 1; \\ b \neq & primitive \ 10^{th} \ root \ of \ unity, \\ \left[1:g(c):c\right], & for \ c \neq 0, c^3 \neq 1. \end{cases}$$

where $r = -b^5 + b^4 + b^3 + b^2 + b - 1$ and $s = (b^2 - 3b + 1)(b - 1)^2(b^2 + b - 1)^3$. Moreover g(c) is a root of

$$Z^{6} + (c^{3} - 1)Z^{5} + (1 - c^{3})Z^{4} + (-1 - c^{3})Z^{3} + (1 - c^{3})Z^{2} + (c^{6} - c^{3})Z + c^{3}.$$

•
$$|\sigma| = 6 \iff [a:b:c] = \begin{cases} [1:b:\zeta], [1:b:b\zeta], & for \ b \neq 0, b^3 \neq 1, \\ [1:\zeta:c], [1:c\zeta:c], & for \ c \neq 0, c^3 \neq 1, \zeta \neq 1. \end{cases}$$

Proof. Refer to Routine A.3.

Remark A.2. Although there is no apparent pattern for the parameters (1, b, c) of a given $|\sigma| = n$, we believe that there are the following restrictions of b and c for $n \ge 4$.

- 1. When b is arbitrary and we solve for c, we must have that $b \neq 0$ and $b^3 \neq 1$. Moreover for
 - $4|n: b \neq \text{ primitive } n^{th} \text{ root of } 1$,
 - *n* even, $4 \not\mid n$: $b \neq$ primitive $\left(\frac{n}{2}\right)^{th}$ root of 1,
 - n odd: $b \neq$ primitive $(2n)^{th}$ root of 1.
- 2. When c is arbitrary and we solve for b, we must have that $c \neq 0$ and $c^3 \neq 1$.

Computer Routines

Affine / Maple routines are given as follows:

AFFINE / MAPLE
Sample routine
Sample comment

Routine A.3. This is the proof of Proposition A.1.

```
### MAPLE ###
## Let |sigma| =:n, and take n>1. Make sure to change n as needed!
n:= 6;
## Let us define sigma.
p:=(x,y,z)->a*c*y^2-b^2*x*z:
q:=(x,y,z)->b*c*x^2-a^2*y*z:
```

```
r:=(x,y,z)->a*b*z^2-c^2*x*y:
x[0]:=x: y[0]:=y: z[0]:=z:
## We assume that a=1.
a:=1:
```

For a given n, let x[n], y[n], z[n] be the entries of the image ## of sigma^n. Let us define sigma^n. for k from 0 to n-1 do x[k+1]:=p(x[k],y[k],z[k]); y[k+1]:=q(x[k],y[k],z[k]); z[k+1]:=r(x[k],y[k],z[k]); end do:

We want (x[n]:y[n]:z[n]) = (x:y:z) projectively. ## So we need f=y*x[n]-x*y[n], g=z*x[n]-x*z[n], h=z*y[n]-y*z[n] ## to be zero for some parameters (a,b,c), subject to the ## defining relation of E. f:=y*x[n]-x*y[n]: g:=z*x[n]-x*z[n]: h:=z*y[n]-y*z[n]: E:=((1+b^3+c^3)/(b*c))*(x*y*z)-x^3-y^3: ## We will replace the term z^3 with the expression E. d[1]:=2^n+1:

Special care is needed for the n=2 and n=3 cases. ## In general, reduce f,g,h with respect to the relation of E. ## For n>6 we reduce f first, then g and h later.

140

```
for k from 1 to n-1 do
f:= expand(f-(add( coeff(f,z^i)*z^i, i=3..d[k]+1 ))
        + add( coeff(f,z^j)*z^(j mod 3)*E^(iquo(j,3)), j=3..d[k]+1));
g:= expand(g-(add( coeff(g,z^i)*z^i, i=3..d[k]+1 ))
        + add( coeff(g,z^j)*z^(j mod 3)*E^(iquo(j,3)), j=3..d[k]+1));
h:= expand(h-(add( coeff(h,z^i)*z^i, i=3..d[k]+1 ))
        + add( coeff(h,z^j)*z^(j mod 3)*E^(iquo(j,3)), j=3..d[k]+1));
d[k+1]:= iquo(d[k],3) + irem(d[k],3);
```

end do:

```
## Now the degree of f,g,h with respect of z should be <3.
degree(f,z); degree(g,z); degree(h,z);</pre>
```

```
## We can either: (I) Let b be arbitrary and solve for c, or
## (II) Let c be arbitrary and solve for b
## Here is the computation for (I):
for k from 1 to 1 do
if type(n,odd) then
v:=solve(coeff(f,x^(d[1]))=0,c);
w:=solve(coeff(coeff(f,x^(d[1]-3)),y^3)=0,c);
else
v:=solve(coeff(coeff(f,x^(d[1]-1)),y)=0,c);
w:=solve(coeff(coeff(f,x^(d[1]-4)),y^4)=0,c);
end if end do:
```

```
## Here is the computation for (II):
for k from 1 to 1 do
if type(n,odd) then
v:=solve(coeff(f,x^(d[1]))=0,b);
w:=solve(coeff(coeff(f,x^(d[1]-3)),y^3)=0,b);
else
v:=solve(coeff(coeff(f,x^(d[1]-1)),y)=0,b);
w:=solve(coeff(coeff(f,x^{(d[1]-4)}, y^{4})=0, b);
end if
           end do:
## We create a list of potential nonzero unique solutions
lenv:=nops([v]):
                     lenw:=nops([w]):
u:=[]:
for i from 1 to lenv do
for j from 1 to lenw do
t:=evalb(v[i]=w[j]):
if t=true then
if v[i] <> 0 then
u:=[op(u),v[i]];
end if
           end if
                      end do:
                                   end do:
lenu:=nops(u):
with(ListTools):
                     L:=MakeUnique(u);
                                            lenL:=nops(L);
```

So L is the list of solutions for which we need verify
that f,g,h (now reduced by the relation of E) are 0

```
## Case (I), letting b be arbitrary we use the subroutine:
for k from 1 to lenL do
c:=expand(L[k]): print(simplify(c));
newf:=expand(f): newg:=expand(g): newh:=expand(h):
print(k, simplify(newf), simplify(newg), simplify(newh));
end do:
## (II) For c arbitrary we use the subroutine:
for k from 1 to lenL do
b:=expand(L[k]): print(simplify(b));
newf:=expand(f): newg:=expand(g): newh:=expand(h):
print(k, simplify(newf), simplify(newg), simplify(newh));
end do:
```

```
## If the output is "0,0,0", then L is the desired list of
## parameters (1,b,c) so that sigma(1bc)^n = 1.
## Eliminate by hand parameters (1,b,c) for which
## sigma(1bc)^m = 1 with m<n.</pre>
```

ORDER 2 RESULTS:

Compute f,g,h (modulo the relation of E) for sigma^2 as described
at the beginning of the routine. In particular, we want
(reduced) f to equal zero for some parameters (1,b,c).
> collect(f,[z,y,x]);

 $((c - b^3 c) y^3 + (b^2 c - b^5 c) x^3) z^2 + [...]$

Consider the coefficient of y^3*z^2: c(b^3-1).
If equal to zero, then c=0 or b^3=1.
If c=0 and b is arbitrary, then either E is not smooth
or |sigma|=3. Therefore we take c nonzero.

First let b=1. Then we get that c arbitrary yields a solution.
> L:=MakeUnique(u);

$$L := [c[1]]$$

> b:=1;

b := 1

> (simplify(f),simplify(g),simplify(h));

0, 0, 0

```
## Now let b:=exp(2*Pi*I/3);
```

> L:=MakeUnique(u);

L := $[1, -1/2 + 1/2 I 3^{(1/2)}, -1/2 - 1/2 I 3^{(1/2)}]$

so c^3=1, and S is not a Sklyanin algebra.

Likewise result for b:=exp(4*Pi*I/3);

Therefore the only solution is (1,b,c) = (1,1,c)

for c nonzero, not a third root of 1.

ORDER 3 RESULTS:

(I) b nonzero and arbitrary, solving for c:

> L:=MakeUnique(u); lenL:=nops(L);

L := [] lenL := 0

(II) same as (I)

ORDER 4,5,6 RESULTS for subroutines (I, II) are given in Prop. A.1

Routine A.4. We explicitly list the 1-dimensional representations of

 $S_{def}(a, b, c, d_i, e_i)$ for each of the cases: $d_i = 0$, $e_i = 0$, and $d_2 = d_3 = e_2 = e_3 = 0$. We use the following short list of commands in Maple, and by rescaling we assume that the parameter a is 1.

MAPLE

a:=1; f:=a*y*z + b*z*y + c*x^2 + d1*x + e1; g:=a*z*x + b*x*z + c*y^2 + d2*y + e2; h:=a*x*y + b*y*x + c*z^2 + d3*z + e3; v:=solve({f=0, g=0, h=0}, [x,y,z]);

For $d_i = 0$, there is one 1-dimensional representation:

$$v := [[x = (-e2 c e1^2 b^2 - 2 e2 c e1^2 b {...}] - 4 %1^2 e2^2 b) / ((1 + b) (%1^2 c^3 e2 {...}] + c e1^2 b^2 + c e1^2) %1), y = - %1 (2 e3 c {...}] + 2 %1^2 c^3 e3 e2) /$$

$$(-e2 c e1^2 b^2 {\ldots} - 4 \% 1^2 e2^2 b), z = \% 1]$$

For $e_i = 0$, there are five 1-dimensional representations:

$$[x = - (\%1 \ d1 \ b^3 \ \{...\} \ d3) / (c \ (\%1 \ d3 \ b^3 \ \{...\} - d1^2)^2),$$

y = %1,
z = (-%1^2 \ b^4 \ d1 - {...} d2 \ d3^2 \ c^2) / (c \ {...} - d1^2)]]

%1 := RootOf((3 c³ + {...} + 5 d³² b⁴ d¹²)

For $d_2 = d_3 = e_2 = e_3 = 0$, there are three 1-dimensional representations:

 $[x = ((\%1 b^3 + \{...\} + c d1) c \%1^2) / (e1 + \%1 c d1),$ y = - (e1 b + %1 b c d1 + e1 + %1 c d1)/(%1 b^3 + {...} + c d1), z = %1 (1 + b)]]

Routine A.5. Here we compute the point scheme of a central extension D of a threedimensional Sklyanin algebra. A detailed point scheme computation in provided in Proposition II.15 for example. Recall the definition of a d^{th} truncated point scheme V_d (Definition II.4) and the results of Definition-Lemma II.6. Furthermore, note that V_1 for D is isomorphic to \mathbb{P}^3 . Considering the multilinearizations of relations of D, we get that $\mathbb{M}_i \cdot [x_{i+1} : y_{i+1} : z_{i+1} : w_{i+1}]^T = 0$ for

$$\mathbf{M_i} = \begin{pmatrix} cx_i & bz_i & ay_i & d_1x_i + e_1w_i \\ az_i & cy_i & bx_i & d_2y_i + e_2w_i \\ by_i & ax_i & cz_i & d_3z_i + e_3w_i \\ -w_i & 0 & 0 & x_i \\ 0 & -w_i & 0 & y_i \\ 0 & 0 & -w_i & z_i \end{pmatrix}$$

To show that the point scheme data (V_d, π_d) of D stabilizes at d = 2, we must verify that rank $(M_i) \ge 3$ for all $i \ge 1$. Suppose not, then we get from the routine below that $x_i = y_i = z_i = w_i = 0$, a contradiction. Here we drop the index i.

```
### MAPLE ###
```

```
with(LinearAlgebra):
```

M : =	<<	с*х	Ι	b*z	I	a*y	I	d1*x+e1*w	>,
	<	a*z		c*y		b*x	I	d2*y+e2*w	>,
	<	b*y		a*x		C*Z		d3*z+e3*w	>,
	<	-W	I	0	I	0	I	x	>,
	<	0	I	-w	I	0	I	У	>,
	<	0	I	0	I	-w	I	z	>>;

Create a list of the 4X4 full submatrices of M (in row notation).

```
## There are 6 choose 4 (so 15) of them.
R:=[];
for i from 1 to 6 do
R:=[op(R),Row(M,i)]:
end do:
with(combinat):
S:=choose(R,4):
```

```
## Convert the matrices in row notation into the "Matrix" format.
L:=[]; L1:=[];
for n from 1 to 15 do
for i from 1 to 4 do
A[i]:=convert(S[n][i],Matrix):
end do:
B:=<Transpose(A[1])|Transpose(A[2])|Transpose(A[3])|Transpose(A[4])>:
C:=Transpose(B):
```

```
## Then for each of the 15 matrices,
## we compute the 3x3 minors and make a list of them.
for i from 1 to 4 do
for j from 1 to 4 do
m:=Minor(C,i,j):
L:=[op(L),m]:
```

We set the 240 3x3 minors equal to 0, then solve the system. for i from 1 to 240 do L1:=[op(L1),L[i]=0]: end do: v:=solve(L1,[x,y,z,w]); v := [[x = 0, y = 0, z = 0, w = 0]]

Hence if all of the 3×3 minors of M_i vanish, then we get a contradiction. Thus rank $(M_i) \ge 3$ for $i \ge 1$. Now given the map $\pi_2 : V_2 \to \mathbb{P}^3$, we know that the point scheme P_D of D is isomorphic to the graph of the image of π_2 . Call this image Xand we have that

$$X = \{ [x_1 : y_1 : z_1 : w_1] \in \mathbb{P}^3 \mid \operatorname{rank}(M_1) < 4 \}.$$

To compute X, we need solutions $[x_1 : y_1 : z_1 : w_1] \in \mathbb{P}^3$ so that determinants of the full 4×4 submatrices of M_1 vanish. The following is the Maple routine to achieve such a result; the solutions are explicitly provided in the cases where the parameters d_i or e_i are 0. Here we drop the index 1.

```
### MAPLE ###
## (Optional) d1:= 0; d2:= 0; d3:= 0; or e1:= 0; e2:= 0; e3:= 0;
          or d2:= 0; d3:= 0; e2:= 0; e3:= 0;
##
## Define the matrix M1.
with(LinearAlgebra):
M:= << c*x | b*z
                  | a*y | d1*x+e1*w >,
     < a*z | c*y | b*x
                           | d2*y+e2*w >,
     < b*y | a*x | c*z
                           | d3*z+e3*w >,
     < -w | 0
                  0
                           | x
                                       >,
```

< 0 | -w | 0 | y >, < 0 | 0 | -w | z >>;

Create a list of the 4X4 full submatrices of M (in row notation).
There are 6 choose 4 (so 15) of them.
R:=[];
for i from 1 to 6 do
R:=[op(R),Row(M,i)]:
end do:
with(combinat):
S:=choose(R,4):

```
## Convert the matrices in row notation into the "Matrix" format.
## Now for each matrix in the list, compute its determinant.
## Create a list of the 15 determinants excluding
## redundancies and zeros. Call this list T.
t:=[]:
for n from 1 to 15 do
for i from 1 to 15 do
A[i]:=convert(S[n][i],Matrix):
end do:
B:=<Transpose(A[1])|Transpose(A[2])|Transpose(A[3])|Transpose(A[4])>:
C:=Transpose(B):
d:=Determinant(C);
if d<>0 then
```

t:=[op(t),d]:

end if: end do:

with(ListTools):

T:=MakeUnique(t):

```
[x = 0, y = -z, z = z, w = 0]
[x = 0, y = RootOf(_Z<sup>2</sup> + 1 - _Z, label = _L1) z, z = z, w = 0]
```

 $[x = -z (c^3 e^2 - {...} + e^2 b^3 e^3 - {...} / e^2 b^3 e^3 - {...}]$

%1 := RootOf((c e1 a² e2 + {...} - a³ e²²) _Z³ , label = _L2)

Thus in the case $d_i = 0$, the cardinality of the set S is 1. In fact,

$$S = \left\{ \left[\frac{c^3 + \{\dots\}}{c^2(e_2^3 - \{\dots\})} : R : 1 : \sqrt{e_2^3 + \{\dots\}} \right] \right\}$$

where R is the "Rootof" expression above.

for e1 = e2 = e3 = 0: [x = RootOf(c _Z^3 a b + (-y b^3 z - y a^3 z - y c^3 z) _Z + y^3 b c a + a b z^3 c), y = y, z = z, w = 0] [x = 0, y = -z, z = z, w = 0] [x = 0, y = RootOf(_Z^2 + 1 - _Z, label = _L1) z, z = z, w = 0] [x = 0, y = - d2 w / c, z = 0, w = w] [x = 0, y = 0, z = 0, w = w] [x = - d1 w / c, y = 0, z = 0, w = w] [x = 0, y = 0, z = z, w = - c z / d3]

 $[x = \%1 \ z \ c \ (-d2 + \%1 \ d3) \ / \ (d2 \ \%1^2 \ - \ d3) \ (a + b),$

$$y = \%1 z$$
, $z = z$, $w = -z (\%1^3 - 1) c / d2 \%1^2 - d3$]

Thus in the case where e_i = 0, the cardinality of the set ${\mathcal S}$ is 5. In fact,

$$\mathcal{S} = \left\{ \begin{bmatrix} 0:0:0:1 \end{bmatrix}, \ \begin{bmatrix} \frac{-d_1}{c}:0:0:1 \end{bmatrix}, \ \begin{bmatrix} 0:\frac{-d_2}{c}:0:1 \end{bmatrix}, \ \begin{bmatrix} 0:0:\frac{-d_3}{c}:1 \end{bmatrix}, \ \end{bmatrix}$$

$$\left\{\frac{Rc(-d_2+Rd_3)}{(d_2R^2-d_3)(a+b)}: R:1:\frac{(R^3-1)c}{d_2R^2-d_3}\right\}$$

where R is the "Root of" expression above.

%1 := RootOf(e1^2 _Z^4 + {...} + _Z^3 c d1 e1, label = _L6)

Thus in the case where $d_2 = d_3 = e_2 = e_3 = 0$, the cardinality of the set S is 3. We can interpret the points of S as in the previous cases.

Routine A.6. We verify the construction of the rank two tensors of $D(1, b, c, d_1, d_2, d_3, e_1, e_2, e_3)$ (and hence of $D(1, 1, c, d_1, d_2, d_3, e_1, e_2, e_3)$) used in the proof of Lemma V.18 (and Lemma V.27) in §5.4.

AFFINE

Define the algebra D(1,b,c,d_1,d_2,d_3,e_1,e_2,e_3)
declare(c,constant,d1,constant,d2,constant,d3,constant,
e1,constant,e2,constant,e3,constant);
declare_weights(x,1,y,1,z,1,w,1);
ALL_DOTSIMP_DENOMS:[];
r1:y.z+b*z.y+c*x.x+d1*x.w+e1*w.w;
r2:z.x+b*x.z+c*y.y+d2*y.w+e2*w.w;
r3:x.y+b*y.x+c*z.z+d3*z.w+e3*w.w;
r4:x.w-w.x; r5:y.w-w.y; r6:z.w-w.z;
set_up_dot_simplifications([r1,r2,r3,r4,r5,r6],7);
y n n n [x,y,z,w];

Define the rank two tensor expressions au-bv
declare(b1,constant,u2,constant,u3,constant,u4,constant,

v1, constant, v4, constant);

a1: b1*v1;	a2: v1*b1*u2;	a3: b1*v1*u3;	a4: v4*b1;
b2: b1*u2;	b3: b1*u3;	b4: b1*u4;	
u1: 1;	v2: u2*v1;	v3: v1*u3;	
r: a1*x+a2*y+	a3*z+a4*w;	s: b1*x+b2*y+b3*z	+b4*w;
u: u1*x+u2*y+	u3*z+u4*w;	v: v1*x+v2*y+v3*z	;+v4*w;

```
## Verify that ru-sv in the ideal of relations of D
dotsimp(r.u-s.v);
(D28) 0
```

Routine A.7. For D with $|\sigma| = 2$, we compute central annihilators for the line modules $M_l = D/Du + Dv$ given in Lemma V.27. This result is used for the proof of Lemma V.28 in §5.4. In particular, we need scalars a_{ij} so that:

- $f_i = a_{i1}x + a_{i2}y + a_{i3}z + a_{i4}w$ for i = 1, 2, and;
- $f_1 u + f_2 v = \alpha w^2 + \beta \hat{g_2}$ for some $[\alpha : \beta] \in \mathbb{P}^1$;

where \hat{g}_2 is the degree 2 central element of D not in $k[w^2]$. As mentioned in the proof of Lemma V.28, we assume that $[\alpha : \beta] = [\gamma : 1]$ for $\gamma \in k$.

```
### AFFINE ###
```

Define and simplify the expression f1*u + f2*v.

declare(a11, constant, a12, constant, a13, constant, a14, constant,

a21, constant, a22, constant, a23, constant, a24, constant);

f1:a11*x+a12*y+a13*z+a14*w;

f2:a21*x+a22*y+a23*z+a24*w;

declare(u2, constant, u3, constant, u4, constant, v1, constant, v4, constant); v2: u2*v1; u1: 1; v3: v1*u3; u: u1*x+u2*y+u3*z+u4*w; v: v1*x+v2*y+v3*z+v4*w; dotsimp(f1.u+f2.v); /R/ (((a22 C u3 - a23 C u2) v1 + a12 C u3 - a13 C u2) (Y . z) + ((a22 C u2 - a23 C²) v1 + a12 C u2 - a13 C²) (Y . Y) + ((- a23 u3 + a22 C) v1 - a13 u3 + a12 C) (Y . X)+ ((a21 C u3 - a23 C) v1 + a11 C u3 - a13 C) (X . z) + ((- a23 u3 + a21 C u2) v1 - a13 u3 + a11 C u2) (X . Y) + ((- a23 C² u2 + a21 C) v1 - a13 C² u2 + a11 C) (X . X) + (a23 C v4 + (- a23 d3 + a24 C) u3 v1 + a13 C u4 + (- a13 d3 + a14 C) u3) (w . z)+ (a22 C v4 + (a24 C u2 - a23 C d2) v1 + a12 C u4 + a14 C u2 - a13 C d2) (w . Y) + (a21 C v4 + (- a23 C d1 u2 + a24 C) v1 + a11 C u4 - a13 C d1 u2 + a14 C) (w . X) + (a24 C v4 + (- a23 e3 u3 - a23 C e1 u2 - a23 C e2) v1 + a14 C u4 - a13 e3 u3 - a13 C e1 u2 - a13 C e2) (w . w))/C ### MAPLE ### ## We extract the coefficients of the simplified expression f1*u + f2*v. Call such coefficients "annrel**". ## annrelyz:=(1/c)*((a22*c*u3 - a23*c*u2)*v1 + a12*c*u3 - a13*c*u2);

annrelyy:=(1/c)*((a22*c*u2 - a23*c^2)*v1 + a12*c*u2 - a13*c^2);

```
annrelyx:=(1/c)*((-a23*u3 + a22*c)*v1 - a13*u3 + a12*c);
annrelxz:=(1/c)*((a21*c*u3 - a23*c)*v1 + a11*c*u3 - a13*c);
annrelxy:=(1/c)*((-a23*u3 + a21*c*u2)*v1 - a13*u3 + a11*c*u2);
annrelxx:=(1/c)*((-a23*c^2*u2 + a21*c)*v1 - a13*c^2*u2 + a11*c);
annrelwz:=(1/c)*(a23*c*v4 + (-a23*d3 + a24*c)*u3*v1 + a13*c*u4
+ (- a13*d3 + a14*c)*u3);
annrelwy:=(1/c)*(a22*c*v4 + (a24*c*u2 - a23*c*d2)*v1 + a12*c*u4
+ a14*c*u2 - a13*c*d2);
annrelwx:=(1/c)*(a21*c*v4 + (-a23*c*d1*u2 + a24*c)*v1 + a11*c*u4
- a13*c*d1*u2 + a14*c);
annrelwy:=(1/c)*(a24*c*v4 + (-a23*e3*u3 - a23*c*e1*u2 - a23*c*e2)*v1
+ a14*c*u4 - a13*e3*u3 - a13*c*e1*u2 - a13*c*e2);
```

```
## Now considering the coefficients of the central element \hat{g2},
## we solve for the a(ij).
```

```
## Note that to get a solution, we must also solve for u2 and u3.
v:=solve({annrelyz=0,annrelyy=(c<sup>4</sup>-c)*d2,annrelyx=(-c<sup>3</sup>+1)*d3,
```

```
annrelxz=0,annrelxy=(-c^3+1)*d3,annrelxx=(c^4-c)*d1,
annrelwz=d3^2-c^2*d1*d2,annrelwy=-c^2*d1*d3 + c^3*d2^2,
annrelwx=-c^2*d2*d3 + c^3*d1^2},
```

```
[a11,a12,a13,a14,a21,a22,a23,a24,u2,u3]);
```

We get one long solution, v.

> nops(v);

Set the a(ij) equal to the corresponding solutions of v[1][k].

a11:= rhs(v[1][1]);	a12:= rhs(v[1][2]);
a13:= rhs(v[1][3]);	a14:= rhs(v[1][4]);
a21:= rhs(v[1][5]);	a22:= rhs(v[1][6]);
a23:= rhs(v[1][7]);	a24:= rhs(v[1][8]);
u2:= rhs(v[1][9]);	u3:= rhs(v[1][10]);

Now gamma equals the following:

annrelww:=(1/c)*(a24*c*v4 + (-a23*e3*u3 - a23*c*e1*u2 - a23*c*e2)*v1

+ a14*c*u4 - a13*e3*u3 - a13*c*e1*u2 - a13*c*e2);

gamma:= simplify(annrelww);

This is a long expression, and it is dependent on u4, v1, v4.

BIBLIOGRAPHY

BIBLIOGRAPHY

- [AdJ] M. Artin and A. J. de Jong. Stable orders over surfaces (course notes). Math 711, Winter 2004, University of Michigan.
- [Ami74] S. A. Amitsur. Polynomial identities. Israel J. Math., 19:183–199, 1974.
- [Art69] Michael Artin. On Azumaya algebras and finite dimensional representations of rings. J. Algebra, 11:532–563, 1969.
- [Art92] Michael Artin. Geometry of quantum planes. In Azumaya algebras, actions, and modules (Bloomington, IN, 1990), volume 124 of Contemp. Math., pages 1–15. Amer. Math. Soc., Providence, RI, 1992.
- [AS87] Michael Artin and William F. Schelter. Graded algebras of global dimension 3. Adv. in Math., 66(2):171–216, 1987.
- [AS95] M. Artin and J. T. Stafford. Noncommutative graded domains with quadratic growth. Invent. Math., 122(2):231–276, 1995.
- [ATVdB90] M. Artin, J. Tate, and M. Van den Bergh. Some algebras associated to automorphisms of elliptic curves. In *The Grothendieck Festschrift, Vol. I*, volume 86 of *Progr. Math.*, pages 33–85. Birkhäuser Boston, Boston, MA, 1990.
- [ATVdB91] M. Artin, J. Tate, and M. Van den Bergh. Modules over regular algebras of dimension 3. Invent. Math., 106(2):335–388, 1991.
- [AVdB90] M. Artin and M. Van den Bergh. Twisted homogeneous coordinate rings. J. Algebra, 133(2):249–271, 1990.
- [AZ94] M. Artin and J. J. Zhang. Noncommutative projective schemes. Adv. Math., 109(2):228–287, 1994.
- [AZ01] M. Artin and J. J. Zhang. Abstract Hilbert schemes. Algebr. Represent. Theory, 4(4):305–394, 2001.
- [Ber78] George M. Bergman. The diamond lemma for ring theory. Adv. in Math., 29(2):178–218, 1978.
- [BG97] K. A. Brown and K. R. Goodearl. Homological aspects of Noetherian PI Hopf algebras of irreducible modules and maximal dimension. J. Algebra, 198(1):240–265, 1997.
- [BG02] Ken A. Brown and Ken R. Goodearl. *Lectures on algebraic quantum groups*. Advanced Courses in Mathematics. CRM Barcelona. Birkhäuser Verlag, Basel, 2002.
- [BG06] Enrico Bombieri and Walter Gubler. *Heights in Diophantine geometry*, volume 4 of *New Mathematical Monographs*. Cambridge University Press, Cambridge, 2006.
- [BJL00] David Berenstein, Vishnu Jejjala, and Robert G. Leigh. Marginal and relevant deformations of N = 4 field theories and non-commutative moduli spaces of vacua. Nuclear Phys. B, 589(1-2):196–248, 2000.

- [BRS10] J. Bell, D. Rogalski, and S. J. Sierra. The Dixmier-Moeglin equivalence for twisted homogeneous coordinate rings. *Israel J. Math.*, 180:461–507, 2010.
- [BS] J.E Björk and J.T. Stafford. email correspondence with M. Artin.
- [BT07] Roland Berger and Rachel Taillefer. Poincaré-Birkhoff-Witt deformations of Calabi-Yau algebras. J. Noncommut. Geom., 1(2):241–270, 2007.
- [Cha60] Stephen U. Chase. Direct products of modules. *Trans. Amer. Math. Soc.*, 97:457–473, 1960.
- [CS08] Thomas Cassidy and Brad Shelton. Generalizing the notion of Koszul algebra. *Math.* Z., 260(1):93–114, 2008.
- [EG] P. Etingof and V. Ginzburg. Noncommutative del pezzo surfaces and calabi-yau algebras. http://arxiv.org/abs/0709.3593.
- [For74] Edward Formanek. Noetherian PI-rings. Comm. Algebra, 1:79–86, 1974.
- [Gro61] A. Grothendieck. Éléments de géométrie algébrique. II. Étude globale élémentaire de quelques classes de morphismes. *Inst. Hautes Études Sci. Publ. Math.*, (8):222, 1961.
- [GW04] K. R. Goodearl and R. B. Warfield, Jr. An introduction to noncommutative Noetherian rings, volume 61 of London Mathematical Society Student Texts. Cambridge University Press, Cambridge, second edition, 2004.
- [Har77] Robin Hartshorne. *Algebraic geometry*. Springer-Verlag, New York, 1977. Graduate Texts in Mathematics, No. 52.
- [Kee00] Dennis S. Keeler. Criteria for σ -ampleness. J. Amer. Math. Soc., 13(3):517–532 (electronic), 2000.
- [KL00] Günter R. Krause and Thomas H. Lenagan. Growth of algebras and Gelfand-Kirillov dimension, volume 22 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, revised edition, 2000.
- [Kra] U. Krahmer. Notes on koszul algebras. http://www.impan.gov.pl/~kraehmer/ connected.pdf.
- [KRS05] D. S. Keeler, D. Rogalski, and J. T. Stafford. Naïve noncommutative blowing up. Duke Math. J., 126(3):491–546, 2005.
- [Lam99] T. Y. Lam. Lectures on modules and rings, volume 189 of Graduate Texts in Mathematics. Springer-Verlag, New York, 1999.
- [LB94] Lieven Le Bruyn. The arithmetic of Sklyanin algebras. I. The defining equations. Comm. Algebra, 22(1):51–81, 1994.
- [LB95a] Lieven Le Bruyn. Central singularities of quantum spaces. J. Algebra, 177(1):142–153, 1995.
- [LB95b] Lieven Le Bruyn. Conformal sl₂ enveloping algebras. Comm. Algebra, 23(4):1325–1362, 1995.
- [LBSVdB96] Lieven Le Bruyn, S. P. Smith, and Michel Van den Bergh. Central extensions of three-dimensional Artin-Schelter regular algebras. *Math. Z.*, 222(2):171–212, 1996.
- [LS93] Thierry Levasseur and S. Paul Smith. Modules over the 4-dimensional Sklyanin algebra. Bull. Soc. Math. France, 121(1):35–90, 1993.

[MR01] J. C. McConnell and J. C. Robson. Noncommutative Noetherian rings, volume 30 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, revised edition, 2001. With the cooperation of L. W. Small. [NSa] T. Nevins and S. Sierra. Canonical birationally commutative factors of noetherian graded algebras. in preparation. [NSb] T. Nevins and S. Sierra. Moduli spaces for point modules on naïve blowups (preprint). http://arxiv.org/abs/1009.2061. [Pio96] D. I. Piontkovskii. Gröbner bases and the coherence of monomial associative algebra. Fundam. Prikl. Mat., 2(2):501-509, 1996. [Pio08] Dmitri Piontkovski. Coherent algebras and noncommutative projective lines. J. Algebra, 319(8):3280-3290, 2008. [Pol05] A. Polishchuk. Noncommutative proj and coherent algebras. Math. Res. Lett., 12(1):63-74, 2005.[PP05] Alexander Polishchuk and Leonid Positselski. Quadratic algebras, volume 37 of University Lecture Series. American Mathematical Society, Providence, RI, 2005. [RRZ06] Z. Reichstein, D. Rogalski, and J. J. Zhang. Projectively simple rings. Adv. Math., 203(2):365-407, 2006. [RS]D. Rogalski and S. Sierra. Some noncommutative projective surfaces of gk-dimension 4 (preprint). http://arxiv.org/abs/1101.0737. D. Rogalski and J. J. Zhang. Canonical maps to twisted rings. Math. Z., 259(2):433-[RZ08] 455, 2008. [Ser 55]Jean-Pierre Serre. Faisceaux algébriques cohérents. Ann. of Math. (2), 61:197–278, 1955.[Smi94] S. P. Smith. The four-dimensional Sklyanin algebras. In Proceedings of Conference on Algebraic Geometry and Ring Theory in honor of Michael Artin, Part I (Antwerp, 1992), volume 8, pages 65–80, 1994. [Smi97] Karen E. Smith. Fujita's freeness conjecture in terms of local cohomology. J. Algebraic Geom., 6(3):417-429, 1997. [SS93] S. Paul Smith and J. M. Staniszkis. Irreducible representations of the 4-dimensional Sklyanin algebra at points of infinite order. J. Algebra, 160(1):57–86, 1993. [ST94] S. Paul Smith and John Tate. The center of the 3-dimensional and 4-dimensional Sklyanin algebras. *K-Theory*, 8(1):19–63, 1994. [ST01] Brad Shelton and Craig Tingey. On Koszul algebras and a new construction of Artin-Schelter regular algebras. J. Algebra, 241(2):789–798, 2001. [Sta] J. T. Stafford. Noncommutative projective algebraic geometry (course notes). Math 715, Winter 2007, University of Michigan. [Sta02] J. T. Stafford. Noncommutative projective geometry. In Proceedings of the International Congress of Mathematicians, Vol. II (Beijing, 2002), pages 93–103, Beijing, 2002. Higher Ed. Press. [Ste] D. Stephenson. The geometry of noncommutative graded algebras (preliminary version).

- [SV02a] Brad Shelton and Michaela Vancliff. Schemes of line modules. I. J. London Math. Soc. (2), 65(3):575–590, 2002.
- [SV02b] Brad Shelton and Michaela Vancliff. Schemes of line modules. II. Comm. Algebra, 30(5):2535–2552, 2002.
- [SvdB01] J. T. Stafford and M. van den Bergh. Noncommutative curves and noncommutative surfaces. *Bull. Amer. Math. Soc.* (N.S.), 38(2):171–216 (electronic), 2001.
- [SW] S. Paul Smith and Chelsea Walton. Module categories of degenerate Sklyanin algebras (title subject to change). work in progress.
- [SZ97] Darin R. Stephenson and James J. Zhang. Growth of graded Noetherian rings. Proc. Amer. Math. Soc., 125(6):1593–1605, 1997.
- [Wal09] Chelsea Walton. Degenerate Sklyanin algebras and generalized twisted homogeneous coordinate rings. J. Algebra, 322(7):2508–2527, 2009.
- [Zha96] J. J. Zhang. Twisted graded algebras and equivalences of graded categories. Proc. London Math. Soc. (3), 72(2):281–311, 1996.
- [Zha98] James J. Zhang. Non-Noetherian regular rings of dimension 2. Proc. Amer. Math. Soc., 126(6):1645–1653, 1998.
- [Zwi09] Barton Zwiebach. A first course in string theory. Cambridge University Press, Cambridge, second edition, 2009. With a foreword by David Gross.